# Adaptation challenges and opportunities for the European energy system

Building a climate-resilient low-carbon energy system





# Adaptation challenges and opportunities for the European energy system

Building a climate-resilient low-carbon energy system



European Environment Agency

Cover photo: © Leyla Emektar, Picture 2050/EEA

# Legal notice

The contents of this publication do not necessarily reflect the official opinions of the European Commission or other institutions of the European Union. Neither the European Environment Agency nor any person or company acting on behalf of the Agency is responsible for the use that may be made of the information contained in this report.

# **Copyright notice**

© European Environment Agency, 2019 Reproduction is authorised provided the source is acknowledged.

More information on the European Union is available on the Internet (http://europa.eu).

Luxembourg: Publications Office of the European Union, 2019

ISBN 978-92-9480-065-7 ISSN 1977-8449 doi:10.2800/227321



European Environment Agency Kongens Nytorv 6 1050 Copenhagen K Denmark

Tel.: +45 33 36 71 00 Web: eea.europa.eu Enquiries: eea.europa.eu/enquiries

# Contents

Ac	knov	vledgements	4
Ex	ecuti	ive summary	5
1		oduction	
	1.1	Purpose and content of this report	
	1.2	Scope of the report	
	1.3	What do adaptation and climate resilience mean?	
2	The	European energy system nowand in the future	
	2.1	The European energy system	
	2.2	Building a low-carbon energy system	24
3	Clim	nate change impacts on the European energy system	
	3.1	Overview of climate change projections for Europe	32
	3.2	Impacts of increasing temperature	35
	3.3	Impacts of changing water availability	37
	3.4	Impacts of changes in extreme climate-related events	39
	3.5	Impacts of changes in coastal and marine hazards	42
	3.6	Further impacts on renewable energy potential	43
	3.7	Cross-cutting impact assessments of the European energy system	45
	3.8	Summary of adaptation needs, opportunities and option	47
4	Buil	ding a climate-resilient energy system	51
	4.1	Adaptation types, actors and barriers	51
	4.2	European Union policies and actions	60
	4.3	Activities by European countries	69
	4.4	Activities by international organisations	74
	4.5	Adaptation case studies from energy utilities and network providers	77
5	Con	clusions and outlook	82
An	nex	1 Information sources for country overview table	86
Ab	brev	riations	94
Re	fere	nces	96

# Acknowledgements

# **Report coordination**

This report was coordinated by Hans-Martin Füssel (European Environment Agency, EEA) with the support of André Jol (EEA). It was written by Hans-Martin Füssel with contributions from Blaz Kurnik, Mihai Tomescu, Wouter Vanneuville (EEA), Matthew Smith, Federica Gerber, Irati Artola (Trinomics, Netherlands), Veronique Adriaenssens (Arcadis, Belgium), Andrea Bigano (European Topic Centre on Climate Change impacts, vulnerability and Adaptation (ETC/CCA), *Centro Euro-Mediterraneo sui Cambiamenti Climatici* (CMCC), Italy), Pieter Vingerhouts (European Topic Centre on Climate change Mitigation and Energy (ETC/CME), VITO, Belgium) and Hans Eerens (ETC/CME, Netherlands).

We would like to acknowledge the kind support of a wide range of experts and stakeholders who provided comments on draft versions of this report.

# Comments from the European Commission and EU Agencies

Claus Kondrup, Andras Toth (DG Climate Action), Ville Niemi (DG Energy) and Rafael Muruais Garcia (EU Agency for the Cooperation of Energy Regulators (ACER)).

# Comments from National Reference Centres

Petra Mahrenholz (German Environment Agency, Germany), Ioanna Tsalakanidou (Ministry for the Environment and Energy, Greece), Valentina Rastelli (Istituto Superiore per la Protezione e Ricerca Ambientale (ISPRA), Italy), Herdis Laupsa (Norwegian Environment Agency, Norway), Carla Manuela Aleixo Martins (*Direcção-Geral Energia e Geologia*, Portugal), Jose Paulino (*Agência Portuguesa do Ambiente*, Portugal), Linda Kaneryd (Swedish Energy Agency, Sweden), Asa Sjöström (Swedish Meteorological and Hydrological Institute, Sweden), Ayse Yildirim Cosgun (Ministry of Agriculture and Forestry, Turkey) and Ross Lowrie (Environment Agency (England), United Kingdom).

## Members of the Stakeholder Group not listed above and additional expert reviewers

Samuel Almond (European Centre for Medium-Range Weather Forecasts (ECMWF), United Kingdom), Paul Behrens (Leiden University, Netherlands), Gert de Block (European Federation of Local Energy Companies (CEDEC), Belgium), James Falzon (European Bank for Reconstruction and Development (EBRD), United Kingdom), Bernard Gindroz (CEN-CENELEC, Belgium), Simon Jude (Cranfield University, United Kingdom), Krzysztof Laskowski (Eurelectric, Belgium), Caroline Lee (International Energy Agency (IEA), France), Michael Mullan (Organisation for Economic Co-operation and Development (OECD), France), Andrea Nam (CEN-CENELEC, Belgium), Susanne Nies (ENTSO-E, Belgium), Pedro Paes (Energias de Portugal (EDP), Portugal), Mathis Rogner (International Hydropower Association (IHA), United Kingdom), Marius Stankoweit (Climate Service Center (GERICS), Germany), Alberto Troccoli (World Energy & Meteorology Council (WEMC), United Kingdom) and Molly Walton (IEA, France).

# **Executive summary**

#### **Key messages**

- The European energy system increasingly needs to adapt and become more climate resilient in the context of continuing climate change, modern societies' increasing dependence on a reliable energy supply and an increasing share of climate-sensitive renewable energy sources.
- Climate change and extreme weather events increasingly impact all components of the energy system. They affect the availability of primary energy sources (in particular renewable energy sources), the transformation, transmission, distribution and storage of energy, and energy demand. It is crucial that these impacts are considered in the clean energy transition.
- Key EU climate and energy policies and strategies promote the mainstreaming of climate change adaptation into energy policies. The development of the Energy Union — which aims to make energy more secure, affordable and sustainable — provides important opportunities for further integrating climate change adaptation in European and national energy planning.
- Most European countries have addressed the energy sector in national climate change impact, vulnerability and risk
  assessments, national adaptation strategies and/or action plans. Further action is needed to consider the impacts of
  climate change in the development of national climate and energy plans and long-term strategies under the Energy Union,
  and in the update of national adaptation strategies and action plans. Governments can further facilitate adaptation
  through the regulation of energy markets and other policy interventions, as well as through 'soft' measures that focus on
  information provision and exchange.
- Many energy utilities, network providers and other stakeholders in the energy sector are already addressing adaptation needs. All market actors in the energy sector, including business associations, should consider strengthening climate resilience as an integral part of their business.

#### Chapter 1 – Purpose and scope of this report

This EEA report identifies the challenges of, and opportunities for, climate change adaptation and climate resilience in the context of a decarbonising energy system in Europe. It intends to support the efforts of the European Commission, national governments and non-state actors involved in planning, reporting, reviewing, implementing and revising relevant policies. The report provides information on the climate impacts and adaptation challenges associated with different energy technologies, gives an overview of the state of adaptation related to the energy system in Europe and presents good practice adaptation examples. The report concludes by identifying opportunities for further action by key adaptation actors and enablers in Europe.

# Chapter 2 – The European energy system now and in the future

The key driver for changes in the global and European energy system is the need for a clean energy transition that drastically reduces greenhouse gas emissions. The EU has adopted several quantitative targets related to the energy system in its 2030 climate and energy framework. The European Commission has proposed a strategy for a climate-neutral economy by 2050, including several long-term decarbonisation scenarios up to 2050.

The share of renewable energy sources in primary energy supply has more than tripled and their share in electricity generation has more than doubled since 1990. All global and European decarbonisation scenarios agree that these shares will continue to increase rapidly. Furthermore, the role of electricity as an energy carrier is increasing in all decarbonisation scenarios. These developments require the strengthening of electricity grids, enhancing the level of interconnection and increasing electricity storage.

The energy sector is a large user of water and land, both of which can be impacted by climate change. **The clean energy transition in Europe presents both opportunities and challenges for climate change adaptation.** On the one hand, replacing coal-fired power plants by photovoltaics and wind power radically reduces greenhouse gas emissions and water consumption, thus contributing to mitigation as well as adaptation in water-scarce regions. On the other hand, biofuels, and carbon capture and storage need more water and/or arable land than many conventional energy technologies.

# Chapter 3 – Climate change impacts on the European energy system

Anthropogenic climate change has already significantly affected the European climate, and further change is inevitable. The most important changes for the energy system include increases in mean and extreme air and water temperatures, and changes in annual and seasonal water availability, extreme climate-related events, and coastal and marine hazards.

Warming temperatures decrease energy demand for heating, but increase energy demand for cooling. They can also affect electricity generation and transmission, as well as fossil fuel extraction and transport. Water availability is generally projected to increase in northern Europe and decrease in southern Europe, but with marked seasonal differences. These changes can affect cooling water availability for thermal power plants, hydropower and bioenergy potential, river-borne fuel transport and energy demand for water provision. Climate change can also affect the potential for wind and solar power, but available projections are associated with significant uncertainty.

Many extreme weather events, including heat waves, heavy precipitation events, storms and extreme sea levels are projected to increase in frequency and/or magnitude as a result of climate change. Without appropriate adaptation measures, direct economic losses to the European energy system could amount to billions of euros per year by the end of the century, with much larger indirect costs. Climatic risks to the energy system can be further aggravated by the combination of different climatic hazards or by extreme events that affect several components of the energy system simultaneously.

Climate change impacts and related adaptation needs vary significantly across energy system components and European regions (see Map ES.1). Some impacts can be economically beneficial, such as reduced energy demand for heating. However, many impacts are adverse for the energy sector and/or society as a whole, such as reduced cooling water availability for thermal power plants in many regions and increasing risks for energy infrastructure from extreme weather events and sea level rise. Northern Europe experiences both beneficial and adverse impacts on its energy system, whereas southern European regions face overwhelmingly adverse impacts.

### Chapter 4 – Building a climate-resilient energy system

Building climate resilience comprises addressing the impacts of weather hazards on existing energy infrastructure and its operation, as well as considering the impacts of long-term climate change on newly planned infrastructure. Given the diversity of adaptation challenges across regions and energy system components, careful assessment of the relevant risks and options, as well as coordinated action by a wide range of public and private stakeholders, is necessary to ensure the clean energy transition is also climate-resilient.

There are both synergies and trade-offs between climate change adaptation, mitigation and wider sustainability objectives. The important connections between energy policy and other policy areas call for a comprehensive policy approach that considers multiple societal and policy objectives jointly.

Businesses are key actors in strengthening the climate resilience of the energy system. However, market actors in the energy system face a number of barriers that may impede the implementation of effective adaptation actions. Well-designed European and national policies can play a key role in overcoming these barriers.

Key EU climate and energy policies and strategies promote the mainstreaming of climate change adaptation into energy policies. These include the EU adaptation strategy, the Regulation on the governance of the Energy Union and climate action, the Commission proposal for a long-term strategy 'A Clean Planet for All' for a climate-neutral economy by 2050 and the Regulation on risk-preparedness in the electricity sector. The EU also supports building climate resilience in the energy system by requiring climate-proofing of major new energy infrastructure,



due to complex economic and environmental effects

Other energy sources and carriers (nuclear, electricity, heating and cooling)

funding relevant research and innovation projects, and developing climate services for the energy sector as part of the Copernicus services.

Almost all European countries have concluded a national climate change impact, vulnerability or risk assessment that covers the energy sector. Most countries also include energy as a relevant sector in their national adaptation strategies and/or plans. However, available government documents provide only limited evidence for the implementation of

adaptation actions in the energy sector. Individual countries are facilitating adaptation in the energy system by providing guidelines for vulnerability assessment and resilience planning, through support for the development of weather and climate services, and through reporting obligations for infrastructure providers.

Many energy utilities and network providers are already adapting their activities to the observed and projected impacts of climate change.

Some of these activities have been triggered or facilitated by government policies and regulation. Various international governmental and sectoral organisations are supporting these actions by providing guidance, developing information sharing platforms, facilitating communities of practice and acting as catalysts for the development of relevant services.

## Chapter 5 – Conclusions and outlook

Many asset owners and managers, as well as policy-makers at the EU, national and regional levels, and other stakeholders are already addressing adaptation needs in the energy system. However, some stakeholders are only beginning to acknowledge the relevance of climate change impacts on their activities, or they are experiencing barriers to taking action. More can and should be done to ensure that the energy system in Europe is climate resilient now and in the future.

The development of the Energy Union and the EU long-term strategy on climate action provide important opportunities for mainstreaming climate change adaptation in the planning and implementation of a decarbonised energy system in Europe through more coordinated actions, reporting and mutual learning among all involved actors. National (and sub-national) governments play an important role in facilitating climate change adaptation in the energy system. National climate change impact, vulnerability and risk (CCIV) assessments of the energy system, with a strong forward-looking component, are essential for making the clean energy transition climate-resilient. Countries that have not addressed energy as a priority sector or policy area in their national CCIV assessment are encouraged to do so in the future. Furthermore, all countries should consider the impacts of climate change on the current and future energy system in the development of their national climate and energy plans and long-term strategies under the Energy Union, and in the development and update of their national adaptation strategies and action plans.

Countries can facilitate building climate resilience in the energy system through regulation of energy markets and 'soft' measures that focus on information provision and exchange. Such activities can be supported by reporting requirements on climate change risks and adaptation actions for critical infrastructure providers, in particular, where such information is not available from other sources.

All market actors in the energy sector should consider strengthening climate resilience as an integral part of their business. Business associations in the energy sector can support their members in doing so.

# 1 Introduction

#### **Key messages**

- The overarching objective of this EEA report is to identify the challenges of, and opportunities for, climate change adaptation and strengthening climate resilience in the European energy system.
- All parts of the energy system, from energy production and transformation through to transmission, distribution, storage and demand, can be impacted by weather extremes and long-term climate change. It is crucial that these impacts are considered in planning the clean energy transition.
- The European Union is building the Energy Union, which aims to make energy more secure, affordable and sustainable. The goal of climate change adaptation in the energy system is to ensure that the goals of the Energy Union can also be achieved in a changing climate.

# 1.1 Purpose and content of this report

#### 1.1.1 Purpose and target audience

The overarching objective of this report is to identify challenges and opportunities for climate change adaptation and strengthening climate resilience in the context of a decarbonising energy system in Europe. The focus is on how policies at the European, national and — in some cases — subnational levels can support the transformation to a climate-resilient energy system now and in the future.

Since 2004, the EEA has regularly published reports on climate change impacts and vulnerability in Europe, addressing a broad audience (see EEA, 2017b for the latest report). This report is the second to address adaptation needs and opportunities for specific systems and sectors in Europe that are sensitive to climate change and play a key role in the decarbonisation of the economy. The first report addressed the transport system (EEA, 2014). A third report, to be published soon after this report, will address agriculture. Climate change impacts and adaptation will also be addressed in the forthcoming SOER report — The European environment — state and outlook 2020.

This report primarily targets European, national and subnational policymakers in the field of climate change and energy. Additional target audiences are relevant international organisations, regulators and standardisation organisations, business organisations and individual businesses from the energy sector (including producers, transmission system operators and traders). The information may also be relevant for investors, climate service providers, consultancies and researchers addressing the energy system, as well as civil society and the general public.

# 1.1.2 Multiple challenges facing the European energy system

The European energy system faces several important challenges. First, the energy sector is currently the largest emitter of greenhouse gases in Europe owing to its large reliance on fossil fuels. Therefore, the decarbonisation of the energy sector will play a central role in achieving a climate-neutral economy in Europe. The clean energy transition requires wide-ranging changes in how energy is produced, transported and used (EC, 2018j, 2018q). Many, but not all, decarbonisation measures can also reduce other environmental impacts, such as air pollution. Second, Europe's energy supply is highly dependent on imports from outside Europe, including from politically unstable regions. As a result, geopolitical tensions can threaten the security of the energy supply in Europe (EC, 2014). Third, modern societies and economies are increasingly dependent on a reliable energy supply, in particular with regard to electric power. Most economic and financial activities, transport, water supply and the provision of health services and disaster relief support rely on information and communication technologies powered by electricity. Therefore, even

short interruptions in electricity supply can lead to high economic and social costs.

In addition to these challenges, all parts of the energy system, from energy production (<sup>1</sup>) and transformation to its transmission, distribution, storage and demand, can be affected by weather and climate, including long-term climate change (see Figure 1.1 for selected examples). Gradual changes in the climate can affect the availability of important resources such as water for hydropower and for cooling thermal power plants. They can also affect energy demand, in particular in relation to heating, cooling and water supply. Weather extremes such as floods and storms can lead to blackouts due to flooding electric substations or windfall on power lines. Sea level rise can threaten coastal and offshore energy infrastructure.

The clean energy transition increases the need for the energy sector to consider climate variability and climate change due to the increasing share of climate-sensitive renewable energy sources (RES) and the stronger role of electricity as an energy carrier. Considering the important role of secure and affordable energy for European economies and societies, and the massive investments planned in the European energy system, it is crucial that the impacts of climate variability and change on the current and future energy system are considered in the clean energy transition.

#### 1.1.3 Policy context

The European Union (EU) is building the Energy Union, which aims to make energy more secure, affordable and sustainable. Several important pieces of legislation have already been adopted and others are currently in the legislative process. Of particular relevance is the Regulation on the governance of the Energy Union and climate action (EU, 2018d), which strengthens the coordination role of European institutions and creates new planning and reporting requirements for national governments. The underlying planning and reporting processes provide an opportunity to address climate change mitigation and adaptation in the energy sector in an integrated manner. The proposal for a revised regulation on risk-preparedness in the electricity sector (EU, 2018c) can play an important role in preventing, preparing for and managing electricity crises resulting from extreme weather events, even though climate change is not explicitly mentioned. Furthermore, the long-term strategy 'A Clean Planet for All' puts forward a vision for steering the EU towards a CO<sub>2</sub> emissions-free future in 2050 (EC, 2018j), in line with the objectives of the Paris Agreement (UNFCCC, 2015). In response to the challenge from climate change impacts, the EU adopted an EU adaptation strategy in 2013, which was evaluated in 2018 (EC, 2013b, 2018v). For further information on relevant EU policies, see Section 4.2.



#### Source: EEA, adapted from Troccoli (2018).

<sup>(&#</sup>x27;) From a physical perspective, energy can be neither produced nor consumed; it can only be transformed from one form (e.g. solar radiation) to another (e.g. electric power). Nevertheless, international energy statistics use the term 'energy production' and 'energy consumption' to describe usable energy entering or leaving the economy. In the interest of coherence with these established terminologies, this report uses the terms 'energy production' and 'consumption' instead of the more exact terms 'energy supply' and 'energy use'.

## 1.1.4 Outline

This report is structured as follows.

**Chapter 1** describes the context, purpose and scope of this report. It also defines key terms used throughout the report.

**Chapter 2** provides an overview of the current status of the energy system in Europe, including its interaction with other systems and sectors. It also discusses how global and European policies and other drivers are expected to affect the energy system in the future, and includes relevant scenarios.

**Chapter 3** gives an overview of past and projected climate change across European regions, and describes the main impacts of these changes on the European energy system. The chapter concludes with a summary of the main adaptation needs and opportunities for each energy system component and related adaptation options.

**Chapter 4** discusses the adaptation challenge of the European energy system from an actor's perspective. It starts with an overview of the main types of adaptation action and the key actors involved, and then discusses the role of governments in facilitating adaptation in the energy system. The remainder of the chapter is devoted to a review of the main adaptation policies and actions by the EU, national governments, international organisations and selected other actors. The chapter concludes by presenting adaptation case studies from various infrastructure providers from the European energy sector.

**Chapter 5** presents the main policy-relevant conclusions of this report.

# 1.2 Scope of the report

## 1.2.1 Thematic scope

This report looks at the energy system and focuses on climate change adaptation and strengthening climate resilience (see Section 1.3 for a definition of these terms). Considering that climate change mitigation and adaptation in the energy system need to be addressed jointly, this report also addresses the synergies (i.e. where mitigation activities also have adaptation benefits, or vice versa) and trade-offs (i.e. where an adaptation action increases the mitigation challenges, or vice versa) between these two policy objectives. However, the focus of this report is on the additional challenges created by a changing climate.

This report covers the whole energy system, from primary energy production to energy use, and includes all energy sources and carriers. The definition of the energy system applied here includes the 'traditional' energy sectors, such as oil and gas extraction, transport and distribution; electricity generation (from all sources), transformation, transmission, storage and distribution; and heating and cooling production and distribution. However, it also comprises activities and actors outside the 'traditional' energy sector that can play an increasing role in a decentralised energy system, such as biomass production and transport for energy production, and electricity production and storage by individuals. Particular attention is given to components and technologies that are highly sensitive to climate change and variability and/or that are expected to become more important in the context of the clean energy transition. As a result, there is a strong focus on low-carbon technologies and on the electricity system. Energy use for various purposes is considered only insofar as the demand is climate sensitive (in particular heating and cooling) and may require adaptation actions within the energy sector. The use of energy carriers for non-energy purposes is outside the scope of this report.

This report applies the following categorisation of the energy system into components (IEA, 2016b):

- 1. **Primary energy production:** the extraction or production of energy resources as inputs to energy transformation processes. It includes the extraction of the main fossil energy sources (coal, oil and natural gas) and of uranium. It also includes renewable energy sources such as hydropower, bioenergy, solar power and wind energy.
- 2. Energy transformation: the transformation of energy resources into secondary energy carriers. This includes electric power and heat generation technologies, including renewable (e.g. biomass-based) and fossil energy, and fuel refining.
- 3. Transport, transmission, storage and distribution: this component comprises power grids, gas and oil pipelines and their supporting infrastructure, such as storage facilities, substations and logistical assets. It further includes pumped hydropower storage, batteries and other storage technologies. It also includes transport of fuels and carbon capture and storage.
- 4. **Energy demand:** this covers the use of energy in buildings, households, industry, businesses and transport (i.e. the power, heating and cooling

demand of all consumers). The non-energy use of energy sources (e.g. oil use for plastic production) is outside the scope of this study.

An overview of key actors and enablers that play a role in climate change adaptation in the energy system is provided in Section 4.1.2.

The European Atomic Energy Community (also known as Euratom) is legally separate from the EU, and issues relating to nuclear safety are outside the mandate of the EEA. Adaptation challenges that nuclear power plants have in common with other thermal power plants (e.g. cooling water availability) are addressed in this report. In contrast, potential adaptation challenges relating to the safety of operating nuclear power plants and to nuclear waste disposal are not addressed in this report.

# 1.2.2 Geographical scope

The geographical scope of this report comprises the 33 member countries of the EEA. They comprise the 28 EU Member States (as of May 2019) as well as lceland, Liechtenstein, Norway, Switzerland and Turkey. The EEA's Western Balkan cooperating countries are not explicitly addressed in this report. However, they are strongly interconnected with EU power and gas networks, and many of the findings in this report also apply to them. Upstream activities (e.g. oil and gas extraction) outside Europe are beyond the scope of this report, but the import of energy carriers is briefly discussed, as it may generate adaptation needs for infrastructure located within Europe (e.g. pipelines or sea terminals).

This report presents information for the largest group of countries for which this information is readily available. As a result, there is some variation in the geographic coverage between and within chapters. For example, Section 2.1 relies strongly on energy statistics from Eurostat, which are most readily available for EU Member States. Similarly, Section 4.2 focuses on EU policies, which are not directly applicable to the non-EU member countries of the EEA. In contrast, Section 2.2 relies largely on the EU long-term strategy 'A Clean Planet for All', which covers several countries that are neither EU Member States nor EEA member countries.

There are considerable differences across Europe in climate change impacts as well as in the physical assets, ownership and management of energy infrastructure. These differences in adaptation needs and opportunities require adaptation policies, strategies and actions that are tailor-made for a specific country or region. To support regionally specific adaptation planning, Chapter 3 analyses climate change and its main impacts on the energy system in seven European regions. Considering the particular focus of this report on electricity supply and demand, the **regional country groupings** used in Chapter 3 are based on the regional wholesale electricity markets of the EU (including Switzerland) (EC, 2018u). The non-EU member countries of the EEA were added to those regions to which they are most closely connected. The island countries of Cyprus and Iceland, which are not connected to the European electricity grid, were added to the regions with the most similar climatic conditions. The resulting regionalisation is as follows:

- 1. Northern Europe: Denmark, Estonia, Finland, Iceland, Latvia, Lithuania, Norway, Sweden.
- 2. British Isles: United Kingdom, Ireland.
- 3. Central western Europe: Austria, Belgium, France, Germany, Liechtenstein, Luxembourg, Netherlands, Switzerland.
- 4. Central eastern Europe: Croatia, Czechia, Hungary, Poland, Romania, Slovakia, Slovenia.
- 5. Iberian Peninsula: Spain, Portugal.
- 6. Apennine Peninsula: Italy, Malta.
- 7. South-eastern Europe: Bulgaria, Cyprus, Greece, Turkey.

Chapter 4 gives a brief overview of the national-level adaptation policies relating to energy; it also presents various examples of policies and actions at national and subnational level. However, a systematic assessment of adaptation needs and actions in the energy system for all countries is beyond the scope of this report.

# 1.3 What do adaptation and climate resilience mean?

Different communities of policymakers, practitioners and scholars use somewhat different terms for managing the impacts of climate variability and change. The United Nations Framework Convention on Climate Change (UNFCCC) (UN, 1992), as well as EU climate change policy, distinguishes two fundamental policy options to limit the risks of climate change for societies: mitigation and adaptation. Energy policies and practice commonly use the term climate resilience

#### Box 1.1 Definitions of key terms used in this report

Mitigation (of climate change): 'a human intervention to reduce emissions or enhance the sinks of greenhouse gases.'

Adaptation (to climate change): 'In human systems, the process of adjustment to actual or expected climate and its effects, in order to moderate harm or exploit beneficial opportunities.'

**Maladaptation:** 'actions that may lead to increased risk of adverse climate-related outcomes, including via increased GHG emissions, increased vulnerability to climate change, or diminished welfare, now or in the future. Maladaptation is usually an unintended consequence.'

**Hazard:** 'the potential occurrence of a natural or human-induced physical event or trend that may cause loss of life, injury, or other health impacts, as well as damage and loss to property, infrastructure, livelihoods, service provision, ecosystems and environmental resources.'

**Vulnerability:** 'the propensity or predisposition to be adversely affected. Vulnerability encompasses a variety of concepts and elements including sensitivity or susceptibility to harm and lack of capacity to cope and adapt.'

**Risk:** 'the potential for adverse consequences where something of value is at stake and where the occurrence and degree of an outcome is uncertain.'

**Resilience:** 'the capacity of social, economic, and environmental systems to cope with a hazardous event or trend or disturbance, responding or reorganizing in ways that maintain their essential function, identity, and structure, while also maintaining the capacity for adaptation, learning, and transformation.'

Source: IPCC (2018).

**Climate-resilient infrastructure** 'is designed, built and operated in a way that anticipates, prepares for, and adapts to changing climate conditions. It can also withstand, respond to, and recover rapidly from disruptions caused by these climate conditions.'

Source: OECD (2018).

for reducing the negative impacts of climate-related hazards, which is closely related to adaptation. Box 1.1 provides definitions of these and other relevant terms.

Although the original concepts and definitions of the terms 'adaptation' and '(climate) resilience' differ, they are now often used in connection with each other (e.g. EC, 2018q). In this report, the expressions 'adaptation to (climate change)' and strengthening 'climate resilience' are used broadly, and largely interchangeably, for all efforts that address climate-related challenges to the energy system, relating to both short-term climate hazards and long-term climate change. It should be stressed that adaptation and resilience in the energy system not only refer to the design and building of infrastructure but also include the management and operation of energy infrastructure and the underlying policies and institutions. Further information on different types of adaptation is provided in Section 4.1.

The goals of climate change adaptation and resilience building in the energy system are to ensure a secure and affordable energy supply now and in the future while supporting the clean energy transition and other societal objectives (EC, 2018c). More specifically, adaptation aims to ensure minimal disruption to energy production, transformation, transport and consumption in the long term, while also protecting the value of public and private investments in the sector and the interests of society and the economy overall (Boston, 2013). It can also address risks to public safety that could originate from energy infrastructure during extreme climate conditions, such as sparking wildfires (see Box 3.1). Failing to adapt energy systems to climate change could result in severe consequences such as blackouts, which have a significant impact on modern economies that are reliant on stable access to energy.

The energy system has strong links with other systems and sectors due to its use of water and land (see Section 2.1.5 for further information on the energy-water-food nexus). Therefore, a sustainable energy supply should minimise conflicts, and exploit synergies, with other relevant policy areas, such as water and flood management, biodiversity protection and agricultural policy. Adaptation actions with substantial negative impacts on other policy goals or societal objectives are often referred to as maladaptation, which should be avoided. Synergies and trade-offs between adaptation, mitigation and other policy objectives are discussed further in Section 4.1.4.

# 1.4 Development of this report

This report has been developed by a team of authors from the EEA, two of its European Topic Centres, and external consultants engaged under a framework contract with the EEA. The key methods used were literature review, and stakeholder and expert consultation.

**Literature review** has been central to all parts of the report. This report draws on numerous publications by international organisations, EU institutions, national governments, sector associations, academic research institutions and other relevant organisations.

**Stakeholder and expert consultation** has also been a key element in the development of this report. The EEA convened a stakeholder group for this report, which comprised representatives from international organisations, the European Commission, national governments, regulators, sector associations and infrastructure operators from the energy sector, and research projects. These stakeholders were first invited to comment on the annotated outline of the report and to suggest additional literature sources. In September 2018, the EEA organised a stakeholder workshop where the scope and content of this report was discussed further. Finally, the stakeholders were invited to provide comments on the draft report in parallel with the European Environment Information and Observation Network (Eionet) review.

Adaptation case studies in this report show examples of how energy producers and infrastructure providers throughout Europe have addressed climate-related challenges and opportunities in their planning and operation. Section 4.5 of this report includes five in-depth case studies that are based on literature review and interviews with relevant stakeholders. They have also been published on the European Climate Adaptation Platform (Climate-ADAPT) (<sup>2</sup>), a data and information-sharing platform co-managed by the Commission and the EEA. Shorter illustrative case studies, based on literature review only, are included as text boxes in the main text.

<sup>(2)</sup> https://climate-adapt.eea.europa.eu

# 2 The European energy system now and in the future

#### **Key messages**

- The key driver for changes in the global and European energy system is the need for a clean energy transition that drastically reduces greenhouse gas emissions. The EU has adopted several quantitative targets related to the energy system in its 2030 climate and energy framework. The European Commission has proposed a long-term strategy that includes several long-term decarbonisation scenarios up to 2050.
- Energy supply in Europe is still dominated by fossil fuels, the majority of which are now imported. However, the share of renewable energy sources in primary energy supply has more than tripled and their share in electricity generation has more than doubled since 1990. All global and European decarbonisation scenarios agree that these shares will continue to increase rapidly.
- The energy sector is a large user of water and land. The interdependencies represented by the energy-water-land nexus are expected to intensify due to climate change impacts and an increasing share of renewable energy sources. Some low-carbon energy technologies have the potential to contribute positively to this nexus, whereas others may increase competition for scarce water and/or land resources.
- The interconnection between European electricity grids has increased in recent years. Electricity storage is becoming increasingly important for managing the growing share of intermittent renewable energy sources in electricity generation. The role of electricity as an energy carrier, and of electricity grids and storage, is further increasing in all decarbonisation scenarios.
- Final energy consumption in Europe has remained largely constant since 1990. The decarbonisation pathways in the EU long-term strategy place a strong focus on decreasing overall energy demand through increasing energy efficiency and, possibly, changes in consumer behaviour.

The energy sector is of high economic importance in Europe. According to provisional estimates, it comprises more than 110 000 enterprises, employs around 1.6 million people and has a turnover of around EUR 1.9 trillion along the supply chain in the EU (Eurostat, 2018c). However, the relevance of the energy sector for overall society is much greater than these numbers suggest. Energy enables a multitude of services, including heating and cooling, lighting, telecommunication, information technology and transport. Energy has therefore been characterised as the 'lifeblood' of modern societies.

This chapter outlines the current energy system in Europe, as well as scenarios for its future development, which are driven to a large degree by the need for a clean energy transition.

## 2.1 The European energy system

This section examines the key components of the European energy system. It provides a

descriptive and quantitative characterisation of each component and the most relevant developments. It also reviews the interaction of the energy system with other sectors. This report applies the Eurostat terminology of energy aggregates (see Box 2.1), and this section relies largely on Eurostat energy balances data. Most of these data are also available for three of the five non-EU member countries of the EEA — Iceland, Norway and Turkey — but not for Switzerland and Liechtenstein. However, some statistics (in particular those relating to the import of energy carriers) are available only for the 28 EU Member States (EU-28), and data reporting from non-EU countries is sometimes delayed. For reasons of consistency and to avoid any confusion about which countries are covered in a particular figure or statement, all quantitative data in this section refer to the EU-28. Relevant information on particular circumstances in non-EU member countries of the EEA is provided in the text.

#### Box 2.1 Main energy aggregates according to Eurostat

**Primary energy:** the first energy form in the production process for which various energy uses are in reality practised. Eurostat's methodology is based on the physical energy content method. For directly combustible energy products (e.g. coal, crude oil, natural gas, biofuels, waste) it is their actual energy content measured by their net calorific value. For products that are not directly combustible, the application of this principle leads to the choice of heat as the primary energy form for nuclear, geothermal and solar thermal power and to the choice of electricity as the primary energy form for solar photovoltaic, wind, hydro, tide, wave and ocean power.

The choice of the physical energy content method for determining the primary energy equivalent of fuels can give a distorted picture of the relative importance of different energy carriers. For example, in a hypothetical scenario in which 50 % of electricity is produced from nuclear fuels and 50 % from wind energy, nuclear energy would be described as having a primary energy share of 75 % and wind as having only a 25 % share. Hence, the importance of most renewable energy sources in the overall energy system is under-represented by their primary energy equivalent.

**Primary production:** this refers to any kind of extraction or exploitation of energy products from natural sources. For fossil and other fuels, it refers to the extraction from the environment. It also includes electricity and heat according to the choice of the primary energy form (electricity generation using hydro, wind and solar photovoltaic power).

**Final energy consumption:** the total energy consumption in all end-use sectors (including industry, transport, services and others). By and large, it accounts for the amount of energy delivered to final consumers.

**Source:** Adapted from EU, 2008b; Eurostat, 2018e.

#### 2.1.1 Primary energy supply

Figure 2.1 shows the development of the primary energy supply in Europe (EU-28) over time for different energy carriers, distinguished by domestic production and net imports. Eurostat data do not distinguish between domestic production and imports of nuclear fuel. According to data from the Euratom Supply Agency, the share of domestic uranium production in overall uranium supply has been consistently below 10 % since 1995 (ESA, 2019, Indicator 5).

Europe relies on a broad range of energy sources. Fossil fuels still dominate the primary energy supply in Europe (74 % in 2017). However, the share of renewable energy sources (RES) in the primary





Notes:Electricity imports are not shown in this figure because net extra-EU energy imports were negligible. Nuclear fuel is mostly imported.Source:EEA, based on data from Eurostat (2019b).

energy supply of the EU-28 has grown significantly, from 4 % in 1990 to 14 % in 2017. Oil, used primarily as a transport fuel, and natural gas remain the most important energy sources in Europe, accounting for 36 % and 23 % of the total primary energy supply in 2017, whereas the share of coal declined from 26 % in 1990 to 13 % in 2017. The share of nuclear energy remained more or less constant at 12 %. The interpretation of these numbers should consider the limitations of the physical energy content method used to determine the primary energy content of different energy carriers (see Box 2.1). Further information on the increasing role of RES for electricity production is provided in the next section.

The domestic production of fossil fuels in Europe has decreased, from 39 % of the EU primary energy supply in 1990 to 18 % in 2017. Most of the decrease is due to declining coal production, but Poland and Germany, among other countries, still have significant coal production. Extraction from North Sea natural gas fields is carried out by the Netherlands, Norway (not included in Figure 2.1) and the United Kingdom. Norway and the United Kingdom are also significant oil producers in Europe.

**Europe's energy supply is strongly dependent on imported fuels.** The share of imported fuels in the primary energy supply of the EU-28 increased from 44 % in 1990 to 56 % in 2017, with most of the increase occurring before 2007. The large decline in the production of domestic fossil fuels was compensated in almost equal parts by increasing shares of imported fossil fuels (from 44 % to 55 %), mostly from increasing gas imports, and of domestic RES (from 4 % to 13 %). The share of imports would be lower if Norway were included, as it is an important producer of oil and gas and also an important source of imports into the EU. More than 80 % of the EU's natural gas imports originate from four countries (Russia, Norway, Algeria and Qatar). Oil and coal imports come from a more diverse range of countries (Eurostat, 2018c). In recent years, the EU has spent about EUR 1 billion every day on energy imports, with considerable fluctuations depending on the price of crude oil (Eurostat, 2019c). The annual spending on energy imports to the EU amounts to approximately 2 % of the total gross domestic product.

## 2.1.2 Energy transformation

#### Electricity generation

Figure 2.2 shows the development of gross electricity generation in Europe (EU-28) over time for different energy sources.

**Electricity generation in Europe remains dominated by fossil fuel-based thermal power plants, but their share has dropped** from 57 % in 1990 to 44 % in 2017. The share of nuclear power has also declined, from



Figure 2.2 Gross electricity generation in the EU-28

Source: EEA, based on data from Eurostat (2019b).

31 % in 1990 to 25 % in 2017. However, nuclear power still plays an important role in electricity generation in half of the EU-28. France alone is responsible for around half of the nuclear power production in the EU. Nuclear power is being phased out in Germany.

Electricity generation from renewable sources has grown substantially in the EU, from 13 % in 1990 to 31 % in 2017. The share of hydropower has remained largely constant over time, in the range of 10-13 %. Wind and solar power have grown considerably, from almost nothing in 1990 to 11 % and 4 % of the power supply, respectively, in 2017. That was the first year in which wind power production exceeded hydropower production in the EU. The share of biomass in power production has increased from less than 1 % in 1990 to more than 5 % in 2017. The share of RES in general would be considerably higher if data from the non-EU member countries of the EEA, where hydropower is often the dominant source of electricity generation, were included.

The growth in renewables, in particular wind and solar power, is the major development in the European power system. In 2016, renewables accounted for 86 % of the newly installed electricity generation capacity in Europe (EEA, 2017e). This growth is expected to continue as policies focus on decarbonisation and on promoting renewable energy production (see Section 2.2). Further increases in the growth in renewable generation capacity are expected as onshore and offshore wind power and utility-scale photovoltaics (PV) increase their cost-competitiveness compared with electricity generation from fossil fuels (IRENA, 2018c).

#### Heating and cooling

Heating and cooling represent an important share of Europe's energy consumption. Energy consumption for heating is clearly dominant in most countries. It comes from a variety of sources, including solid, liquid and gaseous fossil fuels and biofuels, electricity and solar thermal energy. Energy consumption for cooling is much lower in most countries, but it is increasing due to socio-economic changes and climate change.

The EU's **heating and cooling strategy** was launched in 2016 (EC, 2016e). Significant progress has been made across Europe in terms of increasing the share of RES in the energy mix for heating and cooling. However, the majority of heating and cooling energy is still generated from fossil fuels (66 % in total, 42 % from natural gas). Electricity and district heating account for 21 %, which is mainly based on fossil fuels. Biomass is used in about 12 % of heating and cooling production, notably in Austria, Finland and Sweden. Solar thermal

and geothermal power, and heat pumps are still marginal in most European countries. However, biogas and heat pumps have had high annual growth rates since 2005 (Eurostat, 2018f; Camia et al., 2018; Heat Roadmap, 2019).

The decarbonisation of industrial and domestic heating processes still presents a challenge (Steinbach et al., 2017). Meeting decarbonisation targets will require a combination of energy efficiency improvements (including insulation in buildings) and new or innovative heating and cooling solutions. These include combined heat and power generation, heating and cooling from renewable sources, power-to-heat and/or district heating and cooling. Suitable solutions will differ across regions, partly reflecting regional variations in heating and cooling demand.

# 2.1.3 Transportation, transmission, storage and distribution

Energy is provided to final consumers through transmission and distribution networks or transport services (e.g. for solid fuels). These networks and services are reinforced and supported by storage facilities and technologies. This infrastructure provides the backbone of Europe's energy system, linking energy supply and demand. Energy transport, transmission, storage and distribution generally involves large-scale infrastructure that can be disrupted by extreme weather events related to climate change. Damage to infrastructure can cause long-lasting and costly supply disruptions, such as power outages, as well as costly equipment repairs or replacement.

**European electricity grids have increased their level of interconnection in recent years.** There are plans to further enhance cross-border electricity transmission capacity — most notably through the EU's Trans-European Networks for Energy (TEN-E) Regulation and strategy (EC, 2018w; EU, 2013b) and the projects of common interest (EC, 2019d). The TEN-E Regulation has recently been evaluated, and it would be repealed by the proposed Regulation establishing a Connecting Europe Facility (Rademaekers et al., 2018; EC, 2018s).

In 2014, the European Council formulated the objective that EU Member States should have a cross-border electricity transmission capacity that is equal to at least 10 % of their domestic electricity generation capacity by 2020 (EC, 2018l). This target was later raised to 15 % by 2030 (EU, 2018d). Seventeen Member States are on track to reach the targets, while Cyprus, Poland, Spain and the United Kingdom are not expected to meet the interconnection target for 2020 (EC, 2017f). A more interconnected EU grid can increase both vulnerability and resilience, depending on the particular situation and the management of the overall system.

Recent developments in EU energy market legislation include network improvements to support the connection of renewable energy generation to local distribution grids (EC, 2016b, 2018r). Smart grids and other innovations play a role in increasing the resilience of electricity networks and the integration of renewable energy (Schaber et al., 2012; Becker et al., 2014; EDSO for Smart Grids, 2018). Innovations in the area of offshore wind energy include the expansion and improvement of long-distance transmission networks through the development of specialist insulation and superconducting materials for subsea cables.

The European **gas network** can be subdivided into long-distance transport pipelines and distribution grids. Long-distance gas transport is strongly influenced by Europe's dependence on gas imports from Russia. The EU is aiming to reduce its dependence on Russian gas for political and strategic reasons. It has started to do this with the construction of reverse-flow pipelines (in which gas can flow in both directions, thereby increasing capacity) and by increasing connectivity between Member States. However, Germany is currently planning to expand import capacity from Russia through the Nord Stream 2 project (Nord Stream 2, 2018).

Liquefying and transporting natural gas (liquefied natural gas or LNG) has had a large impact on the global gas market in recent years, by enabling significant natural gas imports from places such as Algeria and Qatar. This also reduces EU import dependency on specific gas pipelines, thereby increasing overall system resilience. Various Baltic Sea states have plans to build new LNG storage terminals. The shale gas boom in the United States provides another source of LNG imports, at least as long as current long-term market prospects remain the same. Conventional gas and LNG terminals and refineries are largely based in coastal areas. Their infrastructure and operation is therefore vulnerable to storms, storm surges and sea level rise.

**Oil infrastructure** in the EU is largely owned by private energy companies. A large number of oil refineries in Europe are located on coasts where they are vulnerable to storm surges and wave activity. Pipelines and associated oil infrastructure may also be vulnerable to strong winds.

**Energy storage** is an increasingly important part of the energy system. Traditionally, this has focused on facilities for the **storage of oil**, as all EU Member States are obliged to have oil stocks equivalent to 90 days of oil consumption. Norway, Switzerland and Turkey are bound by similar rules as Member States of the International Energy Agency (IEA). This obligation stems from the oil shocks of the 1970s and is intended to prevent supply disruptions. **Natural gas storage** is not regulated in the same way as oil storage (i.e. there are no quantitative requirements for emergency stocks). Gas storage is vital to guarantee the continuity of gas supply in winter when demand is highest, thereby contributing to security of supply.

Electricity storage improves the resilience and flexibility of power supply. This is becoming increasingly important to network operators that need to manage the increasing share of intermittent RES in electricity generation. The need for electricity storage capacity is expected to boom in coming years (EC, 2013c). Pumped hydroelectricity has been the main form of large-scale electricity storage employed in Europe. It has generally been used for electricity generation during daily peak hours and pumping during low-demand nightly hours in thermal electricity-dominated systems. In Norway, however, electricity needs have been covered by hydropower with sufficient reservoir capacity to meet demand in cold winters. This capacity has also led to studies on how Norwegian large reservoir hydropower can deliver balancing capacity across Europe, and on the technical, economic and environmental challenges that are linked to the operation of hydropower plants for long-term storage. There remains significant potential to develop pumped hydropower in Norway and elsewhere if the business case is sufficiently strong (EASE/EERA, 2017). For variation over hours, large-scale battery storage has very recently started to be deployed successfully as the cost of batteries declines and the technology improves. This technology will compete with the traditional pump storage hydropower that has reservoir capacity for a few hours of operation only. The growth in the use of electric vehicles and residential batteries, combined with smart devices and contracts, has the potential to greatly increase the availability of electricity storage and consequently to improve system resilience. However, such growth may be limited with current battery technologies as a result of resource limitations including cost, space and rare minerals. Hydrogen storage is just starting to become part of the energy system, but it may experience substantial growth in the future.

Heating and cooling storage in underground aquifers is receiving increasing attention as a way to optimise the thermal conditions of buildings. Several such systems have been installed in the Netherlands, where it is becoming a standard for new buildings; it has a large potential for application in other countries as well (Bloemendal and Jaxa-Rozen, 2018). The vulnerability of storage technologies to climate change impacts varies by type. The vulnerability of above-ground storage facilities is similar to that for other key fossil fuel infrastructure. Subterranean storage infrastructure should be largely unaffected. Pumped hydropower storage needs to be distinguished by technology. Closed-loop pumped storage plants are rather insensitive to climate variability and change, whereas open-loop plants have similar susceptibilities as run-of-river and reservoir hydropower systems. The novelty of other storage technologies presents some uncertainties regarding their potential climate vulnerabilities, although temperature is known to affect the effectiveness of batteries.

### 2.1.4 Energy demand

Energy demand is the fundamental driver behind the energy system. Aggregate energy demand in Europe appears to have stabilised in recent years, but important shifts in demand are still taking place between and within sectors. Energy demand is also affected by climate change, primarily through changes in ambient temperature, which determine the demand for heating and cooling in buildings.

#### Final energy consumption

Figure 2.3 shows the development of final energy consumption in Europe (EU-28) by sector.

Final energy consumption in Europe (EU-28) has remained largely constant since 1990. It mostly

grew until 2006, then declined until 2014, and started increasing again afterwards. Industry is the only sector with an overall declining trend in energy use. Most of the energy savings in industry are due to increasing energy efficiency and structural change towards less energy-intensive industries (Voigt et al., 2014). This includes the relocation of carbon-intensive production outside Europe, also known as carbon leakage (Paroussos et al., 2015). The recent economic crisis also played a role by reducing industrial output and thereby the total energy demand (EC, 2017b). Energy use by households does not show a clear trend, and energy demand in the services and transport sectors is still increasing. Rising energy demand in the services sector can be explained by economic growth and structural change. The growth of energy use in transport is primarily driven by an increase in air travel and road freight transport.

### Fuel mix in the transport sector

The transport sector is still strongly dominated by oil use. Oil accounts for 94 % of the sector's energy consumption, and the sector accounts for 80 % of the total oil consumption in the EU. Alternative fuels, particularly liquid biofuels, liquefied petroleum gas and natural gas have emerged as a small but increasing share of the transport energy mix since 2006. Driven by biofuel targets in the Renewable Energy Directive, the share of biofuels in the energy mix of the road transport sector has increased from a negligible level in the early 2000s to 3.5 % in 2016 (Eurostat, 2018b). Electricity has remained a small





part of the transport fuel mix. Sales of hybrid and electric vehicles have grown in the past decade, but they still represent only around 1.5 % of all new vehicles sold in the EU in 2017. In Norway, their share is already more than 30 % (EEA, 2018b). Current and expected growth in electric vehicle uptake across Europe will significantly increase the share of electricity in the energy mix of the transport sector.

### Heating and cooling demand

Space heating and cooling account for roughly half of the total final energy demand in the EU (Heat Roadmap, 2019). A large share of the energy consumption in the built environment and in industry is for heating purposes. The use of low-temperature heat for heating in the built environment accounts for 27 % of the total final energy consumption in the EU. Gas is the primary fuel used for this purpose. An additional 16 % of the final energy consumption is used for process heating in industry (Heat Roadmap, 2019). Process heat is primarily produced from natural gas (39 %), coal (17 %) and other fossil fuels (19%); the remainder comes from renewable sources. For high-temperature process heating, hardly any renewable sources are used. Cooling, whether for industrial processes or for use in buildings, is currently a relatively small part of final energy demand. Cooling demand in the EU has undergone a large increase, from 63 TWh in

Figure 2.4 Energy-water nexus



Source: EEA, based on EP (2012).

1990 to 152 TWh in 2016, but it still accounted for only 1.2 % of the energy use in buildings in 2016 (IEA, 2018a).

### 2.1.5 The energy-water-land nexus

The energy sector uses significant amounts of water and land. Hence, it may be in competition with other sectors and systems that use these resources, such as agriculture and natural ecosystems. These links are important considerations in the development of energy strategies.

#### Energy-water nexus

The energy sector is a large user of water both globally and in Europe. Consumptive water use by the energy system can directly compete with other water users, such as agriculture and industry. Non-consumptive water use can have substantial ecosystem impacts, in particular when thermal power plants discharge cooling water back to the environment at above ambient temperature or when storage hydropower plants can affect the seasonality of river flows. At the same time, the water sector is a major energy user, accounting for around 4 % of global electricity consumption (IEA, 2018b). The term 'energy-water nexus' is used to describe the interdependence between these two key systems underpinning economic and social development (see Figure 2.4).

The interdependencies represented by the energy-water nexus are expected to intensify over time as a result of climate change impacts, changing consumption patterns and population growth (Eurostat, 2018d). The decline in thermal power generation as envisaged in most decarbonisation scenarios is expected to lead to reduced water withdrawals but higher water consumption by the energy sector in most advanced economies (Fricko et al., 2016). Several studies have pointed out that increasing water stress can constrain energy production in (parts of) Europe (see Section 3.3). Some low-carbon energy technologies have the potential to contribute positively to this nexus, whereas others may lead to competition or conflict. For example, replacing thermal power generation with solar PV and wind energy reduces freshwater use per unit of energy generated, whereas water use for energy crops and for concentrated solar power (CSP) can be higher than for fossil fuel power plants (see Box 2.2). Therefore, an energy strategy focusing on bioenergy or CSP can increase competition with other water-using sectors (Berrill et al., 2016).

#### Box 2.2 Renewable energy sources and water use

The expansion of renewable energy can create synergies at the energy-water nexus, if appropriate RES are chosen. In many regions in southern Europe, water needs for agriculture, industry, energy, tourism and household consumption are increasing while water availability is decreasing due to climate change. In this situation, replacing fossil fuel power plants with renewable energy sources with low water use, in particular wind power and solar photovoltaics, can be a win-win strategy to decrease water scarcity. Figure 2.5 shows that as energy production from renewables in Spain and Greece has increased, freshwater abstractions for power production has decreased. Obviously, the relationship between the two variables depends on the particular choice of renewable energy sources.





Source: Based on data from Eurostat (2018a).

Concentrated solar power (CSP) produces electricity by concentrating solar irradiation to heat a gas or liquid, which is in turn used to drive an electrical generator. CSP is one of the most water-intensive energy production technologies (Frisvold and Marquez, 2014) because of its reliance on large quantities of water for cleaning and cooling the mirrors used in installations. This can lead to scarcity of water resources, particularly in arid regions where such facilities are likely to be situated. While CSP presents a potentially important energy source, its water intensity illustrates significant challenges for the energy-water nexus. Efforts are being made to improve the technology and reduce its water use, for example through the EU-funded MinWaterCSP project. This project is developing solutions such as new cooling systems, heat transfer surfaces and mirror cleaning techniques that aim to reduce CSP water usage. Such advances in technology could reduce water evaporation losses by up to 95 % (INEA, 2018). The energy-water nexus can also influence the choice of cooling technology for power generation. Once-through cooling systems are the cheapest and most efficient cooling systems, but they require the highest water withdrawals. Dry cooling is the most expensive and least efficient technology, but it uses the least amount of water. Wet cooling towers have intermediate costs and efficiency compared with the first two options. Their water withdrawals are lower than for once-through cooling systems, but their consumption is higher, which can introduce important trade-offs. As a result, the level and type of water stress in a region can play an important role in the choice of the most suitable cooling technology (IEA, 2018b).

Energy is important for the provision of water for a wide range of uses that are crucial for the health and well-being of the population. These uses include the provision of safe and clean drinking water, water desalination, wastewater treatment, and water pumping and distribution for all uses (IEA, 2018b). Failures in the energy system, whether linked to water shortages or not, have the potential to disrupt these vital functions.

The water sector in Europe may be able to reduce its energy use in the future. Some technologies, such as wastewater treatment with energy recovery or biogas production, may even generate more energy than they use (IEA, 2018b). At the same time, energy use may increase in the most water-stressed regions of Europe, where desalination is increasingly used (see Section 3.3.4).

The important system implications of the energy-water nexus are beginning to be taken into account by policymakers (see also Section 4.1.4). Further information on managing this nexus is available in various recent publications from international organisations and academic researchers (e.g. IEA, 2014, 2016c; IRENA, 2015; Pittock et al., 2015).

## Energy-land nexus

The term 'energy-land nexus' describes the interaction between energy and other land uses, such as food production and nature protection. **The energy-land nexus is of growing importance, because many RES have a large land footprint.** The land requirement is largest for bioenergy, which relies on plant sources for the generation of energy. The common practice of bioenergy provision from crops is likely to reduce the land available for food production. Innovative initiatives attempt to remedy conflicts between energy and other land uses. For example, agro-photovoltaics (the co-production of food and energy) can significantly increase land use efficiency (Fraunhofer ISE, 2017).

Available studies have produced very different estimates for the land use requirements of renewable versus conventional energy technologies. One recent review study found that various RES require about 1-4 orders of magnitude (i.e. factor 10 to 10 000) more land than fossil and nuclear energy sources (van Zalk and Behrens, 2018). However, studies including life-cycle analysis and/or impacts over time come to very different conclusions. A comprehensive review by the International Resource Panel found that the land impacts of RES can be both smaller or larger than those of conventional technologies (UNEP, 2016). One study from the United States has suggested that the cumulative land use impact of fossil energy technologies exceeds that of solar-, wind- and hydropower after a few years or decades (this is even faster for surface coal mining) (Trainor et al., 2016). The only study with a European focus found that the land use intensity of wind power in the EU can be of the same order of magnitude as that of fossil and nuclear energy sources if life-cycle impacts are considered (UNCCD/IRENA, 2017).

The wide range of available results emphasizes that it can be misleading to directly compare the land use impacts of renewable and conventional energy technologies. Any such comparison is highly sensitive to methodological choices, such as how to value the impacts of depletive versus non-depletive land uses over time, how to value the use of arable land versus marginal land versus offshore areas, how to value the degree of impact (e.g. mostly visual impact from wind mills versus radical land transformation from surface coal mining or the extraction of non-conventional fossil fuels), how to account for multiple or non-competitive land uses (e.g. roof-top PV, residual forest biomass and wind mills on farmland), and whether to account for upstream impacts in other regions. Rather than attempting to minimise energy-related land use based on any given indicator, the development of clean energy strategies should focus on jointly maximising the various services (such as energy production, food production, nature protection, housing and recreation) that can be provided sustainably by the available land resources in a given region.

#### Integrated policy perspective

The water and land requirements of the energy system discussed above lead naturally to integrating energy with food and water in the broader perspective of the **energy-water-land nexus** (or **energy-water-food nexus**). This nexus has also gained attention as an overarching global sustainable development policy perspective (Leck et al., 2015).

The cross-sectoral interactions imply that **EU** environmental policy legislation plays an important

role in the energy sector and its adaptation to climate change. In particular, the Water Framework Directive (WFD) emphasises the quality of hydromorphological conditions and aims to achieve good ecological status in all water bodies in the EU (EU, 2000). This may prevent energy projects that are liable to influence aquatic ecosystems, such as hydropower and thermal power plants. Preamble 16 states that the energy sector must sustainably manage and protect water, and further integration into various policy areas is encouraged. The WFD requires river basin management plans (RBMPs) to be drawn up to protect water environments across Europe. RBMPs should highlight how climate change projections have been incorporated into projects, how monitoring programmes detect climate change impacts, and how robust selected measures are to projected climate conditions. The Floods Directive, which complements the WFD, requires Member States to develop flood risk management plans (FRMPs) (EU, 2007). FRMPs focus on measures for prevention, protection and preparedness to deal with flooding in both inland and coastal water. They are often combined with RBMPs, and coordination between the two is encouraged in the directives.

# 2.2 Building a low-carbon energy system

A key driver for changes in the global and European energy system is the need for a clean energy transition. Such a transition needs to drastically reduce greenhouse gas (GHG) emissions, which lead to a changing climate; it should also reduce air pollutant emissions, which are threatening human health and the environment. This section gives an overview of relevant targets and scenarios at the global and European scale.

# 2.2.1 Global policy targets and energy scenarios

# **UNFCCC** Paris Agreement

The overarching aim of the Paris Agreement under the United Nations Framework Convention on Climate Change (UNFCCC) is to ensure that the increase in global average temperature stays well below 2 °C above pre-industrial levels, while also pursuing efforts to limit this rise to 1.5 °C (UNFCCC, 2015). All signatories have committed to determining, planning and reporting on measures intended to mitigate global warming through nationally determined contributions (NDCs).

The current commitments in the form of the NDCs are not sufficient to reduce global emissions to a level

consistent with the goals of the Paris Agreement. The latest United Nations Environment Programme (UNEP) Emissions gap report stated that 'Pathways reflecting current NDCs imply global warming of about 3 °C by 2100, with warming continuing afterwards'. It stressed that 'countries need to strengthen the ambition of NDCs and scale up and increase effectiveness of domestic policy to achieve the temperature goals of the Paris Agreement' (UNEP, 2018).

# IPCC SR1.5 pathways

The Intergovernmental Panel on Climate Change (IPCC) Special Report Global warming of 1.5 °C (IPCC, 2018) found that pathways compatible with a global warming of 1.5 °C agreed on a marked increase in the share of renewables, which may cover half to two thirds of global primary energy supply by 2050; a strong decrease in coal use, with the residual to be used only in combination with carbon capture and storage (CCS) by 2050; a rapid decline in the carbon intensity of electricity generation, coupled with a marked increase in the electrification of all end uses; and a sharp decrease in energy demand driven by a pronounced lifestyle shift. All these actions should be supplemented by increased efforts to create negative carbon emissions, for example through the use of sustainable bioenergy combined with CCS.

# IEA scenarios

The IEA has developed global energy scenarios up to 2040 (IEA, 2018b). Among the three main scenarios are (1) Current Policies, which consider only the impacts of policies already legislated for as of mid-2018; (2) the New Policies Scenario, which also includes the likely effects of policies announced by August 2018, including those to meet the NDCs for the Paris Agreement; and (3) the Sustainable Development Scenario, which presents an integrated approach to delivering on the Paris Agreement, universal access to modern energy by 2030 and improving air quality in order to significantly reduce pollution-related premature deaths. These IEA scenarios, but also the IEA Energy or Reference Technology Scenario (IEA, 2017), illustrate that significant additional action is needed if global emissions are to be made consistent with the goals of the Paris Agreement.

# 2.2.2 European policy targets and energy scenarios

This section gives an overview of how the required transition in the energy system is rooted and reflected in EU policies and forward-looking





**Source:** Adapted from EC (2019b).

Upwards revision control by 2030

scenarios and how these scenarios frame the need for energy system adaptation now and in the future. Relevant EU policies are presented in more detail in Section 4.2.

The **2030 climate and energy framework** aims to cut GHG emissions by 40 % by 2030, compared with 1990 levels (European Council, 2014). The initial framework involved increasing the share of RES to at least 27 % and increasing energy efficiency by 27 %. The last two targets have recently been strengthened, with revised targets of 32 % for RES and 32.5 % for energy efficiency (see Figure 2.6) (EU, 2018a, 2018b).

#### The proposal for a EU long-term strategy

'A Clean Planet for All', which was published in November 2018, puts forward a vision to steer the EU economy and society towards a (largely) CO<sub>2</sub> emissions-free future by 2050 (EC, 2018j). According to the Commission, the strategy shows how Europe can lead the way to climate neutrality by investing in realistic technological solutions, empowering citizens and aligning action in key areas, such as industrial policy, finance and research, while ensuring social fairness for a just transition. The strategy covers seven strategic areas, proposing joint action in each of these: (1) energy efficiency; (2) deployment of renewables; (3) clean, safe and connected mobility; (4) competitive industry and circular economy; (5) infrastructure and interconnections; (6) bio-economy and natural carbon sinks; and (7) CCS to address remaining emissions. The Commission invites all EU institutions, the national parliaments, the business sector, non-governmental organisations, cities and communities, as well as citizens, to participate in an EU-wide informed debate on the strategy. This should allow the EU to adopt and submit an ambitious strategy to the UNFCCC by early 2020 as requested under the Paris Agreement.

The long-term strategy is based on eight decarbonisation pathways, all of which were developed as a departure from the same baseline scenario (EC, 2018j, 2018q). They cover key features of the energy system (supply, demand, prices and investments) and all GHG emissions for the period up to 2050 (or even 2070) for all EU Member States, all candidate countries, Norway, Switzerland, and Bosnia and Herzegovina.

	Electrification (ELEC)	Hydrogen (H2)	Power-to-X (P2X)	Energy Efficiency (EE)	Circular Economy (CIRC)	Combination (COMBO)	1.5°C Technical (1.5TECH)	1.5°C Sustainable Lifestyles (1.5LIFE)
Main Drivers	Electrification in all sectors	Hydrogen in industry, transport and buildings	E-fuels in industry, transport and buildings	Pursuing deep energy efficiency in all sectors	Increased resource and material efficiency	Cost-efficient combination of options from 2°C scenarios	Based on COMBO with more BECCS, CCS	Based on COMBO and CIRC with lifestyle changes
GHG target in 2050	-80% GHG (excluding sinks) ["well below 2°C" ambition]					-90% GHG (incl100% GHG (incl. sinks) sinks) ["1.5°C" ambition]		
Major Common Assumptions	<ul> <li>Higher energy efficiency post 2030</li> <li>Market coordination for infrastructure deployment</li> <li>Deployment of sustainable, advanced biofuels</li> <li>Moderate circular economy measures</li> <li>Digitilisation</li> <li>Significant learning by doing for low carbon technologies</li> <li>Significant improvements in the efficiency of the transport system</li> </ul>							
Power sector	Power is nearly decarbonised by 2050. Strong penetration of RES facilitated by system optimization (demand-side response, storage, interconnections, role of prosumers). Nuclear still plays a role in the power sector and CCS deployment faces limitations.							
Industry	Electrification of processes	Use of H2 in targeted applications	Use of e-gas in targeted applications	Reducing energy demand via Energy Efficiency	Higher recycling rates, material substitution, circular measures	Combination of most Cost-	COMBO but stronger	CIRC+COMBO but stronger
Buildings	Increased deployment of heat pumps	Deployment of H2 for heating	Deployment of e-gas for heating	Increased renovation rates and depth	Sustainable buildings	efficient options from "well below 2°C" scenarios with targeted application (excluding CIRC)		CIRC+COMBO but stronger
Transport sector	Faster electrification for all transport modes	H2 deployment for HDVs and some for LDVs	E-fuels deployment for all modes	Increased modal shift	Mobility as a service			<ul> <li>CIRC+COMBO but stronger</li> <li>Alternatives to air travel</li> </ul>
Other Drivers		H2 in gas distribution grid	E-gas in gas distribution grid				Limited enhancement natural sink	<ul> <li>Dietary changes</li> <li>Enhancement natural sink</li> </ul>

Table 2.1	Scenario overview from the proposed EU long term strategy 'A Clean Planet for All'
-----------	--

Source: EC (2018q).

Table 2.1 summarises the main features of these scenarios. The Joint Research Centre has provided the global context to the EU long-term strategy in its Global Energy and Climate Outlook 2018, which includes a detailed analysis of sectoral mitigation options towards a global low-emissions economy (Keramidas et al., 2018).

The baseline is largely built on the 2016 EU Reference Scenario (EC, 2016g), but it assumes the achievement of the latest energy and climate 2030 targets. It also incorporates several policy proposals in the field of energy, transport and land use, land use change and forestry, and an update on the prospects of relevant technologies. The scenario assessment encompasses three categories of pathways: the first includes five pathways that rely on a wide range of measures, each exploring a different technological option. Three of these scenarios are driven by the different emphasis on alternative decarbonised energy carriers (electrification, hydrogen or e-fuels), while the remaining two scenarios focus on demand-side measures (enhanced energy efficiency or an enhanced circular economy). All these scenarios attain at best an 80 % emission reduction

by 2050 (compared with 1990). The second category includes only the sixth pathway ('COMBO'). This pathway combines all the options considered in the previous five decarbonisation scenarios and reaches a 90 % emission reduction, which is compatible with a 'well below 2 °C' trajectory of global mean temperature increase by the end of the century. The unavoidability of residual GHG emissions from the agricultural sector prevents full decarbonisation (which is necessary for a 1.5 °C trajectory) in the absence of measures to achieve negative emissions. The last category of pathways (seventh and eighth scenarios) explore possible solutions to push the system towards full decarbonisation. They put the emphasis on negative emission technologies ('1.5TECH') and on sustainable lifestyles ('1.5LIFE'), respectively, whereby the latter entails a paradigm shift in consumers' choices and a stronger circularity of the EU economy. The shift towards an increasingly decarbonised future is present in all eight scenarios, but it is most prominent in the last three. All scenarios also assume a marked increase in energy efficiency, whereby the residential and transport sectors are responsible for the largest cuts in final energy consumption.

The main aggregate outcomes for the energy sector in the eight decarbonisation scenarios are depicted in Figure 2.7 and Figure 2.8. The prominent role of renewables is apparent in Figure 2.7, which also highlights the significant role of nuclear energy and the impossibility of fully eliminating fossil fuels. The importance of a widespread electrification of energy uses scenario is highlighted in Figure 2.8, which also points to a virtual extinction of solid fossil fuels and to a significant contribution of biomass and of hydrogen in the fullest decarbonisation scenarios.

## 2.2.3 Key trends for the future energy system in Europe

The European energy system will experience major changes in the coming decades as it transitions towards a low-carbon energy system. Further to the scenarios from the EU long-term strategy presented in the previous section, several organisations have developed European energy scenarios with time horizons between 2030 and 2050. Relevant scenarios include the REmap scenarios of the International Renewable Energy Agency (IRENA) (IRENA, 2018b) and the 10-year network development plan (TYNDP) scenarios jointly developed by the European Network of Transmission System Operators — Electricity and Gas (ENTSO-E and ENTSO-G) in their TYNDP 2018 scenario report (ENTSOs, 2018).

All available scenarios agree that achieving the goals of the Paris Agreement requires (1) a highly decarbonised electricity generation sector, (2) the preponderance of renewables in the energy mix, (3) the electrification of most energy uses, primarily transport, and (4) a key role for energy efficiency. Other developments are dependent on the ambition level of the decarbonisation scenario, socio-economic developments and preferences, and assumptions regarding key technological developments, such as carbon capture and sequestration/use, hydrogen and electric mobility.

This section summarises key trends in the future energy system to assess their implications for climate change adaptation and resilience. Broadly speaking, energy efficiency and water-efficient RES tend to reduce adaptation challenges, whereas increased electrification by climate-dependent and water-intense RES tend to increase adaptation challenges. However, the specific adaptation challenges and opportunities of a particular energy technology differ considerably across Europe depending on the climatic, environmental and other characteristics of a particular region. More detailed information about climate change and its impacts on various components of the energy system, and about related adaptation needs and options, is presented in Chapter 3.

#### Shifts in primary energy sources and technologies

Rapid growth in RES is a central element in all decarbonisation scenarios. RES growth has been fastest in the power sector, and this trend is expected to continue. Electricity production will increasingly depend on time-variable RES, in particular wind and solar power, which are directly dependent on weather conditions. Achieving long-term climate targets also requires substantial growth in RES in other sectors and uses, in particular transport and heating. Regional strategies for RES expansion need to consider the energy-water-land nexus (see Section 2.1.5) and assess the viability of specific RES under changing climate conditions.

Hydropower will continue to play an important role in the European RES mix, both as an energy source and for energy storage. Hydropower facilities are likely to see technological modifications in the future based on the development and expansion of hydropower pumped storage, which will increase energy system resilience. Expansion opportunities are constrained by wider sustainability concerns, in particular conflicts with nature protection. The greatest remaining potential for new hydropower developments is in eastern Europe, especially in the Western Balkan region, where the EU has taken a strong interest in developing the region's energy sectors (IHA, 2018). Hydropower is highly sensitive to climate conditions, with beneficial as well as adverse impacts of climate change expected in various European regions. Any expansion of hydropower should consider the viability of investments under changing climate conditions, in addition to other sustainability concerns.

Natural gas (i.e. fossil methane) can be an important transition fuel. Natural gas is less carbon-intensive than coal if fugitive emissions are well controlled. It is more readily available than RES in the short term and already has well-established infrastructure. For these reasons, natural gas is seen in many scenarios as providing an important option for a move away from coal and towards (future) low-carbon energy sources. It can also serve as a backup for time-variable RES. Natural gas may be increasingly replaced by low-carbon gases (biogas, biomethane, synthetic methane and green hydrogen) from renewable sources. There are considerable differences between scenarios regarding the overall role of gas in the future





**Source:** Adapted from EC (2018q).







as well as the share of low-carbon gas (EC, 2018q). Most gas infrastructure is resilient to a wide range of climate conditions. Adaptation challenges can arise for offshore infrastructure, coastal infrastructure (e.g. LNG terminals) and infrastructure in permafrost regions at risk of thawing.

#### Shifts in final energy carriers

Fossil fuels for transport and heating need to be replaced by low-carbon energy carriers. For transport, these include electricity (for electric and hybrid vehicles), hydrogen (for fuel cell vehicles) and biofuels or e-fuels (for vehicles with internal combustion engines). Supporting infrastructure, such as charging infrastructure for electric vehicles, needs to be designed to be operational in a wide range of climate conditions. Options for decarbonising heating include electric heat pumps, biomass district heating systems and the use of waste heat. No particular adaptation challenges are foreseen for these technologies.

#### Hydrogen could play an important role in a future

**energy system.** There are ambitious plans for a green hydrogen economy (including power-to-hydrogen), for example in the Netherlands (Noordelijke Innovation Board, 2018). Scenarios for hydrogen development are associated with considerable uncertainty, because most technologies are still in an early stage of commercial development. It is not currently possible to identify particular adaptation challenges.

#### Changes in transmission and storage infrastructure

RES may be located far away from the main energy consumers, and their availability over time does not necessarily match the demand curve. Therefore, an energy system with a stronger role of variable RES, and of electricity as an energy carrier, requires substantial expansions of transmission and storage infrastructure.

The expansion and strengthening of electricity grids are important for the integration of increasing shares of intermittent renewables in the European energy system and for limiting the requirements for costly backup capacity, such as gas-fired power plants. This includes expanding cross-border connections and developing 'smart grids' (Schaber et al., 2012; Becker et al., 2014). European electricity network interconnection has improved in recent years, but additional strengthening of grid networks is needed and expected. The vulnerability of the network to climate extremes, such as heat waves, storms, and snow and ice accumulation, is a relevant cause of concern (see Section 3.4 and the case studies in Sections 4.5.5 and 4.5.6). An increased level of monitoring and control can significantly reduce service interruptions and improve the resilience of the low- and medium-voltage grids.

# Increasing energy storage is essential for managing the variability of renewables.

Increasing storage capacity is a priority for the Commission, which has proposed a market design initiative to introduce elements that facilitate investments in energy storage (EC, 2017a). Pumped hydropower is currently the most important technology for large-scale energy storage, but it is sensitive to water availability. Rapid advances in battery technologies, and associated cost reductions and increased uptake at both industrial and household levels, can have significant benefits for energy system resilience. Closer links between the electricity and gas sectors through power-to-gas may also be part of the solution.

The gas infrastructure may expand, for the reasons discussed above. Adaptation challenges are related mostly to offshore and coastal infrastructure and to infrastructure in permafrost regions.

CCS could be an important element in a decarbonising energy system. However, the feasibility of large-scale CCS deployment is uncertain because of the limited evidence gathered so far. Furthermore, CCS decreases the efficiency of power plants, which in turn increases their fuel and water use. Therefore, the introduction of CCS can increase adaptation challenges related to the energy-water nexus (Chandel et al., 2011; Byers et al., 2016). Under the assumption that additional biomass resources are available for energy provision, the use of bioenergy combined with CCS could lead to negative carbon emissions. However, this approach is as yet unproven and could further increase challenges related to the energy-water-land nexus (EASAC, 2019).

#### Changes in energy demand and societal changes

The deep decarbonisation pathways in the EU long-term strategy 'A Clean Planet for All' place a strong focus on decreasing overall energy demand. This will be achieved through energy efficiency improvements in buildings and electrical devices, and, potentially, changes in consumer behaviour. Climate change can also induce changes in total and peak energy demand for heating and cooling (see Sections 3.2.2 and 3.2.3). Changes in energy demand affect the overall level of the mitigation and adaptation challenge.

The pervasiveness of activities and services that depend on a continuous electricity supply has increased the social and economic costs of blackouts. Studies from several European countries have estimated that the societal costs of a power cut can be 10-100 times higher than the direct costs to the energy company (see Box 4.1 for further information). Stronger societal expectations regarding a stable electricity supply are an important driver for a more resilient energy system, which includes climate resilience as a central component.

New low-carbon technology developments may also lead to structural changes, in particular growth in decentralised energy solutions. These can create adaptation challenges if decisions affecting the management of complex infrastructure are taken by a decentralised group of new market actors rather than centrally by more experienced actors. At the same time, decentralised actors can be more flexible and are more knowledgeable about the local situation, including specific adaptation needs and options.

# 3 Climate change impacts on the European energy system

## **Key messages**

- Anthropogenic climate change has already significantly affected the European climate, and further change is inevitable. The most important changes for the energy system include increases in mean and extreme air and water temperatures, and changes in annual and seasonal water availability, extreme climate-related events, and coastal and marine hazards.
- Warming temperatures decrease energy demand for heating, but increase energy demand for cooling. They can also affect electricity generation and transmission as well as fossil fuel extraction and transport.
- Water availability is generally projected to increase in northern Europe and to decrease in southern Europe, but with marked seasonal differences. These changes can affect cooling water availability for thermal power plants, hydropower and bioenergy potential, river-borne fuel transport and energy demand for water provision.
- Climate change can also affect the potential for wind and solar power, but available projections are associated with significant uncertainty.
- Many extreme weather events, including heat waves, heavy precipitation events, storms and extreme sea levels, are
  projected to increase in frequency and/or magnitude as a result of climate change. Without appropriate adaptation
  measures, direct economic losses to the European energy system could amount to billions of euros per year by the end of
  the century, with much larger indirect costs.
- Climatic risks to the energy system can be further aggravated by the combination of different climatic hazards or by extreme events that affect several components of the energy system simultaneously.
- Climate change impacts and related adaptation needs differ widely across Europe, depending on climatic and environmental conditions as well as the structure of the regional energy system. Northern Europe experiences both beneficial and adverse impacts on its energy system, whereas southern European regions experience overwhelmingly adverse impacts.
- Most adaptation challenges to the energy system can be addressed by technical and/or management options, but many of these options are associated with additional economic costs and/or environmental impacts.

The evidence for the anthropogenic influence on the global climate system is unequivocal. Anthropogenic climate change has already led to substantial changes in the averages and extremes of many climate variables. Further climate change is inevitable, but the magnitude and pace depends on the success of global climate mitigation policies. Past and future changes in climate are affecting many aspects of human societies, including the energy system (IPCC, 2013, 2014).

This chapter gives an overview of climate change impacts on the European energy system and related adaptation needs. Section 3.1 gives an overview of important changes in the climate system for which projections are available across Europe. The

following sections look at the main impacts on the energy system and associated adaptation options caused by key changes in the climate system (see Figure 3.1), namely increasing temperature (Section 3.2), changing water availability (Section 3.3), changes in extreme climate-related events (Section 3.4) and changes in coastal and marine hazards (Section 3.5). Section 3.6 addresses further climate change impacts on renewable energy sources that do not fit easily in any of the previous sections. Section 3.7 reviews cross-cutting assessments of climate change impacts on the European energy system that complement the information presented in the previous sections. The concluding Section 3.8 presents a summary of adaptation needs, opportunities and options for the European energy system.



#### Figure 3.1 Important climatic changes and their impacts on the energy system

Source: EEA.

The high import dependence of the European energy system leaves it vulnerable to supply disruptions in countries from which Europe imports large shares of fuels (see Section 2.1.1). However, the effects of climate change impacts on the production and transport of energy carriers outside Europe on the European energy system are outside the scope of this report.

# 3.1 Overview of climate change projections for Europe

Table 3.1 gives an overview of the projected changes in several climate and climate-related variables that are particularly relevant from the perspective of the energy system, for each of the European regions identified in Section 1.2. The focus of this table is on the direction of the projected changes (i.e. increase or decrease) rather than their magnitude. Table 3.2 presents the specific information sources underlying Table 3.1. Most of the information is drawn from the EEA report Climate change, impacts and vulnerability in *Europe — an indicator-based report* (EEA, 2017b) and the underlying indicators. Where available, the climate change projections are based on a high emissions or forcing scenario (3) (e.g. representative concentration pathway (RCP) 8.5) for the late 21st century. This choice has been made for technical reasons, because high forcing scenarios typically generate a clearer anthropogenic climate change signal than natural climate variability; it does not imply that such a scenario is considered likely. In some cases,

<sup>(3)</sup> An 'emissions scenario' describes the potential evolution of several GHGs and other radiatively active emissions, whereas a 'forcing scenario' describes the radiative forcing to the climate system resulting from these emissions. The 'representative concentration pathways' (RCPs) adopted by the IPCC Fifth Assessment Report are forcing scenarios rather than emissions scenarios. Although the two terms are not identical, they are closely related. In particular, a high (low) emissions scenario is associated with a high (low) forcing scenario. This report sometimes uses the more intuitive term 'emissions scenario' in connection with RCPs even though the term 'forcing scenario' would be more exact.

projections are based on a different scenario or time horizon depending on data availability. The use of different scenarios prevents a direct comparison of the magnitude of change across climate variables. However, the direction of the projected changes (i.e. significant increase or decrease) shown in Table 3.1 is considered robust to the choice of the particular emissions scenarios and data source. Further information on the magnitude, regional and/or seasonal pattern of changes in the climate variables shown in Table 3.1 and on the underlying projections is available in the respective EEA indicators and the other information sources cited in Table 3.2.

More detailed information on past and projected climate change in Europe is available from the **Climate Data Store (CDS) of the Copernicus Climate Change Service (C3S)** (<sup>4</sup>) and from the **IMPACT2C web atlas** (<sup>5</sup>). Climate information with particular relevance for the energy sector is available in the **CLIM4ENERGY visualisation tool** (<sup>6</sup>) (see also Section 4.2.4).

## Table 3.1 Projected changes in climatic conditions across Europe

		Northern Europe	British Isles	Central western Europe	Central eastern Europe	lberian Peninsula	Apennine Peninsula	South- eastern Europe
ient rature	Air temperature	仓	仓	仓	仓	仓	仓	仓
Ambient temperature	River temperature	仓	仓	仓	仓	仓	仓	仓
	Annual precipitation	仓	仓	$\bigtriangledown$	$\bigtriangledown$	Û	$\Sigma$	$\Sigma$
ter bility	Annual river flow	仓	仓	$\bigtriangledown$	$\bigtriangledown$	Û	$\Sigma$	$\Sigma$
Water availability	Low river flow*	$\bigtriangledown$	Û	$\Sigma$	$\Leftrightarrow$	Û	Û	Û
	Summer soil moisture**	$\bigtriangledown$	$\Sigma$	$\Sigma$	$\Leftrightarrow$	Û	Û	Û
ents	Heat waves	仓	仓	仓	仓	仓	仓	仓
Extreme climate-related events	Inland floods	$\Leftrightarrow$	仓	仓	仓	$\Leftrightarrow$	$\bigtriangledown$	$\bigtriangledown$
Extreme ate-related	Wind storms	$\nabla$	$\bigtriangledown$	$\bigtriangledown$	$\bigtriangledown$	$\Sigma$	$\Sigma$	$\Sigma$
clim	Forest fire danger	$\Leftrightarrow$	仓	$\bigtriangledown$	$\Leftrightarrow$	仓	仓	仓
Coastal and marine hazards	Relative sea level	$\Leftrightarrow$	仓	仓	仓	仓	仓	仓
Coast marine	Storm surges and wave length	$\bigtriangledown$	$\bigtriangledown$	$\bigtriangledown$	$\bigtriangledown$	$\Leftrightarrow$	$\Leftrightarrow$	$\Leftrightarrow$

û Increase throughout the region

. . . . .

Decrease throughout the region

Inconsistent or limited changes

\* A downward arrow indicates a lower streamflow during low flow events, i.e. more severe river flow droughts.

\*\* A downward arrow indicates lower soil moisture (in summer), i.e. more soil water stress.

Ω

**Source:** The data sources are specified in Table 3.2.

(4) https://climate.copernicus.eu and https://climate.copernicus.eu/climate-data-store

(5) https://www.atlas.impact2c.eu/en

(6) http://c4e-visu.ipsl.upmc.fr

Climate category	Climate indicator	Sources	Notes
Ambient temperature	Air temperature	Jacob et al., 2013; EEA, 2017b, Map 3.4	Based on projected changes in annual air temperature by the late 21st century under the RCP8.5 scenario from EURO-CORDEX simulations.
	River temperature	van Vliet et al., 2013a; EEA, 2017b, section 4.3.5	Available projections for river temperature do not distinguish between European regions, but river temperature generally follows air temperature (Mohseni and Stefan, 1999).
Water availability	Annual precipitation	Jacob et al., 2013; EEA, 2017b, Map 3.8	Based on projected changes in annual precipitation by the late 21st century under the RCP8.5 scenario from EURO-CORDEX simulations. Reductions in precipitation during summer are projected even for some regions where annual water availability is projected to increase or remain constant.
	Annual river flow	Bisselink et al., 2018, Figure 20	Based on projected changes in average streamflow by the late 21st century under the RCP8.5 scenario as calculated by the Lisflood model.
	Low river flow	EEA, 2017b, Map 4.12, updated based on personal communication with Luc Feyen (JRC)	Based on projected changes in the 10-year river water deficit by the late 21st century for the RCP8.5 scenario as calculated by the Lisflood model. The river water deficit is defined as the cumulative discharge below a certain threshold (in this case the fifth percentile of daily river flow in the baseline period).
	Summer soil moisture	Bisselink et al., 2018, Figure 41	Based on the projected change in root soil moisture stress in summer for a global warming of 2 °C as calculated by the Lisflood model. (This model simulation has not been performed for the RCP8.5 scenario.)
Extreme climate-related events	Heat waves	Russo et al., 2014; EEA, 2017b, Map 3.6	Based on projections of changes in 'very extreme' heat waves (see the reference for an exact definition) by the late 21st century under the RCP8.5 scenario from a CMIP5 ensemble of global climate models. Qualitatively consistent results were derived with regional climate models (Smid et al., 2019).
	Inland floods	Alfieri et al., 2015; EEA, 2017b, Map 4.8, updated based on personal communication with Luc Feyen (JRC)	Based on projected changes in peak 100-year river discharge by the late 21st century for the RCP8.5 scenario as calculated by the Lisflood model. Pan-European projections for flash floods are not currently available.
	Wind storms	Donat et al., 2011; EEA, 2017b, Map 3.11	Based on projections of changes in extreme wind speed by the late 21st century under the IPCC SRES A1B scenario from global and regional climate model ensembles, supplemented by other modelling studies. The level of uncertainty for this indicator is larger than that for other indicators.
	Forest fire danger	EEA, 2017b, Map 4.18; updated based on de Rigo et al., 2017, Figure 8	Based on the projected change in average forest fire danger (Canadian Fire Weather Index) by the late 21st century under the RCP8.5 scenario.
Coastal and marine hazards	Relative sea level	IPCC, 2013; EEA, 2017b, Map 4.5	Based on the projected change in relative sea level by the late 21st century for the RCP4.5 scenario based on an ensemble of CMIP5 climate models. Quantitative projections for the Black Sea are not available. The value for 'South-eastern Europe' is therefore based on projections for the eastern Mediterranean only.
	Storm surges and wave height	Vousdoukas et al., 2017, Figure 5; EEA, 2017b, Section 4.2.2	Based on projections of the combined effect of storm surges and waves to 100-year extreme sea levels by 2050 for RCP8.5.

#### Table 2.2 Infe .... Table 2.4 ٤..

Note:

CMIP5, Coupled Model Intercomparison Project. The term 'climate-related event' comprises extreme weather events (i.e. heat waves and wind storms) as well as hazards that are clearly linked to extreme climate conditions (i.e. inland floods and forest fire danger).
## 3.2 Impacts of increasing temperature

Increasing temperature can affect all components of the energy system. Arguably, changes in energy demand (i.e. heating and cooling demand) are more important than changes in energy production, conversion and transmission.

## 3.2.1 Changes in ambient temperature

Air temperature is rising both globally and in Europe. Globally, 17 of the 18 warmest years on record have occurred since 2000. European land temperatures in the most recent decade (2009-2018) were 1.6-1.7 °C above pre-industrial levels. Winter temperatures have increased most in northern Europe, and summer temperatures have increased most in southern Europe. This increasing trend is likely to continue in the future as a result of the emissions already released to the atmosphere; the magnitude of future warming depends on emissions in the coming decades (EEA, 2019c).

In parallel with air temperature, **lake and river temperatures in Europe have increased** by 1-3 °C over the past century. Inland surface waters are projected to continue to warm further, driven by changes in regional air temperature (Dokulil, 2013; EEA, 2016c).

## 3.2.2 Total energy demand for heating and cooling

Heating and cooling in residential buildings and businesses (excluding process heat in industry) use about one third of the EU's final energy (EC, 2016c). This part of the energy balance is directly sensitive to changes in temperature (and to a lesser degree, in humidity). Heating degree days (HDD) and cooling degree days (CDD) are proxies for the energy required to heat or cool, respectively, a home or a business. HDDs and CDDs are defined relative to a baseline outside temperature below or above which a building is assumed to need heating or cooling.

The EEA indicator on heating and cooling degree days is based on sub-daily temperature data and uses baseline temperatures for HDDs and CDDs of 15.5 °C and 22 °C, respectively (Spinoni et al., 2015; EEA, 2019d). According to this indicator, **the number of population-weighted HDDs in Europe decreased on average** by 6.5 HDDs per year over the period 1981-2017; this corresponds to 0.29 % per year, relative to the average over this period. **The number of population-weighted CDDs increased** by 0.9 CDDs per year over the same period, which corresponds to 1.0 % per year. In absolute terms, the decrease in HDDs is much larger than the increase in CDDs, whereas in relative terms, the increase in CDDs is much larger than the decrease in HDDs, because of lower absolute values (EEA, 2019d).

Map 3.1 shows projected trends for changes in HDDs and CDDs until 2100 for two different global forcing scenarios (RCP8.5 and RCP4.5). Model results for a scenario compatible with the temperature stabilisation target in the Paris Agreement were not available. The geographical patterns reflect the pace of regional warming in different seasons as well as regional heating and cooling demand. As a result, the largest decrease in HDDs is projected for northern Europe where heating demand is highest whereas the largest increase in CDDs is projected for southern Europe, where cooling demand is highest.

The actual energy demand for heating and cooling depends not only on the HDDs and CDDs at the specific location, but also on a variety of socio-economic and technological factors, such as living space per person, building design and insulation, availability and type of heating and cooling systems, energy prices and income levels, and behavioural aspects. Importantly, heating and cooling systems are often based on different technologies, with different primary energy needs and economic costs (Auffhammer and Mansur, 2014; Buceti, 2015).

Several model-based studies agree that the projected changes in temperature reduce the total energy demand in cold countries in Europe, whereas energy demand in warm countries increases. The studies also agree that increases or decreases in total energy or electricity demand at the national level as a result of climate change alone will be below 5 % by the middle of the century (Mima and Criqui, 2015; Damm et al., 2017; Wenz et al., 2017). Although these changes are rather minor, adaptation needs can arise from their combination with changes in socio-economic characteristics (e.g. increased availability of cooling systems) and from changes in peak energy demand (see Map 3.1).

## 3.2.3 Peak electricity demand for cooling

Climate change affects not only overall energy demand but also peak demand. **Peak electricity demand for cooling, which is almost exclusively provided by electricity, will increase throughout Europe.** The largest absolute increases in electricity peak demand for cooling have been projected for Italy, Spain and France (Damm et al., 2017). A different study, using wider country coverage and somewhat different assumptions, projects increases in average daily electricity peak load over the 21st century under a high emissions scenario of up to 8 % for some Western



#### Map 3.1 Projected change in heating and cooling degree days





**Note:** The maps show projected changes in HDDs (left column) and CDDs (right column) over the period 1981-2100 for two emission scenarios: RCP4.5 (medium emissions, top row) and RCP8.5 (high emissions, bottom row). The maps are based on the ensemble mean of 11 bias-adjusted EURO-CORDEX simulations. Changes are expressed as linear trends per year.

Sources: EEA (2019); Spinoni et al. (2018).

Balkan countries, with somewhat lower values for EEA member countries (Wenz et al., 2017).

The main adaptation challenge relates to the stability of electricity networks during heat waves when an increased peak electricity demand for cooling may coincide with limited cooling water supply for thermal power generation. In countries and regions where air conditioning is common, network stability during heat waves needs to be considered in the planning and management of electricity production and distribution infrastructure. The stability of the electricity supply during heat waves can be increased by strict efficiency standards for cooling equipment, which limit the demand peak; increased transmission capacity, including international linkages; sufficient backup capacity; and shifts to power sources with low water use, in particular solar photovoltaics (PV). Cooling needs can also be reduced through improved building design, such as insulation, orientation and sun protection for windows, and through cooling technologies that do not use electricity, such as heat-driven absorption cooling and the direct utilisation of cooling water where available. A recent review highlights assessments of the costs and benefits of adaptation options for cooling as a key knowledge gap (Tröltzsch et al., 2018).

# 3.2.4 Efficiency of electricity generation and transmission

**Increasing temperatures, including extreme heat, are also affecting electricity generation and transmission.** They lead to efficiency losses of thermal power plants (i.e. fossil fuel, biomass and nuclear power plants). The loss in power output is estimated to be around 0.5 % per degree (Cronin et al., 2018). These losses can be reduced by pre-cooling the air used in combustion, but this also consumes energy.

The rated capacity of power lines and transformers can be reduced during heat waves. Studies in the United States have estimated the reduction in transmission capacity of power lines at about 1.5 % per degree of summer warming (Sathaye et al., 2013). This decrease can threaten the security of the electricity supply in warm regions if it coincides with peaks in electricity demand for cooling during heat waves. Adaptation measures include installing conductors with hotter operating limits, the use of 'low-sag' conductor material in transmission lines or increasing the overall transmission capacity of power lines (see the case study in Section 4.5.5).

## 3.2.5 Fossil fuel extraction and transport

The largest impacts of a warming climate on fossil fuel extraction and transport in Europe are expected for the Arctic region. This region is already experiencing very high temperature increases, and this trend is expected to continue (Huang et al., 2017). As a result of this warming, previously inaccessible fossil energy resources may become accessible. If these resources are exploited, this may lead to positive impacts on the local economy. It could also increase energy security if these fuels replace fuels from geopolitically unstable regions. However, such developments can have significant negative impacts on the regional environment, which is characterised by fragile and already changing ecosystems. Furthermore, increased production and use of fossil fuel resources in the Arctic would counteract climate change mitigation efforts (EEA, 2017g). Currently, fossil fuel extraction in the (European and non-European) Arctic is limited. Several planned projects have been halted or cancelled because of excessive cost, unknown human and environmental risks, significant stakeholder opposition and regulatory challenges (Koch, 2015; Schaps, 2015).

Energy infrastructure in the (European and non-European) Arctic is affected by degrading permafrost. This could result in damage to existing gas and oil pipelines and subsequent disruptions to the energy supply (Forzieri et al., 2018; Hjort et al., 2018).

# 3.3 Impacts of changing water availability

Climate change is affecting water availability, and there are large variations across European regions. These changes, combined with increases in water temperature, can have substantial impacts on different forms of power production; they can also affect energy demand, for example through increased energy needs for desalination or irrigation.

## 3.3.1 Changes in water availability

Climate change is projected to intensify the hydrological cycle in general and to change precipitation patterns across Europe. The most relevant hydrological variables from the perspective of the energy system are annual river flow and river flow droughts (for thermal power plants and hydropower plants) and soil water availability (for bioenergy production). These variables are driven by the combined effect of changes in precipitation patterns, temperature and the physiological responses of plants to increasing atmospheric  $CO_2$  concentrations.

Past and projected changes in precipitation show marked differences across Europe. Northern Europe has generally become wetter in recent decades (up to 70 mm per decade), whereas southern Europe has generally become drier (up to 90 mm per decade). This trend is projected to continue in the future but with marked seasonal differences. The strongest decrease is projected for southern Europe in the summer (EEA, 2017d).

**Projected changes in annual river flow show a pattern similar to that for annual precipitation.** Decreasing river flows are projected for southern European countries (particularly Cyprus, Greece, Italy, Malta, Portugal, Spain, and Turkey) and increasing flows for northern European countries (EEA, 2016a; Bisselink et al., 2018).

Droughts can be distinguished into meteorological, soil moisture (or agricultural), hydrological (including river flow) and socio-economic droughts. The severity and frequency of meteorological and hydrological droughts has increased in parts of Europe, particularly south-western and central Europe. Available studies project large increases in the frequency, duration and severity of meteorological and river flow droughts in most of Europe over the 21st century, except for northern European regions (EEA, 2019f; Bisselink et al., 2018).

Changes in summer soil moisture follow a pattern similar to that of changes in river flow droughts. Decreases in summer soil moisture (i.e. increases in soil moisture stress) are projected for most European regions, except for northern and north-eastern Europe. Summer soil moisture is projected to decrease in some regions where annual precipitation is projected to increase, such as north-western Europe (EEA, 2016b; Bisselink et al., 2018).

## 3.3.2 Cooling water availability for thermal power plants

Most thermal power plants (i.e. fossil fuel, biomass and nuclear power plants) are reliant on large quantities of water for cooling. Depending on the specific cooling technology, cooling water is either discharged to a water body at higher temperature or evaporated in cooling towers. Thermal power plants cooled with freshwater are vulnerable both to reductions in cooling water availability and to increases in its temperature. These hazards, individually or in combination, can lead to partial or complete shutdowns of power plants. Shutdowns can occur even if the amount of streamflow is still sufficient for cooling purposes, because the water temperature of the river into which the cooling water is discharged must not exceed a threshold temperature established to protect its ecosystems. For example, during the heat waves in 2003, 2006, 2009 and 2018, numerous nuclear reactors in France and other countries were temporarily shut down because of cooling water shortages and/or to prevent temperature exceedances in rivers (Paskal, 2009; Dubus, 2010; Rübbelke and Vögele, 2011; Reuters, 2018). Furthermore, increased water temperature causes efficiency losses in power production and associated output losses. The combined summer heat wave and drought of 2003 is estimated to have reduced thermoelectric power utilisation rates in Europe by about 5 % (van Vliet et al., 2016a).

Several studies have used coupled climate-energy-water models to assess the impacts of future climate change on electricity production in Europe. They have estimated that climate change could decrease the usable water capacity of thermoelectric power plants in Europe by more than 15 % in some cases by the middle of the century and would increase mean annual wholesale prices for electricity for most European countries (van Vliet et al., 2012, 2013b, 2016b). However, there are large differences depending on the region, the season and the specific cooling technology applied.

Adaptation options for thermal power plants in water-stressed regions include technological changes such as closed cooling and dry cooling systems. However, dry cooling is more costly and could result in efficiency losses in the order of 10 % (Murrant et al., 2015; van Vliet et al., 2016c). Alternatively, condensing power plants could be converted to combined heat and power plants that use the excess heat (e.g. in district heating systems). A more fundamental adaptation option is switching to alternative generation technologies with low water use (e.g. wind and solar PV). Carbon capture and storage (CCS) decreases the efficiency of thermal power plants and increases their cooling water requirements. Therefore, water scarcity may limit the feasibility of CCS as a decarbonisation strategy in regions with cooling water constraints.

A recent study assessed climate change impacts on more than 1 300 thermoelectric power plants and more than 800 water basins throughout Europe. The study found that climate change would increase the number of basins under water stress and that the majority of vulnerable basins in Europe are located in the Mediterranean region. It also considered four adaptation options to alleviate water stress caused by electricity generation: (1) seawater cooling, (2) dry air cooling, (3) early retirement, and (4) enhanced renewable energy generation. Additional seawater cooling of coastal units was seen as the most effective option for the Mediterranean region by 2030 (Behrens et al., 2017).

## 3.3.3 Hydropower potential

The dependence of hydropower plants on streamflow for power production makes them sensitive to changes in rainfall, snowfall, and snow and glacier melt (Majone et al., 2016; Bonjean Stanton et al., 2016). In locations where streamflow is expected to decline, this would undermine the productivity of hydropower installations and their cost effectiveness compared with other generation technologies. An increase in streamflow could increase hydropower production, but high river flows can also damage hydropower plants. Run-of-river plants have limited regulating capacity and are thus directly affected by changes in streamflow. Hydropower projects with substantial reservoir storage can be more resilient to short-term changes in river flows, and they can provide additional services, such as flood protection. However, they are affected by increasing temperatures, which increases evaporation from the reservoir. More than half of Europe's hydropower capacity is held in pumping-storage power plants, which buffer not only against fluctuating streamflow but also against fluctuating electricity demand (World Energy Council, 2019).

Hydropower generation in Europe is dominated by the Nordic countries (in particular Norway and Sweden) and the Alpine region. Increased annual or seasonal water availability is projected for most of these regions due to increasing precipitation and possibly increased glacier melt. However, hydropower is also an important energy source in southern European countries that face decreasing water availability, in particular in Turkey (which has considerable expansion plans), but also in Italy, Portugal and Spain.

Glacial retreat caused by rising temperatures can have different impacts on hydropower. In the short to medium term, glacial retreat can increase hydropower production by increasing streamflow (see the case study in Section 4.5.3 for an example). However, excessive water availability can challenge hydropower storage capacities, with the risk of power outages. Glacial retreat may also increase sediment transport, particularly in Alpine regions. This sediment can accumulate in reservoirs and reduce their storage capacity if not managed correctly. It can also flow into water intake channels, thereby reducing water flow, while also causing turbine deterioration (Gaudard and Romerio, 2014; International Hydropower Association, 2019). In the long term, glacial retreat decreases streamflow, in particular in the spring and summer melt season.

## 3.3.4 River-borne fuel transport

Climate change is projected to increase the risk of river flow droughts in most parts of Europe. Low water levels in rivers during drought conditions can restrict inland navigation, including the transport of coal and oil by barge. For example, the unprecedented drought that affected considerable parts of Europe in the spring and summer of 2018 resulted in shipping restrictions on the Rhine, Elbe and Danube rivers. Shallow waters required vessel operators to reduce loads, which led to increased shipping and storage costs and capacity limitations, with subsequent knock-on effects throughout the energy supply chain (Lehane, 2018).

## 3.3.5 Energy demand for water provision

Increasing temperatures are the main means by which climate change affects energy demand, but changes in water availability can also have an effect. **Decreasing precipitation**, **increasing evaporation due to increasing temperatures and salt-water intrusion due to rising sea levels**, **combined with socio-economic changes, could increase the energy demand to provide sufficient water** for agriculture, households and other uses. The EU currently has around 13 % of the global desalination capacity, most of it in Spain (IEA, 2016c). The demand for energy-intensive seawater desalination could increase in water-stressed regions in southern Europe, but quantitative projections are not available. Energy demand may also increase in other elements of the water supply chain.

# 3.4 Impacts of changes in extreme climate-related events

Extreme weather and climate-related events, such as coastal and inland flooding, storms, hail and lightning, heat waves, droughts and wildfire events, can affect the energy system in many ways (Schaeffer et al., 2012; EC, 2013a; Troccoli, 2018). This section focuses on the direct physical impacts on energy infrastructure. Selected impacts related to heat waves and droughts have already been presented in Sections 3.2 and 3.3.

Many extreme weather events are projected to increase in frequency and/or magnitude as a result of climate change. Without appropriate adaptation, direct economic losses to the European energy system could amount to billions of euros per year by the end of the century (Forzieri et al., 2016, 2018). Overall impacts on society could be much higher than the direct losses, because damage to critical energy infrastructure can result in failures and cascading effects on related and dependent infrastructures, with far-reaching economic and social consequences (Karagiannis et al., 2017; Varianou Mikellidou et al., 2017). An examination of about 40 major blackouts worldwide over the past 40 years found that extreme weather was the most important primary cause. Within this category, storm damage to the transmission system was the cause of half of the failures. Other notable weather effects were drought (loss of cooling water) and ice build-up on transmission lines (Boston, 2013).

## 3.4.1 Changes in extreme climate-related events

Climate change has resulted in changes to many extreme weather events both globally and in Europe (EEA, 2017b, 2017a). Further changes in the frequency, intensity and location of such events are projected throughout Europe (Forzieri et al., 2016). Extreme weather events relevant to the energy system include heat and cold waves, heavy precipitation events, droughts, and wind and hail storms. Such extreme events can cause or aggravate further climate-related hazards, such as pluvial and fluvial floods and forest fire danger.

The intensity of **heavy precipitation** events has generally increased since the 1950s in northern and north-eastern Europe, whereas changes have been more inconsistent in southern and south-western Europe. Heavy precipitation events are projected to become more frequent in most parts of Europe in the future (EEA, 2019e). As a result, **an increase in the occurrence and frequency of inland floods** (i.e. pluvial and fluvial floods) is projected for most parts of Europe (EEA, 2017f). Snow and ice can also affect energy infrastructure, in particular electricity transmission and distribution. However, reliable projections for changes in weather conditions that support snow and ice accumulation are not currently available.

Wind storms have not shown robust long-term trends so far, partly due to limited data availability. Climate change simulations show diverging projections on changes in the number of wind storms across Europe. Most studies agree that the risk of severe winter storms, and possibly of severe autumn storms, will increase for the North Atlantic and northern, north-western and central Europe over the 21st century (EEA, 2017h). Convective storms can be associated with hail and/or lightning, which can affect energy infrastructure. However, Europe-wide projections for changes in hail and/or lightning are not currently available (Yair, 2018).

Heat waves and extreme heat events have become much more frequent and intense in recent decades as a result of climate change. For example, 65 % of Europe has experienced all-time record high temperatures in the period 2003-2010 alone. Further temperature records in Europe were broken in subsequent years. Extreme summer heat waves, such as those experienced in different parts of Europe in 2003, 2010 and 2018, will become much more common in the future. For example, at the end of the 21st century, maximum temperatures in 90 % of the summers in southern, central and north-western Europe are projected to be warmer than in any summer in the period 1920-2014 for a high-emission scenario (EEA, 2019c).

The occurrence of forest fires is linked to seasonal meteorological conditions, forest management practices, vegetation coverage and various socio-economic factors. These complex interactions make it difficult to assess historical trends and to project future changes in forest fire risk. Fire danger indices such as the Canadian Fire Weather Index and seasonal severity ratings have been developed in order to rate the potential for fires caused by weather conditions. Available projections for these indices suggest that the duration, severity and area at risk of forest fires will increase in the future in most European regions, with the possible exception of northern and central eastern Europe (EEA, 2019a; de Rigo et al., 2017). However, projections of forest fire risk are sensitive to the particular index used. For example, although available European-level studies suggest inconsistent changes in forest fire risk in an average summer in northern Europe, national-level studies using different indices suggest an increased forest fire risk in Sweden (Sjökvist et al., 2013).

## 3.4.2 Impacts of inland flooding

Inland floods can have significant impacts on energy infrastructure, such as electric substations, and oil and gas pipelines (Varianou Mikellidou et al., 2017). For example, heavy and prolonged rainfall throughout Portugal in 2000 caused damage to levees and the overtopping of dams. The subsequent flooding threatened to damage a major gas pipeline in the area, leading to the evacuation of more than 100 people (Cruz and Krausmann, 2013). The river-based transport of fuels may also be affected.

Increasing heavy precipitation events are expected to increase river and flash floods in many parts of Europe. Country-level analysis shows that the largest increases in flood hazard, and by association the highest needs for adaptation, are projected for the United Kingdom, France, Italy, Romania, Hungary and Czechia (Rojas et al., 2013). Adaptation options include risk-based planning, changing the operational regimes of reservoirs, dyke construction, component-based flood barriers and relocation (Ebinger and Vergara, 2011).

## 3.4.3 Impacts of storms

Wind, hail, ice and snow storms can adversely affect energy transformation and transport, transmission, distribution and storage infrastructure, resulting in blackouts and costly **repairs** (Rübbelke and Vögele, 2011). Power lines and wind turbines are particularly affected. For example, the series of wind storms that struck central western Europe in December 1999 (Storms Lothar and Martin) were estimated to have caused economic losses of EUR 15 billion overall, with associated blackouts affecting 3.4 million people (Groenemeijer, 2015).

Wind storms can affect overhead power lines either directly or indirectly, through vegetation coming into contact with or falling on to them. Ice and snow storms can also have important impacts on electricity transmission infrastructure, leading to costly damage and power outages. Ice can accumulate on power lines and vegetation when freezing rain or fog comes into contact with them at temperatures below freezing. Ice storms are most frequent in northern Europe. The accumulation of wet snow on power lines is promoted by conditions of surface temperatures close to freezing, high humidity and moderate winds, which are often observed over the Mediterranean area (Bonelli et al., 2011; Llasat et al., 2014). The meteorological conditions for snow and ice accretion are complex, and there are currently no reliable long-term projections for these hazards.

The structural enforcement of pylons and improved vegetation management are important elements in making power lines more resilient to storms, but they can be associated with significant costs. Underground cabling is being considered in several European countries as another measure to reduce storm damage, but it is also an expensive adaptation option (see the case study in Section 4.5.6).

Hail storms, which are most frequent in mountainous areas and pre-Alpine regions, can cause costly damage, in particular to solar panels (Ebinger and Vergara, 2011). A trend towards an increasing number of hail storms has been noted in Austria and southern France, and decreasing (but not statistically significant) trends have been noted in parts of eastern Europe. Future projections of hail events are subject to a large degree of uncertainty, but model-based studies show some agreement that hail storm frequency will increase in central Europe (EEA, 2017c).

## 3.4.4 Impacts of heat waves and forest fires

Heat waves can cause overheating in electric transformers, in particular large power transformers, through reduced structural integrity and chemical reactions. This overheating can result in short circuits, power outages and costly repairs (Gao et al., 2018). The risk of transformers overheating during heat waves is further increased by peaks in electricity demand in regions where air conditioning is common.

Heat waves combined with droughts can lead to further adverse impacts on the energy system. They can decrease water levels in storage hydropower plants, thereby reducing their generation and storage capacity. They can also increase the risk of forest fires. On the one hand, forest fires can damage energy infrastructure, in particular wooden power poles. On the other hand, electricity transmission infrastructure is an important ignition cause of forest fires (Camia et al., 2013). Box 3.1 presents a recent example from outside Europe with catastrophic human, social and economic consequences, including for a large electric utility.

## 3.4.5 Combined and cascading effects

The combination of different climatic hazards can aggravate the risks to the energy system. For example, both droughts and heat waves are projected to increase in southern Europe. The simultaneous occurrence of these hazards can limit the energy supply and transmission in hot, water-scarce regions while increasing peak electricity demand for cooling. In extreme cases, this combination of hazards may lead to blackouts, with knock-on effects on other sectors that depend on a stable electricity supply. Another dangerous combination of hazards is river flooding plus a storm surge, which can multiply flooding risks in coastal regions, including risks to the energy infrastructure.

## Box 3.1 Energy infrastructure creating fatal risks to public safety

California suffered from unprecedented wildfires in 2017 and in 2018 due to extremely warm, dry and windy conditions. Many of the fires in 2017 were found to have been caused by electric power and distribution lines, conductors and power poles owned by Pacific Gas and Electric Company (PG&E), which sparked fires as a result of coming into contact with vegetation, sagging or other failures (CAL Fire, 2018a, 2018b). It is suspected that the 2018 Camp Fire, the deadliest and most destructive wildfire in California's history, was also caused by a faulty PG&E transmission tower (Gafni, 2018). Facing USD 7 billion in claims arising from the Camp Fire, PG&E, the largest energy utility company in the United States, had to file for bankruptcy in January 2019 (Tsang, 2019).

### Box 3.2 Examples of widespread power outages resulting from damage to transmission and distribution networks

A large power outage took place in **Italy and Switzerland in September 2003**. The primary cause was a tree flashover that led to the tripping of an electricity transmission line connecting Switzerland and Italy, which was exacerbated by inadequate responses in the immediate aftermath of the event. The load rescheduling led to flashovers and the subsequent tripping of other highly loaded lines connecting Switzerland and Italy. The resulting decrease in tension and frequency caused an almost immediate chain-reaction of power plant shutdowns and cascading failures along the Italian transmission network, which also affected the interconnections with Slovenia and Austria. Power failures lasted from 90 minutes in north-west Italy to 19 hours in Sicily. They affected about 56 million people in Switzerland and Italy, to whom about 180 GWh of electricity failed to be delivered. The economic costs of this blackout have been estimated at EUR 1.2 billion (Walker et al., 2014).

The triggering event for this systemic failure could have been avoided by proper maintenance of transmission lines and control of the foliage surrounding transmission lines in Switzerland. Further analysis of the events highlighted that the network operator in Italy did not act fast enough to reduce the grid capacity being used for pumped storage, which could have eased the burden on transmission lines from increased electricity imports from France. This case highlights that assessing the risks and vulnerabilities of a system, and following through on necessary maintenance measures, is integral to preserving its function in times of stress. By taking stock of potential vulnerabilities to the energy system, the damage costs of EUR 1.2 billion could have potentially been avoided (Johnson, 2008; Böttcher, 2016).

Storm Gudrun caused significant destruction in parts of **northern Europe in January 2005**. In Sweden, 30 000 km of distribution lines were damaged, leading to long-lasting power disruptions for around 730 000 customers. In urban areas with underground cabling, power was restored within a few hours, whereas rural areas experienced outages lasting up to 20 days. The Swedish network operators reported losses of around EUR 250 million, but the overall costs of the power interruptions to society were estimated at EUR 3 billion (Gündüz et al., 2017).

Extreme events can also affect two or more components of the energy system simultaneously. The risk of cascading impacts and system failure is particularly high in the case of power outages, because electric power is central to many other components of the energy system as well as to society at large. Box 3.2 provides two examples of cascading impacts from damage to transmission lines.

# 3.5 Impacts of changes in coastal and marine hazards

## 3.5.1 Changes in coastal and marine hazards

Changes in mean and extreme sea levels, storm surges and wave heights affect coastal hazards, such as inundation, flooding and erosion, as well as marine hazards. This is of relevance for coastal and offshore infrastructure.

Global and European **sea levels** have risen significantly since 1900. Global sea level rise has accelerated from earlier rates of 1.2-1.7 mm per year over the 20th century to at least 3 mm per year in recent decades. **Most coastal regions in Europe have experienced an increase in absolute and relative sea levels, which increases the risk of coastal flooding.** Geographic variations occur as a result of local land subsidence or rise, coastal morphology, changes in wind and wave regimes, and other factors. The model simulations used in the IPCC AR5 projected a rise in global sea level over the 21st century that is likely in the range of 28-98 cm (depending on the emissions scenario), but substantially higher values of sea level rise were not ruled out. This range will be revised in the IPCC Special Report *The Ocean and Cryosphere in a Changing Climate*, which is due for publication in September 2019. Several recent model-based studies, expert assessments and national assessments have suggested an upper bound for 21st century global mean sea level rise in the range of 1.5-2.5 m. (EEA, 2019b; Vousdoukas et al., 2017).

Trends towards increased **wave heights** in winter have been recorded in the North Atlantic and are affecting coastlines in western Europe (Castelle et al., 2018). Projections of the **future contribution of climate extremes (storm surges and wave heights) to extreme sea levels** suggest an increase along the North Sea, British and Baltic Sea coasts, mixed changes in the Mediterranean and decreases along the Norwegian coast and the Atlantic coast of the Iberian Peninsula (Vousdoukas et al., 2017).

## 3.5.2 Impacts on coastal infrastructure

Many types of energy infrastructure, such as oil and gas terminals, refineries, nuclear and fossil fuel power plants, are often placed in close proximity to coastlines or estuaries where they are exposed to various coastal hazards (Cruz and Krausmann, 2013). A comprehensive assessment of climate change risks to key coastal energy infrastructure was conducted in the ClimateCost project (Brown et al., 2014).

**Coastal erosion**, which is facilitated by sea level rise, results in the loss of natural barriers to rising waters. Subsequent damage to coastal infrastructure may occur via direct wave impact, loss of infrastructure stability through land loss, or subsequent flooding events (Azevedo de Almeida and Mostafavi, 2016). Adaptation measures to coastal flooding and erosion are closely connected and often involve physical protection measures. As an example, beach profiling, beach feeding and shingle recycling were used to reduce coastal erosion and associated flooding at the Dungeness nuclear power station in the United Kingdom (DECC, 2010).

**Storm surges** refer to abnormal rises in water levels that are generated by a storm. They are exacerbated by sea level rise and can combine with high-tide events to create extreme sea levels. Coastal energy infrastructure, including ports and refineries related to offshore oil, gas and wind energy production activities, and coastal power plants, are particularly vulnerable to storm surge impacts (Azevedo de Almeida and Mostafavi, 2016). Due to changing water levels and sediment supply, storm surges can also temporarily disrupt the flow of cooling water at coastal power plants.

Tidal dynamics may also be affected as a consequence of sea level rise and storm surges, thereby impacting specific renewable energy sources such as tidal and wave power (Lewis et al., 2017; Pickering et al., 2017).

Adaptation measures against storm surges depend on the specific type of infrastructure. Forecasting and planning around such events is essential to protect vital infrastructure. Physical infrastructure, ecosystem-based solutions for coastal protection and management measures can also increase resilience (see the case study in Section 4.5.4).

## 3.5.3 Impacts on maritime infrastructure

Maritime storms are a major hazard for offshore energy infrastructure. The increase in wind speeds and wave heights during storm events, which may be exacerbated by sea level rise, makes the operation and maintenance of oil and gas platforms and of wind farms more dangerous and they may require temporary shutdowns (Bell et al., 2017).

Responding to the increasing importance of offshore renewable energy production, the EU has funded

several research projects with regard to multi-use offshore platforms (Quevedo et al., 2013; Zanuttigh et al., 2016). There are also innovative efforts under way for offshore system integration that reuse existing sea-based infrastructure (North Sea Energy, 2019).

# 3.6 Further impacts on renewable energy potential

This section summarises further impacts of climate change on renewable energy sources (RES) that do not fit easily in any of the previous sections.

## 3.6.1 Wind power potential

Wind power generation is directly dependent on the availability of wind. If the wind speed is below the cut-in speed (around 3 m/s) or above the cut-out speed (around 20-25 m/s), wind turbines are switched off. Between those values, higher wind speeds allow for higher power production. Wind patterns are affected by large-scale circulation changes due to global warming. Past records of wind patterns have shown varied trends throughout European regions.

Many recent studies have assessed changes in future wind energy potential in Europe under climate change (Tobin et al., 2014, 2016; Reyers et al., 2016; Carvalho et al., 2017; Davy et al., 2018; Moemken et al., 2018; Scott Hosking et al., 2018). These studies have used different emissions scenarios, combinations of global and regional climate models, dynamic and statistical downscaling techniques, bias correction methods and wind turbine characteristics. The studies agree that climate change will have a small effect on the overall wind energy potential in Europe, with changes at the European level in the range of ±5 % during the 21st century. Several studies suggest an increased potential at the European level in winter, but a decrease in summer and autumn. Local and regional changes in annual wind energy potential can be up to ±15 %, with changes up to ±30 % possible for individual seasons. Most studies project an increase in the Baltic Sea region and a decrease in southern Europe whereas both increases and decreases have been projected for other regions.

While there is considerable uncertainty around specific changes in regional wind power potential, the overall magnitude of changes is limited. Therefore, the impacts of climate change should neither undermine nor favour wind energy development in Europe. However, accounting for climate change effects in particular regions may help to optimise wind power development and energy mix plans (Tobin et al., 2014). For example, the Baltic Sea region might become more attractive for wind power development, in particular offshore wind power. In contrast, a renewable electricity strategy for southern European countries or regions may wish to combine wind power with solar power, and possibly other RES, to improve the stability of the electricity supply throughout the year. Climate services can inform siting decisions by providing more detailed information on current and future wind resources.

## 3.6.2 Solar power potential

Solar irradiance, which is affected by changes in cloud coverage and atmospheric water vapour content, is the most important determinant of solar energy potential. Available studies have found both small increases and decreases in solar irradiance and solar PV potential in Europe as a result of climate change (Jerez et al., 2015; Wild et al., 2015). A recent study has found that global and regional climate models do not agree on the direction of future changes in cloud coverage and insolation in Europe (Bartók et al., 2017). Hence, current projections for changes in solar irradiance in Europe cannot be considered robust. The efficiency of PV cells decreases by about 0.4 percentage points per degree of increase above 25 °C of cell temperature (Kaldellis et al., 2014). Therefore, solar PV is slightly affected by ambient temperature and wind speed, as both can have an impact on cell temperature. In summary, the impacts of climate change should neither undermine nor favour the development of solar PV in Europe.

Concentrated solar power (CSP) is more suitable for delivering baseload electricity than solar PV, because the medium heated by the concentrated rays of the sun can store heat for several hours. In contrast to solar PV, CSP benefits from increasing temperatures. The only available study at the European level projects significant increases in solar CSP potential in Europe, with increases in the range of 10-20 % for various European countries by the middle of the century (Wild et al., 2017). However, current CSP technologies require substantial amounts of freshwater for the steam turbines (INEA, 2018). Currently, Spain is the only European country where CSP is deployed commercially. CSP has commercial potential in several Mediterranean countries, but expansion may be constrained by insufficient water availability. Adaptation options that reduce water consumption include alternative cooling technologies, such as dry cooling and hybrid cooling (Frisvold and Marquez, 2014).

## 3.6.3 Bioenergy potential

Climate change can affect bioenergy through changes in the growing season, suitable growing area and productivity of trees and energy crops. However, there is still a great deal of uncertainty about the compound effects of climate change and increasing atmospheric CO<sub>2</sub> conditions on various crops under field conditions. Generally, climate change allows warm-adapted tree species and warm-season crops to expand northwards in Europe, whereas southern Europe is projected to experience a decline in its suitability for forest growth and decreasing crop yields as a result of increasing heat and water stress (EEA, 2017b, Sections 4.4 and 5.3). Biomass availability from forestry can be negatively affected by increasing forest fire risk, particularly in southern Europe (de Rigo et al., 2017).

The two available European-level studies assessing climate change impacts on bioenergy crops found a similar geographic pattern. The first study predicted that the potential distribution of temperate oilseeds, cereals, starch crops and solid biofuels will increase in northern Europe due to increasing temperatures and decrease in southern Europe due to increased drought. Bioenergy crop production in Spain is identified as being particularly vulnerable to climate change. Mediterranean oilseeds and solid biofuel crops (e.g. sorghum, Miscanthus), currently restricted to southern Europe, are predicted to extend further north as a result of higher summer temperatures (Tuck et al., 2006). The second study found similar regional results, but it stressed the potential role of technological developments to mitigate the adverse effects of climate change (Cosentino et al., 2012).

Plans for expanding bioenergy production in water-scarce regions, in particular in southern Europe, need to consider the energy-water-land nexus in order to minimise conflicts with other users (see also Section 2.1.5). Some European power producers use biomass imported from outside Europe (see the case study in Section 4.5.4). This approach can be regarded as a measure for increasing the stability of supply. However, it raises important sustainability questions relating to direct and indirect land use changes potentially triggered by bioenergy production and, to a lesser degree, to emissions from long-range transport. Replacing monocultures with mixed forests and reducing clear-cut areas can contribute to climate-resilient forestry.

## 3.6.4 Supply security from renewable energy sources

One of the main concerns for an energy system based largely on renewable energies is the so-called dunkelflaute (i.e. an extended period of time with very low production of wind and solar energy). In regions where electricity use for heating is widespread, this problem is aggravated by demand peaks during cold winter periods (cold dunkelflaute). Reliable projections of changes in the occurrence of a (cold) dunkelflaute in Europe are not available. The dominant driver behind wind power variability in central Europe is the North Atlantic Oscillation (NAO). A negative phase of NAO is linked to the inflow of Arctic air, which can cause extended periods of low wind power generation combined with increased heating needs. One recent study found that the impacts of NAO variability on the wind and solar power generation in Europe are much larger than the mean impacts of climate change up to 2050. This finding suggests that, if the power system is able to cope with weather variability, it can also cope with climate change impacts (Ravestein et al., 2018).

A study on Germany suggests that a reliable and sustainable electricity supply based on RES that can cope with a cold *dunkelflaute* requires a combination of international interconnections and various flexibility options, including pumped storage hydroelectricity, battery storage (e.g. from electric cars), power-to-gas, and flexible gas and biomass power plants (Huneke et al., 2017). A recent International Renewable Energy Agency (IRENA) report provides a wider perspective on solutions for the integration of variable RES into the power system, which comprises innovative technologies, business models, market design and system operation (IRENA, 2019).

# 3.7 Cross-cutting impact assessments of the European energy system

This section reviews selected relevant assessments of climate change impacts, vulnerabilities and adaptation needs. The focus is on cross-cutting assessments of the European energy system as a whole or of the electricity system. The main purposes are to give an overview of the knowledge base and to guide the interested reader to relevant literature with further details. Selected reports from international organisations and national governments are presented in Chapter 4.

## 3.7.1 Model-based assessments of climate change impacts on the energy system

Several studies have conducted economic assessments of climate change impacts and

adaptation needs in the energy system. The best assessed topics are changes in overall and peak energy demand relating to heating and cooling (see Sections 3.2.2 and 3.2.3).

One of the first comprehensive assessments of the economic impact of climate change on the EU power sector was a study carried out for the Directorate-General for Energy of the European Commission (Rademaekers et al., 2011). This study was used as an input for a Commission staff working document accompanying the EU strategy on adaptation to climate change (EC, 2013a). An updated summary was later published as a journal paper (Lise and van der Laan, 2015). The study used a combination of literature review, stakeholder interviews and quantitative modelling to estimate the investment needs for adaptation in the EU power sector for various technologies, regions and time horizons for two scenarios. Among other things, the study includes assessments of the benefits and costs of a wide range of adaptation measures. The results suggest that the most severe impacts for most power technologies would result from flooding, that sea level rise could have a severe impact on offshore wind and that increasing air temperatures and storms could have a severe impact on grids. The study also found that further investment would be needed in climate change adaptation for hydropower in southern Europe to account for lower average precipitation. The quantitative results rely on wide-ranging assumptions, which are not always traceable. Some key results, such as that overall investment needs are dominated by climate-proofing offshore wind power against the impacts of sea level rise, are not supported by other studies.

Two studies from the **ClimateCost project** used a modified version of the POLES energy model to analyse the impact of climate change on the European energy system (Dowling, 2013; Mima and Criqui, 2015). On the demand side, this study considered changes in heating and cooling demand; on the supply side, it considered changes in the efficiency of thermal power plants and changes in hydropower, wind power and solar PV electricity output. According to these studies, demand-side impacts are larger than supply-side impacts. However, this study considered neither possible change in extreme weather events nor decarbonisation scenarios.

The recently concluded **PESETA III project** of the Joint Research Centre (JRC) assessed the economic impacts of climate change on several European sectors, including energy (Ciscar et al., 2018; Despres and Kitous, 2018). **The overall effect of the projected changes in heating and cooling energy demand on welfare in Europe was** 

# assessed as beneficial, but there were considerable regional variations.

Another recent JRC study conducted within the ENHANCE project combined model-based projections of changes in multiple climatic hazards with sensitivity assessments of different types of critical infrastructure in Europe (EU-28 plus Iceland, Norway and Switzerland) based on an expert survey (Forzieri et al., 2018). The study projects a sharp rise in the expected annual damage (EAD) from climate extremes on critical infrastructure in Europe, including 10 types of energy infrastructure, as a result of climate change. The multi-hazard damage for infrastructure in the energy sector is projected to rise up to 15-fold, from a baseline EAD of EUR 0.5 billion per year to one of EUR 1.8 billion (uncertainty range: EUR 1.1-2.8 billion) in the 2020s, EUR 4.2 billion (EUR 3.0-6.7 billion) in the 2050s and EUR 8.2 billion (EUR 5.0-10.7 billion) per year in the 2080s, respectively. These damages, and the resulting adaptation needs, are dominated by the impacts of droughts and heat waves, in particular in southern Europe, whereas the overall role of floods is projected to decrease. The wider societal and economic impacts of potential blackouts triggered by floods and storms were not considered in this study.

A study conducted within the IMPACT2C project used a consistent modelling approach to assess the impacts of three levels of global warming (1.5 °C, 2 °C and 3 °C) on wind, solar PV, hydropower and thermoelectric power generation potential in Europe (Tobin et al., 2018). The results suggest that climate change has adverse impacts on electricity production in most European countries and for most technologies. Hydropower and thermoelectric generation may decrease by up to 20 %, whereas impacts on solar PV and wind power potential are more limited (and more uncertain). In agreement with other studies, more severe impacts were generally found in southern Europe than in northern Europe. For most countries, greater integration of renewables could reduce the vulnerability of power generation to climate change.

## 3.7.2 Global and European literature reviews

The global and sectoral part of the Intergovernmental Panel on Climate Change (IPCC) Fifth Assessment Report (AR5) includes a section on energy, which covers energy demand, energy supply, transport and transmission of energy, and macroeconomic impacts (Arent et al., 2014, Section 10.2). The regional chapter for Europe includes two short subsections: one on energy production, transmission and use and one on bioenergy production (Kovats et al., 2014, Sections 23.3.4 and 23.4.5). This review highlights changes in hydropower potential and in bioenergy production potential, a decrease in the usable capacity of thermal power plants in summer, a decrease in heating energy needs and a large increase in cooling energy needs. All changes show substantial regional disparities, with beneficial impacts dominating in northern Europe and adverse impacts dominating in southern Europe.

A systematic literature review of 50 peer-reviewed publications on the impacts of climate variability and change on electricity supply in Europe was conducted within the **BASE project** (Bonjean Stanton et al., 2016). This review study points to robust negative impacts of climate change on thermal electricity generation throughout Europe. Impacts on renewable electricity generation show strong regional variation, and the available evidence is considered patchier.

A recent review study has critically reviewed the global academic literature on the impacts of climate change on the energy system, with a focus on supply-side impacts (Cronin et al., 2018). Almost half of the reviewed studies had a focus on Europe. Of these 57 studies, 17 addressed wind energy, 12 thermoelectric power plants, 10 hydropower, 8 bioenergy, 6 solar power, 3 wave energy and 1 transmission and pipelines. The study also reviewed how the impacts of climate change on the energy system are reflected in current integrated assessment models of climate change, and it identified research gaps. Two earlier reviews of the global literature had a more balanced coverage of energy supply and demand (Schaeffer et al., 2012; Chandramowli and Felder, 2014).

The EEA report *Climate change, impacts and vulnerability in Europe* included a dedicated section on energy (EEA, 2017b, Section 5.4). This section includes an indicator on HDDs and CDDs and a review of studies on changes in future energy demand, on changes in electricity production and on impacts on energy infrastructure. The EEA report presents a more recent and more comprehensive review of the relevant literature for Europe compared with the IPCC AR5, but the overall findings are consistent.

Another recent literature review was conducted within the **EU-CIRCLE project**, with a focus on climate change impacts on critical infrastructure in the energy sector (Varianou Mikellidou et al., 2017). Out of the 82 reviewed studies, 39 (47 %) focus on the impacts of climate change, 25 (30 %) discuss adaptation and resilience measures, and only 18 (23 %) discuss interdependencies with other sectors. The study also suggests an integrated risk management framework, which comprises three stages: (1) risk assessment; (2) assessment of interdependencies with other sectors; and (3) assessment of adaptation and resilience options.

Impact/topic	Quantity and quality of information	Key gaps
Supply side		
Hydropower	Good	Assessment of adaptation options
		<ul> <li>Relationship between water availability and actual power generation</li> </ul>
		Effects on overall energy systems
Wind power	Moderate	<ul> <li>Few Europe-wide studies, focus mainly on energy resources and not on energy system impacts</li> </ul>
Solar power production	Moderate	<ul> <li>Few Europe-wide studies, focus mainly on energy resources and not on energy system impacts</li> </ul>
Thermoelectric power	Good	Effect on individual plants has been studied, but not yet impact on energy system
Demand side		
Cooling and heating demand	Good	<ul> <li>Limited research on the interaction between extensive (e.g. expansion in air conditioners, change in building characteristics) and intensive margin (e.g. use of electricity conditional on the demand for appliances)</li> </ul>
		<ul> <li>Most analyses focus on the residential sector, with limited research in other sectors (e.g. industry, commercial, agriculture)</li> </ul>
		<ul> <li>Most analyses focus on degree days, limited assessment of other climate indicators including extreme weather events</li> </ul>
Policy challenges		
Decarbonising the European economy	Good	<ul> <li>Existing studies have a relatively strong focus on electricity supply, with less detail on industry, heating and demand</li> </ul>
Increasing energy security	Moderate	<ul> <li>In terms of the impacts of climate change on energy security of supply, relatively little research on adaptation options</li> </ul>
Increasing energy efficiency	Moderate	<ul> <li>Previous model studies have a relatively simple representation of energy efficiency improvements; only recently have model studies focused on the effect of behavioural change</li> </ul>
		How to support behavioural change is under-researched

## Table 3.3 State of knowledge and key gaps for climate change impacts on energy

Source: Adapted from Tröltzsch et al. (2018).

A brief review of the recent literature on climate change impacts on the European energy system was conducted for the EU long-term strategic vision A Clean Planet for All (EC, 2018q, Section 5.9). The COACCH project has reviewed the state of knowledge regarding climate change impacts and adaptation for key economic sectors (Watkiss et al., 2018; Tröltzsch et al., 2018). A summary of its findings is presented in Table 3.3.

# 3.8 Summary of adaptation needs, opportunities and option

Figure. 3.2 provides an overview of selected climate change impacts for each European region and each

energy system component (see Section 1.2 for details). Symbols are used to classify impacts according to the energy source affected (renewable, fossil and other) and their desirability from a societal perspective, considering economic as well as environmental factors (beneficial, adverse and mixed). Similar impacts for 'neighbouring' regions are combined. There are large differences between regions and between system components. Northern Europe experiences both beneficial and adverse impacts from climate change, whereas southern European regions experience only adverse impacts. The energy transformation component faces only adverse impacts, whereas all other components are projected to experience beneficial as well as adverse impacts, sometimes





Source: EEA.

in the same region. It should be kept in mind that Figure 3.2 presents only a small selection of climate change impacts on the energy system. Furthermore, the various impacts included are of different magnitude and societal relevance. Therefore, care should be taken when interpreting or comparing impacts affecting different regions or system components.

Table 3.4 complements Figure 3.2 by presenting adaptation options for the key climate change risks identified but without regional differentiation. The table shows that there are technical and/or management options to address most challenges that climate change poses to the energy system. However, many adaptation options are associated with additional economic costs and/or environmental impacts. Given that the appropriateness of a particular adaptation option can vary significantly by the specific item being affected and its location, depending on environmental, technological, regulatory and other factors, there is no 'one-size-fits-all' adaptation solution. More detailed information on adaptation options for the power sector is available in a World Business Council for Sustainable Development report (WBCSD, 2014, Tables 5 to 10).

Energy system component	Key climate change risk	Adaptation option
Primary energy production	Challenge of growing energy crops in some regions due to temperature	<ul> <li>Adapted agricultural practices (improved irrigation, more drought- and/or pest-resistant crops)</li> </ul>
	increase and/or reduced water availability	Relocation to different region
	availability	Shift to other (renewable) energy source
	Damage to offshore infrastructure (oil, gas, wind) from increased coastal and marine hazards and storms	Climate proofing of infrastructure
		Adapted maintenance and damage response mechanisms
		<ul> <li>Decommissioning of older offshore oil and gas infrastructures, for which adaptation is not cost-efficient</li> </ul>
	Reduced hydropower production from lower streamflow and/or reservoir levels	Adjusted hydropower management plans
		Shift to other (renewable) energy source
	Increased risk of damage to	Adjusted hydropower management plans
	hydropower stations and downstream risks from river flooding	<ul> <li>Retrofitting of hydropower plants (see the case study in Section 4.5.2)</li> </ul>
	Water shortages for CSP technologies	• More water efficient operations (see Box 2.2)
		<ul> <li>Dry and hybrid cooling technologies</li> </ul>
		<ul> <li>Incorporation of on-site energy storage facilities could compensate for sporadic energy production</li> </ul>
		<ul> <li>Investment in additional personnel for cleaning of mirror components could lead to more efficient water use</li> </ul>
	Increased risk of damage to various infrastructures from coastal and inland flooding	<ul> <li>Climate proofing of infrastructure (e.g. siting on higher ground, protective barriers, hardening, covers) (see the cas study in Section 4.5.4)</li> </ul>
		Adapted maintenance and damage response mechanisms
		<ul> <li>Improved forecasting and planning</li> </ul>
		<ul> <li>Beach profiling and ecosystem-based approaches to reduc coastal erosion and associated flooding</li> </ul>
		<ul> <li>Dyke construction, nature-based solutions (e.g. 'giving roor to the river') and component-based flood barriers</li> </ul>
		Identifying relocation options
	Changes in wind resources	<ul> <li>Placement of new wind farms, supported by climate services</li> </ul>

#### Table 3.4 Adaptation options for key climate change risks to the energy system

Energy system component	Key climate change risk	Adaptation option
Energy transformation	Efficiency losses of thermal power plants due to higher temperatures	Pre-cooling of air used in combustion
	Capacity and/or efficiency decreases in thermal power plants due to reduction of and/or warmer cooling water	More efficient water cooling systems
		Dry cooling technologies
		<ul> <li>Output losses can be compensated for through less water-intensive (renewable) power generation</li> </ul>
	Increased risk of damage from sea level rise, storm surges, flooding and wind storms	<ul> <li>Climate proofing of infrastructure (e.g. siting on higher ground, protective barriers, hardening, covers) (see the case study in Section 4.5.4)</li> </ul>
		Adapted maintenance and damage response mechanisms
		<ul> <li>Improved forecasting and planning</li> </ul>
		<ul> <li>Beach profiling to reduce coastal erosion and associated flooding</li> </ul>
		Dyke construction and component-based flood barriers
		Identifying relocation options
Transport, transmission, storage and distribution	Reduced efficiency of transmission and distribution lines (overhead and underground)	• Adjust thermal rating of equipment (see the case study in Section 4.5.5)
		<ul> <li>Changes to standards of operational assets (see the case study in Section 4.5.5)</li> </ul>
		<ul> <li>Equipment designed for higher temperatures, including high-temperature transformers, high-temperature low-sag conductors and gas insulated lines or substations</li> </ul>
	Damage to pipelines and other infrastructure in Arctic and mountain regions from melting ice and permafrost	<ul> <li>Climate proofing of infrastructure (e.g. reinforced foundations, hardening)</li> </ul>
		Adapted maintenance and damage response mechanisms
	Increased/reduced production	Adjusted hydropower management plans
	potential from pumped storage	Adjusted storage
	Increased risk of damages to infrastructure from sea level rise, storm surges, flooding, snow/ice and wind storms	<ul> <li>Climate proofing of infrastructure (e.g. moving cables underground, siting on higher ground, protective barriers, hardening, covers) (see case studies in Section 4.5)</li> </ul>
		<ul> <li>Adapted maintenance and damage response mechanisms (e.g. improved preventive maintenance of vegetation near overhead lines)</li> </ul>
Energy demand	Increased peak cooling demand	Greater use of demand management technologies
		<ul> <li>Modified building design for improved cooling</li> </ul>
		Efficiency standards for cooling devices
		Early warning systems to alter consumer behaviour
	Higher energy demand in water-scarce regions (for	<ul> <li>Increased energy generation capacity, favouring low water demand technologies (e.g. solar, wind)</li> </ul>
	desalination) and low-lying regions (pumping)	<ul> <li>Adapting existing flood control infrastructure to enable energy generation</li> </ul>

## Table 3.4 Adaptation options for key climate change risks to the energy system (cont.)

Source: EEA, based primarily on IEA (2016b) and Ecofys (2017).

# 4 Building a climate-resilient energy system

## **Key messages**

- There are both synergies and trade-offs between climate change adaptation, mitigation and wider sustainability objectives. The important connections between energy policy and other policy areas call for a comprehensive policy approach that considers multiple societal and policy objectives jointly.
- Businesses are key actors in the design and implementation of resilience-building measures in the energy system. Governments can facilitate such actions through a variety of measures, including policies and regulation, information and services, resilience of state-owned assets, and mobilising financial resources.
- Market actors in the energy system face a number of barriers that may impede the implementation of effective adaptation policies. Well-designed European and national policies can play a key role in overcoming these barriers.
- Key EU climate and energy policies and strategies promote the mainstreaming of climate change adaptation into energy
  policies. The recent evaluation concluded that the EU Strategy for adaptation to climate change has triggered adaptation
  actions in many areas. The development of the Energy Union provides an important opportunity for further integrating
  climate change adaptation in European and national energy planning. The EU also supports building climate resilience
  in the energy system by requiring climate-proofing of major new energy infrastructure, funding relevant research and
  innovation projects, and developing climate services for the energy sector as part of the Copernicus services.
- Almost all European countries have concluded a national climate change impact, vulnerability or risk assessment that covers the energy sector. Most countries also include energy as a relevant sector in their national adaptation strategies and/or plans. However, available government documents provide only limited evidence for the implementation of adaptation actions in the energy sector.
- Individual countries are facilitating adaptation in the energy system by providing guidelines for vulnerability assessment and resilience planning, through support for the development of weather and climate services, and through reporting obligations for infrastructure providers.
- Many energy utilities and network providers are already adapting their activities to observed and projected impacts of climate change. Case studies in this chapter include improved flood risk and water management by hydropower producers, strengthening the biomass supply chain, and increasing the climate resilience of power lines. Some of these activities have been triggered or facilitated by government policies and regulation.
- Various international governmental and sectoral organisations are supporting climate change adaptation by providing guidance, developing information sharing platforms, facilitating communities of practice and acting as catalysts for the development of relevant services.

Many different actors and actions are needed to ensure that the European energy system is adapted to changing climatic conditions and resilient to climatic hazards, now and in the future. This chapter discusses key adaptation types and actors, analyses barriers to action as well as approaches to overcome them, reviews the relevant EU policy context, and presents the relevant activities of European institutions and organisations, countries and other relevant actors.

## 4.1 Adaptation types, actors and barriers

#### 4.1.1 Types of adaptation actions

The typology of adaptation actions used in this report combines elements of adaptation typologies developed by international organisations (see Table 4.1). Adaptation actions are first grouped into building adaptive capacity and delivering adaptation actions. The first category is further divided into information and knowledge, and policies and framework actions, and the second category into management and physical actions. Physical adaptation actions comprise both 'grey' and 'green' actions, as distinguished in a generic EEA typology (see Table 4.1 for a description). 'Green' options can play an important role in reducing risks from climate-related hazards while achieving various co-benefits (EEA, 2017a; OECD, 2018). However, two recent reviews of adaptation options in EU energy policy have identified hardly any 'green' options, which suggests that they have a limited role in strengthening the resilience of the energy system (Capriolo, 2016, section 3.2.5; Ecofys, 2017).

## 4.1.2 Adaptation actors

Many different stakeholders can potentially play a role for climate change adaptation in the European energy system. Figure 4.1 provides a mapping of key stakeholders across the four components of the energy system (see Section 1.2), distinguishing

'key influencers' and 'key actors'. The specific roles of these stakeholders, including the division between public and private actors, differs across European countries reflecting different national energy models (Dallamaggiore et al., 2016; IEA, 2016b; Russel et al., 2018).

Key actors are accountable for one or several activities in the energy system, depending on the regulatory framework and the level of vertical integration in the energy sector. The key actors and their roles are:

- **Primary energy extractors** extract fossil or nuclear fuels.
- Producers and generators produce the power, refined fuels, heat and cold for final use. This may include producers that also operate energy storage facilities, such as pumped hydro plants;
- Transmission system operators (TSOs) and distribution system operators (DSOs) operate the national, regional and local grids, pipelines and networks for distributing power, fuel

Source	Category	Sub-category	Description
EEA (this report)	Building adaptive capacity	Information and knowledge	Collecting and monitoring data, providing climate services, risk assessments, producing guidelines and other measures that facilitate targeted adaptation
		Policies and frameworks	Actions taken by public actors, such as legislation, regulation and standards, which provide obligations, incentives or funding for adaptation actions by organisations or the wider public
	Delivering adaptation actions	Management	Modifying management procedures of relevant public and private actors (such as infrastructure providers)
		Physical	Technological or engineering solutions such as siting, building or strengthening of assets; ecosystem-based approaches (where relevant) are also included
World Bank (Ebinger and Vergara, 2011)	Building adaptive capacity		Collecting and monitoring data, research and awareness raising in order to facilitate the improvement of knowledge systems, improving the capacities of local institutions, forming of partnerships, and supporting public governance
	Delivering adaptation actions		Preventing or reducing risks, sharing responsibility for losses or risks, or creating opportunities
IEA (IEA, 2016b)	Technological and structural measures		Enhancing asset resilience through implementing technological measures to mitigate damage/ reduce losses, or by taking out structural reinforcements or adjustments
	Management and siting measures		Modifying management procedures, placing structures in areas deemed less prone to climate change risks
EEA (EEA, 2013)	Grey actions		Technological or engineering solutions such as building or strengthening of assets
	Green actions		Ecosystem-based approaches that use the services of nature to improve resilience
	Soft actions		Managerial, legal and policy approaches to alter behaviour and governance

## Table 4.1 Typologies of climate change adaptation actions

and/or heat to consumers. They can also be involved in energy storage. TSOs and DSOs (whether public or private) have a key role in implementing physical adaptation actions.

- Consumers influence overall energy demand through their consumption patterns, including those resulting from climate change. Consumers can also have a role in the market, which is important when costs of adaptation measures taken by the energy sector are passed on;
- **Prosumers** are actors that both produce and consume electricity (e.g. a home fitted with PV panels that supplies power to the grid during the day and consumes power from the grid in the evening). Prosumers represent an increasingly important role in achieving the energy transition, but an efficient integration into the European energy system as a whole is lacking (Sajn, 2016; EC, 2017d).

 Other market participants, such as retailers and aggregators, play an important role in the energy system by bridging between producers, TSOs and DSOs, and consumers and prosumers.

Key enablers are stakeholders outside the 'classical' energy sector who influence the operation of the energy system in different ways. The key influencers and their roles are as follows:

• EU, public authorities (national, regional and local government) and regulators are responsible for setting the policy and market frameworks for the European energy system. Among others, regulators can also have the crucial role of determining who pays for resilience building. The public sector also takes a role in fostering innovation in the energy sector, for example through making funding available for research, development and innovation.



- Research and knowledge institutions can influence (adaptation in) the energy system through the provision of relevant knowledge, information and ideas. They include academic research institutions, climate service providers and other knowledge providers.
- Non-governmental organisations (NGOs), media and think-tanks can influence the energy system through their support of certain narratives, movements and policy positions.
- Finance and insurance covers all public and private institutions involved in investing in or insuring the energy sector. They can influence adaptation in the energy system by requiring particular standards or reporting requirements for financing or insuring energy infrastructure.
- Standards organizations develop technical standards, including for energy infrastructure. Standards can be voluntary or binding, when they have been adopted by a regulator. Even voluntary standards can have a large influence if they are widely adopted by industry or by finance and insurance institutions.
- Builders, manufacturers and retailers have a strong influence on the energy demand of buildings and energy consuming goods. The products and technologies they offer can influence the pace of adoption of new technologies.

## 4.1.3 Role of governments

Building climate resilience in the energy system inevitably raises the question who is responsible for delivering adaptation actions. The question on the role of the EU, national and subnational governments, and other public authorities is particularly for the energy system relevant where a critical infrastructure is owned by a wide range of organisations, including private actors, public-private partnerships and local utilities. The debate on the role of governments has been interpreted somewhat differently within Europe. For example, the German adaptation strategy states that 'adaptation to climate change in the energy sector is essentially a task for the energy industry itself', entailing that 'federal and Länder authorities may be able to provide assistance, contribute knowledge and set new regulatory trends' (Government of Germany, 2008, p. 33). Other countries (e.g. Denmark and the United Kingdom) assign a stronger role to public authorities for

fostering, facilitating and monitoring adaptation by private actors (Russel et al., 2018, Section 6.3).

As a starting point, infrastructure providers have a self-interest in protecting their infrastructure and in providing secure services to their customers. However, some unique characteristics of the energy system need to be considered here. First, essential components of the energy system, such as electricity transmission and distributions networks, are (quasi) monopolistic, i.e. there are no competitive markets. Second, the functioning of the energy system in general, and the electricity system in particular, requires the efficient cooperation of many different infrastructure providers across borders. Hence, an individual actor does not necessarily have the information, or incentives, to manage its infrastructure in a way that ensures the stability of the whole system. Finally, a stable energy supply is essential for almost all aspects of modern life, and supply interruptions can lead to very large costs to society (see Box 4.1 for further information). Considering these factors, academic studies as well as reports by international organisations have concluded that governments and public authorities have a strong role in ensuring a secure (as well as affordable and sustainable) energy supply. This includes building climate resilience in the current and future energy system (Schneider, 2014; Cortekar and Groth, 2015; Russel et al., 2018, section 6.3).

Businesses are key actors in the design and implementation of resilience-building measures in the energy sector. Governments can and **should facilitate such** actions through a variety of measures, including policies, information and services, the resilience of state-owned assets, and mobilising financial resources (see Figure 4.2) (IEA, 2015, 2016b). The influence of public policies on critical infrastructure providers can be distinguished between regulatory policies, which are most suitable for incremental adaptation, and developmental or enabling policies, which foster the experimentation required for transformational adaptation (Russel et al., 2018, Section 6.3). Regulatory approaches can range from facilitating voluntary self-regulation through enforced self-regulation, negotiation and delegation to an agency to command and control (Schneider, 2014). Careful consideration of synergies between existing energy policies and regulations and required adaptation measures can help public authorities to find the most efficient policy approach for mainstreaming adaptation action in key sectoral policies (Capriolo, 2016, Section 3.2.5).

### Box 4.1 The societal costs of blackouts

The key argument for the strong role of governments in ensuring a secure energy supply is that the costs of a supply interruption to society are much higher than the costs to the energy sector itself. The societal costs of large-scale blackouts in Europe can easily reach billions of euros (see Box 3.2 for details) (Walker et al., 2014; Küfeoğlu, 2015; Growitsch et al., 2015; Gündüz et al., 2017).

The cost to society of the interruption in the provision of electrical power is commonly measured in terms of value of lost load (VoLL), which can be understood as an economic indicator for the value of electricity security. VoLL estimates are highly sensitive to the characteristics of the outage, in particular the timing and duration, the location, sector/and or social grouping affected, and whether it is planned or unexpected.

A recent literature review of VoLL estimates to households in Europe revealed a wide range, from EUR 0.82 to 66/kWh. Most of these estimates were based on surveys, which used either a willingness-to-pay or willingness-to-accept approach. The same study also applied a production-function approach to develop more consistent VoLL estimates for households. This method led to country estimates of between EUR 3 and 16/kWh, with an EU average of EUR 9/kWh (Shivakumar et al., 2017). Notably, the national VoLL estimates based on the production-function approach are between 25 and 100 times higher than the electricity prices for household consumers, with an EU average of EUR 40/kWh (Eurostat, 2019a).

A study for the British Government used both economic modelling and international case studies to estimate the economic and social costs of electricity shortfalls in the United Kingdom. This study also found very high levels of uncertainty within existing estimates of VoLL in the United Kingdom, and it stressed that VoLL should be regarded as a range rather than a single value (Walker et al., 2014). The study did not conclude with a particular range (nor a central value) for VoLL, but the figures cited within the study are compatible with the range of the EU-wide study cited above.

## Figure 4.2 Role of governments in enhancing energy sector resilience



Sources: Reproduced with permission from IEA (2016b). © OECD/IEA Energy, Climate Change and Environment: 2016 Insights, IEA Publishing. Licence: www.iea.org/t&c

#### Box 4.2 Avoiding maladaptation in the energy system

Avoiding 'maladaptation' is increasingly discussed in the academic literature and in policy discourses around adaptation. However, maladaptation is a complex concept, and many competing definitions exist (Barnett and O'Neill, 2010; Magnan et al., 2016). In a broad sense, maladaptation refers to adaptation strategies or actions with adverse negative effects that are more serious than the climate impacts being avoided, in other words, to 'solutions that are worse than the problem' (Scheraga and Grambsch, 1998). Negative side effects can refer to disproportionate environmental, economic and/or social costs of an action. These side effects may occur further in the future (i.e. actions that reduce short-term climate impacts at the cost of increasing long-term vulnerability); they may also affect other people and locations (e.g. flood protection measures at a river that increase flooding risks downstream) or other sectors and policy objectives (e.g. adaptation measures that vastly increase greenhouse gas emissions).

The IPCC Fifth Assessment Report has devoted a whole section to maladaptation, which includes examples of actions that may be considered maladaptation and discusses causes of as well as experiences with maladaptation (Noble et al., 2014, Section 14.6). Assessments of (mal)adaptation are unavoidably subjective. Hence, the question of whether a particular 'solution is worse than the problem' cannot always be answered unequivocally.

In the context of this report, maladaptation in the energy sector refers to strategies and actions that give excessive priority to an energy adaptation goal over concurring policy goals. An important approach for avoiding maladaptation is to develop clean energy strategies that address climate change mitigation and adaptation perspectives jointly in the context of wider sustainability concerns, considering the interactions of the energy system with water and land systems.

## 4.1.4 Links between climate change adaptation, mitigation and wider sustainability objectives

From an environmental policy perspective, climate change mitigation is driving the transformation of the European energy system, but this transformation needs to be flanked by adaptation and resilience-building measures. Climate change adaptation in the energy system aims to ensure that the main goals of the EU Energy Union, namely a secure, sustainable and affordable energy supply, can be achieved in a changing climate. Furthermore, the energy system may compete for water and (arable) land with other sectors and systems, such as agriculture, forestry and natural ecosystems (for a more detailed discussion of the energy-water-land nexus, see Section 2.1.5).

The important connections between energy and other sectors, systems and policy areas call for a comprehensive policy approach that considers multiple societal and policy objectives jointly. For example, a mitigation strategy that does not consider adaptation needs (e.g. resulting from decreasing water availability) may result in a more expensive or less secure energy supply in the future. At the same time, an adaptation strategy developed in isolation can make it more difficult or costly to achieve mitigation and other policy objectives. Such adaptation measures are often termed 'maladaptation' (see Box 4.2).

There are both synergies and trade-offs between climate change adaptation, mitigation and wider sustainability objectives, as expressed in sectoral EU policies or the 17 Sustainable Development Goals of the 2030 UN Agenda for Sustainable Development (UN, 2015). For example, measures to improve energy efficiency typically have mitigation as well as adaptation benefits, because reduced energy demand reduces not only greenhouse gas (GHG) emissions but also the overall demand for adaptation measures in the energy system. In contrast, end-of-pipe mitigation technologies, such as carbon capture and storage, typically achieve mitigation objectives while increasing adaptation and wider sustainability challenges, because they increase primary energy demand and potential conflicts around scarce water and land resources. Finally, some adaptation policies in the energy sector can increase the mitigation challenge. For example, dry cooling can reduce the water need of a fossil fuel power plant by more than 90 % compared with wet cooling (Macknick et al., 2012), but it decreases its thermal efficiency, thereby leading to higher costs, higher GHG emissions and sustainability challenges in the upstream fuel chain. Table 4.2 shows selected mitigation measures with co-benefits or potential trade-offs with adaptation.

Trade-offs between different policy objectives, such as climate change mitigation and adaptation, cannot be avoided. However, **policymakers and planners need to consider such trade-offs carefully in order to develop a mix of measures that achieves various policy objectives simultaneously**, without undue adverse side effects. The choice of particular energy technologies needs to reflect the local climatic, environmental and other conditions. For example, a renewable energy source with high water requirements can be appropriate in a region with abundant water, but it could make adaptation excessively difficult in a water-constrained region where further decreases in water availability are projected. The **International Resource Panel** has

Mitigation measures with co-benefits for adaptation	Mitigation measures with potential trade-off with adaptation	
Wind and solar PV: lower water use than thermal generation	Inappropriate expansion of biofuels: could exacerbate competition for land and water with agriculture and	
Thermal insulation of buildings: reduces energy demand and improves thermal comfort	biodiversity conservation Hydropower expansion: could increase the complexity of	
Distributed renewable power generation supported by good interconnectivity: reduces disruption of outages	managing water resources Carbon capture and sequestration: increases	
Energy storage: supports intermittent renewable energy sources and provides back-up during demand peaks	cooling water consumption Nuclear power: risk of outages during heat waves and droughts	

#### Table 4.2 Synergies and trade-offs between mitigation and adaptation in the energy system

#### Source: Adapted from EC (2018q).

recently published two thorough reviews of the wider environmental and resource implications of low-carbon energy technologies and of demand-side technologies for reducing energy demand (UNEP, 2016, 2017). These reports provide valuable information for developing low-carbon strategies in a wider sustainability context.

Recent EU policies place a strong focus on the mainstreaming of climate change mitigation and adaption in EU and national planning and reporting processes (see Section 4.2). It is important that European countries follow up when developing their national energy strategies and plans. The brief analysis of national activities in this report (see Section 4.3) suggests that many European countries are still in an early phase of addressing the adaptation needs of their rapidly evolving energy systems.

## 4.1.5 Adaptation barriers — and ways to overcome them

Planning and implementing climate change adaptation policies and measures faces a number of barriers (Moser and Ekstrom, 2010; OECD, 2015). These barriers may impede the implementation of effective and efficient adaptation policies unless targeted efforts are undertaken to overcome them. Some barriers are generic for most adaptation actions (e.g. uncertainty about some aspects of future climate change), whereas others are specific to the energy system or to specific actors in it (e.g. regulatory context). The energy sector remains highly scrutinised by policymakers and regulators given the strategic importance of energy for modern societies. The policy framework, through its strategic, legal and regulatory mechanisms, can help or hinder action on climate change adaptation. Well-designed policies play a key role in addressing adaptation barriers, as they provide the framework for action by other key actors.

This section gives a brief overview of important barriers to adaptation in the energy system and presents some

generic ideas for addressing these barriers, in particular through public policies. This overview is by no means comprehensive; it is intended primarily to give an indication of how public policies can address some commonly cited challenges for adaptation. Non-state actors such as financing and insurance institutions can also play an important role in facilitating adaptation in the energy system, but their role is not discussed in detail in this report. Many examples of how public and private actors at different levels are successfully building climate resilience in the energy system are presented in the following sections of this chapter.

## Awareness of the (potential) need for adaptation

Adaptation requires that a (potential) adaptation actor is aware that past and/or future climate change necessitates doing certain things differently from how they would be done in a stationary climate. Hence, awareness of a potential problem (or opportunity) is a crucial first step in the adaptation policy cycle. Such awareness can come from, among other things, past experiences with managing climate or weather extremes, peer learning, stakeholder and expert meetings, government-led or other climate change impact, vulnerability and risk (CCIV) assessments reports, or a combination thereof.

Governments can help raise awareness of climate change impacts and related adaptation needs among (potential) adaptation actors through targeted CCIV assessment reports, stakeholder events, mandatory climate risk screening in infrastructure planning and approval, reporting requirements for critical infrastructure providers, and other activities.

## Understanding the specific climate change impacts and adaptation challenges

Understanding the current and/or future impacts of climate change on a particular strategy, asset, service or activity is another key requirement for targeted

and effective adaptation action. Important progress has been made in recent years in understanding past climate change, in projecting future change and in understanding its impacts on ecosystems, economic sectors (including the energy sector) and society as a whole, through substantial efforts by many actors at the international, European, national and subnational levels. As a result, many projections of climate change and its impacts in European regions have become more robust. However, information about local-scale changes in climate characteristics relevant for stakeholders in the energy system is still not readily available.

Many actions mentioned above under 'raising awareness', such as CCIV assessments targeted at the energy system, are also relevant for improving understanding of specific climate change impacts. **The development of user-friendly climate services is an important strategy for closing the gap between information needs and availability** (see Sections 4.2.4 and 4.3.3 for further information). Such climate services should provide quality-controlled downscaled climate projections, preferably translated into variables directly relevant to energy stakeholders. Establishing communities of practice that bring energy system stakeholders together with climate experts are further ways in which governments can address the information challenge.

## Decision-making under conditions of uncertainty

All climate change projections are subject to uncertainty. The level of uncertainty depends, among other things, on the climate variable, the spatial resolution and temporal aspects of the specific projection of interest (see Section 3.1). Furthermore, climate change adaptation in the energy system occurs in a dynamic environment that is characterised by rapid, and sometimes unpredictable, changes in the political, regulatory, economic, social and technological context (see Sections 2.2 and 4.2). Although some uncertainties can be reduced by developing and sharing better information (e.g. through targeted climate services), adaptation actors unavoidably need to take decisions in a context of multiple uncertainties.

There are many approaches for (adaptation) decision-making under uncertainty, including robust decision-making, real option analysis and the adaptation pathways approach (Haasnoot et al., 2013; Capela Lourenço et al., 2014; Watkiss et al., 2014; OECD, 2015). Their suitability in a specific context depends on various factors, such as the availability of quantitative information, the type and magnitude of uncertainties, the importance of climate change compared with other factors, the potential costs and risks of acting (too) early versus acting (too) late, and the reversibility of specific adaptation actions.

Most stakeholders in the energy system have experience in decision-making under conditions of uncertainty as part of their everyday business. Governments can facilitate the appreciation of the 'novel' uncertainties related to long-term climate change by developing probabilistic climate projections, adaptation platforms (including decision-making tools) and targeted guidance material, possibly in collaboration with sector associations. Communities of practice can help make this knowledge actionable through mutual learning.

## Identification and assessment of adaptation options

Targeted adaptation requires adaptation actors to be aware of potential adaptation options and to have some understanding of their costs and benefits (in a broad sense). Adaptation options differ widely depending on the adaptation actor and the challenge they are facing (see Chapter 3). Some of them can be addressed by minor changes in established management practices, others involve technological changes associated with increased costs, and still others require fundamental changes in how and where things are done.

An assessment of adaptation options requires detailed knowledge of the specific decision context, which is often available only to the adaptation actors themselves. Governments can facilitate this assessment through some of the abovementioned measures, including targeted adaptation platforms and communities of practice. Reporting requirements and other ways of information sharing can help actors to learn from each other. Furthermore, targeted research and innovation funding can help develop new adaptation options and services or reduce their costs for individual actors.

## Creating a 'business case' for adaptation

Most adaptation actions in the energy system will be implemented by market actors. **Many adaptation options make a 'business case' for individual actors** (i.e. the net present value of their expected benefits outweighs the costs) because they reduce damages or other financial losses, improve revenues, avoid reputation loss or provide other tangible benefits. Hence, a well-informed actor will implement such actions without the need for policy interference. However, some adaptation actions do not make a business case under 'pure' market conditions, because their costs and benefits are unevenly distributed across actors and/or time. One reason is market externalities. For example, the costs of a power blackout to a modern society are much higher than the immediate revenue losses to the power producers or network operators (co-) responsible for it (see Box 4.1). Another reason is system complexity. For example, the stability of the electricity supply in most European countries depends on the cooperation of many actors from different countries. Hence, ensuring or increasing network stability under changing climatic conditions may require coordination among many actors in order to be effective. The importance of efficient coordination has increased as a consequence of the liberalisation of energy markets in Europe, which may mean that energy companies are no longer vertically integrated. Finally, the market discount rates commonly applied in investment decisions may not be compatible with the long asset lifespan of energy infrastructure and the long-term horizon of climate change impacts and adaptation benefits.

## Governments and EU institutions play an important role in facilitating adaptation actions that are

beneficial from a whole-society perspective in situations in which markets provide insufficient incentives to individual actors to achieve socially desirable levels of adaptation. They can do this through additional market incentives, design standards, regulations at national and/or European level, public-private partnerships and other measures. The design, implementation and control of such policy interventions requires considerable expertise in order to avoid inefficiencies and undesired side effects. For example, regulators with price setting or advisory powers significantly influence investments by energy utilities and the way in which investment costs are passed on to consumers. If the criteria used by regulators to assess the permissible investments and cost transfer do not take climate change adaptation into consideration, any extra costs associated with increasingly resilient investments may not be transferred to consumers, generating a further barrier to such investments (Government of the UK, 2012). Box 4.3 provides additional information on barriers to adaptation, as reported by stakeholders from the United Kingdom. Information asymmetry between

## Box 4.3 Barriers to adaptation as reported by British energy stakeholders

Critical infrastructure and service providers in the United Kingdom, including energy regulators, utilities and network operators, have an obligation to report on their climate change risks, adaptation actions and plans, including the barriers encountered (see Section 4.3.3). These reports have allowed for an analysis of barriers to adaptation as perceived by these stakeholders (Government of the UK, 2012, 2017; Street et al., 2016, 2017).

Four barriers to effective adaptation were identified within the current institutional framework in the United Kingdom:

- 1. the lack of quantified objectives, or clear adaptation policy goals and targets at a national or local level;
- 2. the 'lack of coordination and unclear goals' due to complex governance arrangements, involving multiple partners, along with a lack of definition of risk ownership, which can lead to 'fragmented, limited or non-existent adaptation actions';
- 3. the vulnerability of the adaptation delivery mechanism due to the financial fragility of the network of organisations supporting the government and the relevant administrations in delivering adaptation; and
- 4. the lack of a proper assessment of the 'resources and capacity needed to deliver adaptation at the local level'; in many instances, the 'responsibility for delivering adaptation is increasingly devolved to the local level', but local decision-makers often lack adequate skills and knowledge to tackle the challenges brought about by adaptation to climate change.

The reports on the energy sector converged on a few classes of barriers, which appear to be of general relevance beyond the United Kingdom:

- the limited suitability of the available knowledge base in climate change to provide reliable information at the fine level of geographical resolution required to carry out adaptation actions in practice;
- the interdependence with adaption needs in other sectors, particularly in the case of prevention of flooding; and
- the constraints brought about by economic considerations, such as the impacts of the costs of adaptation options on the economic viability of the companies affected, and the uncertainty surrounding the actual magnitude of such costs and hence the actual availability of funds needed to cover them.

private actors in the energy sector and policymakers developing the adaptation policy framework can be a challenge for efficient and effective market regulation.

## 4.2 European Union policies and actions

European energy policy and climate change adaptation policy have largely been developed separately. Recently, **considerable efforts have taken place to mainstream climate change adaptation into EU sectoral policies, including energy policy** (Capriolo, 2016; EC, 2018i, Annex XI). This section gives an overview of EU policies and actions that are relevant to adaptation in the energy system. Euratom policies, including guidelines on the safety of nuclear power plants, are not considered here, because they fall outside the scope of this report (see Section 1.2).

## 4.2.1 European Union adaptation strategy and related activities

European adaptation policy is driven by a mix of EU and national policy and strategies. The EU strategy on adaptation to climate change, adopted in 2013, forms the backbone of adaptation activity in the EU (EC, 2013b). It has three objectives and eight key actions, which are shown in Table 4.3. The EU adaptation strategy includes several supporting documents. Among them, the Commission staff working document on adapting infrastructure to climate change (EC, 2013a) includes a section with a specific focus on the energy transmission and distribution grid.

The EU adaptation strategy has been evaluated recently against the five 'better regulation' criteria by measuring progress for each of its eight actions (EC, 2018v, 2018h, 2018i). This evaluation may lead to a revision of the strategy in the future. The overview below summarises selected findings from the evaluation of the EU adaptation strategy that are relevant for the energy system. For further information, see EC (2018i, Sections 4 and 7).

Action 1 of the EU adaptation strategy concerns the adoption of comprehensive adaptation strategies by all Member States. The Commission has provided guidelines to support Member States in formulating their adaptation strategies. The Commission has also developed an 'adaptation preparedness scoreboard' with key indicators measuring Member States' readiness levels, which was applied in the evaluation of the EU adaptation strategy (EC, 2018h, 2018i, Annex IX). According to this scoreboard, only nine Member States are actively promoting adaptation in the energy sector. Further information on national adaptation action is provided in Section 4.3.

Action 2 has triggered substantial funding for climate change adaptation through LIFE projects. However, none of these projects addresses the energy sector, which is outside the scope of the LIFE programme.

Ob	jectives	Act	tions
1.	Promoting action by Member States	1:	Encourage all Member States to adopt comprehensive adaptation strategies
		2:	Provide LIFE programme funding to support capacity building and step up adaptation action in Europe
		3:	Introduce adaptation in the Covenant of Mayors framework
	Promoting better informed decision-making	4:	Bridge the knowledge gap
		5:	Further develop Climate-ADAPT as the 'one-stop shop' for adaptation information in Europe
3.	Promoting adaptation in key vulnerable sectors	6:	Facilitate the climate proofing of the common agricultural policy, the cohesion policy and the common fisheries policy
		7:	Ensure more resilient infrastructure
		8:	Promote insurance and other financial products for resilient investment and business decisions.

Table 4.3	Objectives and actions of the EU Adaptation Strategy
-----------	--

Note: Climate-ADAPT, the European Climate Adaptation Platform.

**Source:** Adapted from EC (2018q).

Action 3 addresses the Covenant of Mayors for Climate and Energy. By April 2018, 1 076 Covenant signatories from 25 EU Member States, covering around 60 million inhabitants, had committed to conducting vulnerability and risk assessments and to developing, implementing and reporting on adaptation plans (Covenant of Mayors Office, 2015). Across the EU, around 40 % of cities with more than 150 000 inhabitants are estimated to have adopted adaptation plans to protect citizens from climate impacts (EC, 2018v; Reckien et al., 2018). However, there is no systematic information available on the extent to which cities are addressing energy in their adaptation plans.

Action 4 addresses EU research funding and how this is targeted. It has sought to better identify the adaptation knowledge gaps and then to adjust the programming of EU research funding, especially Horizon 2020, to address these gaps. It has also promoted EU-wide vulnerability assessments through the Joint Research Centre. Ultimately, this action is expected to improve adaptation actions, including in the energy system. The Commission has recently published a policy booklet presenting a selection of its research, science and innovation projects on climate change adaptation (EC, 2018e). Further details on this topic with a focus on the energy system are provided in Section 4.2.4; selected results are presented in Section 3.7.

Action 5 focuses on the development of the European Climate Adaptation Platform, Climate-ADAPT (<sup>7</sup>), a knowledge platform for sharing key adaptation-relevant information and best practice examples (EEA/EC, 2019). Climate-ADAPT was relaunched in March 2019 with new functionalities and additional information. Information on Climate-ADAPT is relevant for the energy sector as well. **The case studies presented in Section 4.5 of this report are all published in more detail on Climate-ADAPT**.

Action 6 includes the major structural funds in the EU, the European Regional Development Fund (ERDF) and the Cohesion Fund (CF). Mainstreaming of climate change considerations can have relevant impacts on these projects, including those linked to the energy sector. Within such funding, major projects must integrate climate change adaptation considerations throughout the preparatory and implementation

stages. The guidelines for the individual programmes focus on the methodological approach to vulnerability and risk assessment for major projects, something that is also relevant for energy sector infrastructure. A report on climate change and major projects outlines climate change-related requirements and guidance for major projects in the 2014-2020 programming period with relevance for the energy sector (EC, 2016a).

Action 7 is based on the development of technical standards, strategic guidelines, mandatory climate risk assessments and climate services targeted to the energy, transport and buildings sectors. Appropriate technical standards can assist infrastructure providers in alleviating the adverse impacts of climate change on the operational, financial, environmental and social performance of their infrastructure. This can be achieved by ensuring that infrastructure is climate resilient and has sufficient safety margins. Standards related to climate-sensitive infrastructure should be periodically revised in order to trace the development of climate change. The European standardisation organisation, CEN, and the European electrotechnical standardisation organisation, CENELEC, have an EU mandate to review existing infrastructure standards with adaptation relevance, including in the energy sector (see Box 4.4 for further information). According to current Commission proposals, a wider range of EU-funded infrastructure investments needs to be climate proof during the 2021-2027 EU budget period (EC, 2018v, footnote 51).

Action 8 has seen some progress in the insurance and broader financial sector. Institutional investors, including insurers, may finance investments in the energy sector. They typically focus on re-financing, acquisitions and secondary market activities, essentially taking on established projects from project developers (IRENA, 2018a). They have an important interest in ensuring that the investments they acquire are climate resilient. Action 8 resulted in setting up a high-level expert group on sustainable finance (with EEA participation), the findings of which are at the heart of a Commission action plan on sustainable finance (EC, 2017e, 2018p). These activities aim to redirect private sector investment towards climate change mitigation and adaptation, and resilience (EC, 2018v, footnotes 62 and 63). Further information on this topic is available in Section 4.2.3.

#### Box 4.4 CEN and CENELEC adapting European standards to climate change

**CEN** and **CENELEC** are responsible for numerous product and service standards in the EU. They have taken up a mandate under Action 7 of the EU adaptation strategy to integrate climate change adaptation in technical standards (CEN-CENELEC, 2018). This work is coordinated by the CEN-CENELEC Adaptation to Climate Change Co-ordination Group.

CEN-CENELEC have published a guide for addressing climate change adaptation in standards (CEN-CENELEC, 2016). This guide provides a checklist and decision tree to ensure that standards develop and enhance climate resilience. CEN-CENELEC are also reviewing and revising infrastructure-related standards. In the first phase of the work, the European standards were screened for potential adaptation relevance. The preliminary work programme includes a list of 14 priority standards, 13 of which already exist but need to be revised and one of which is new and needs to be developed. Within the energy sector, three standards relating to gas infrastructure (EN 16348, EN 15399) and to installation and equipment for liquefied natural gas (EN 1473) were identified for revision. In the current second phase (2018-2022), Technical Committees are revising these standards to 'represent best practice in adaptation and resilience building' (CEN-CENELEC, 2018). A tailored guidance document is under development to support further consideration of adaptation to climate change in the future revision of existing standards, as well as development of new ones. A key challenge for the revision of standards is the availability of relevant and reliable climate information, both for the recent past and the future. The Copernicus Climate Change Service (see Section 4.2.4) is seen as a key resource in this respect.

CEN and CENELEC are also in charge of developing and revising standards for electric appliances under the Ecodesign Directive and the Energy Labelling Regulation, which mandate minimum energy efficiency standards and energy performance labels for appliances (EU, 2009, 2017). Of particular relevance in the context of this report are efficiency standards for air conditioners, the use of which is projected to grow rapidly as a result of climate change and socio-economic developments. These standards have been adapted recently by CEN (EU, 2012; EC, 2018g).

## 4.2.2 EU energy policy and related activities

EU energy policy aims to achieve three key goals (or 'pillars'): secure energy supplies (security of supply pillar), affordable prices (competitiveness pillar), and sustainable energy production and use (environment pillar) (EC, 2018n). Climate policies have played an increasingly important role in European energy policy-making since the 1992 Rio Earth Summit (EC, 2015a). The 2020 climate and energy package adopted in 2009 placed climate change mitigation at the centre of EU energy policy. With the European Union firmly committed to the goals of the 2015 Paris Agreement, climate change will remain an important policy focus. Adaptation considerations are increasingly mainstreamed in EU energy policy.

## The Energy Union

The **European Energy Union Strategy** launched in 2015 is a comprehensive European energy policy package that addresses the three pillars of EU energy policy by focussing on five dimensions (EC, 2018c):

- 1. **Security, solidarity and trust** (security of supply pillar): diversifying Europe's sources of energy and ensuring energy security through solidarity and cooperation between EU countries.
- 2. A fully integrated internal energy market (competitiveness pillar): enabling energy to flow freely through the EU, through adequate

infrastructure and without technical or regulatory barriers.

- 3. Energy efficiency and moderation of demand (environment/security of supply pillar): improved energy efficiency will reduce dependence on energy imports, lower emissions, and drive jobs and growth.
- 4. **Decarbonising the economy** (environment pillar): the EU is committed to the Paris Agreement and to retaining its leadership in the area of renewable energy.
- 5. **Research, innovation and competitiveness** (competitiveness pillar): supporting breakthroughs in low-carbon and clean energy technologies by prioritising research and innovation to drive the energy transition and improve competitiveness.

The Energy Union Strategy includes quantitative mitigation targets for 2030, building on the climate and energy package targets for 2030 (see Section 2.2.1 for details) (EC, 2016f).

The development of the Energy Union is supported by several pieces of legislation, some of which are described below.

**The Regulation on the governance of the Energy Union and climate action** (EU, 2018d), adopted in December 2018, includes several requirements for EU Member States that are relevant in the context of this report. First, Member States need to develop integrated **national energy and climate plans (NECPs)** with a 10-year perspective that are aligned with the five dimensions of the Energy Union (Article 3). Climate adaptation and resilience is addressed in two dimensions therein: horizontal 'adaptation goals' are required under the 'Decarbonisation' dimension, whereas objectives relating to 'resilience of regional and national energy systems' need to be included under the 'Energy Security' dimension.

The NECPs will be implemented in the 2021-2030 period, with an update to be provided in 2024 and subsequent plans to be developed every 10 years. Member States were requested to submit their draft plans to the Commission by 31 December 2018. The Commission will provide recommendations by 30 June 2019 (EU, 2018d). In the light of these recommendations, Member States are to submit final plans by 31 December 2019. Starting in March 2023, Member States are required to provide progress reports on the implementation of the NECPs every 2 years (Article 17). All Member States had submitted their draft NECPs by late February 2019 (EC, 2019c). The timing of this EEA report did not enable the analysis of the coverage of adaptation and climate resilience in the draft NECPS, most of which are available only in the relevant national languages.

Member States are also required to develop **long-term strategies** (LTSs) with a perspective of at least 30 years, which shall include 'adaptation policies and measures' (Article 15). The first LTSs need to be submitted by 1 January 2020, and subsequent LTSs by 1 January 2029 and every 10 years thereafter (with updates every 5 years if necessary). Finally, Member States need to **report on national adaptation actions** every 2 years, starting in March 2021 (Article 19).

An Energy Union Committee (related to the integrated national energy and climate progress reports) and a Climate Change Committee (related to the report on national adaptation actions) assist the Commission in implementing this regulation by defining the structure and format of the respective reporting templates. Further assistance will be provided by the EEA in relation to all reporting streams.

Another relevant piece of legislation is a (revised) Regulation on risk-preparedness in the electricity **sector**, which is close to adoption (EU, 2018c). The forthcoming Regulation includes four sets of measures:

- common rules on how to prevent and prepare for electricity crises to ensure cross-border cooperation;
- 2. common rules for managing crisis situations;
- common methods to assess risks relating to the security of supply;
- 4. a common framework for the better evaluation and monitoring security of the electricity supply.

In particular, this Regulation will require the European Network of Transmission System Operators for Electricity (ENTSO-E) to develop regional crisis scenarios for each region. These scenarios inform the development of national risk-preparedness plans that need to be developed by all Member States. The Regulation also includes an assistance mechanism between Member States. The agreed text repeatedly highlights 'extreme weather' or 'severe weather' as the trigger for electricity crisis situations. Interestingly, it does not explicitly refer to 'climate change', even though the impact assessment for this Regulation mentions climate change as a driver for increasing crisis situations (EC, 2016i). It is recommended that the development of crisis scenarios under this Regulation also considers unprecedented extreme weather events that may occur in a changing climate, such as extreme heat waves and droughts, possibly in combination.

## The European Union energy strategies for 2020 and 2030 and the long-term strategy

The EU has developed various strategies to meet the EU Energy Union targets for 2020, 2030 and 2050 (see Section 2.2.2 for details). **The 2020 energy strategy** established five priorities (EC, 2010a):

- Improve energy efficiency by accelerating investment into efficient buildings, products and transport.
- 2. Build a pan-European energy market via infrastructural development to integrate all EU countries to the internal market.
- 3. Protect consumer rights and achieve high safety standards in the energy sector.

- 4. Implement the Strategic Energy Technology Plan to accelerate low-carbon technology development and deployment.
- 5. Safeguard external supplies of energy.

The 2020 climate and energy package set the targets of reducing GHG emissions by 20 % compared with 1990, ensuring a share of 20 % renewable energy sources in energy consumption and achieving a 20 % reduction in total primary energy consumption by 2020 (compared with a business-as-usual baseline).

For the period after 2020, **the 2030 climate and energy framework** (EC, 2018a) puts forward new indicators for the competitiveness and security of the energy system, a new governance system based on national plans, and a new framework committed to the quantitative targets (see Section 2.2.2).

The Commission's proposal for a long-term EU GHG emission reduction strategy, 'A Clean Planet for All' (EC, 2018j), puts forward a vision to steer the EU economy and society towards a (largely) CO<sub>2</sub> emissions-free future in 2050. It also emphasises the importance of adaptation and climate resilience.

## Other European Union policies with adaptation relevance for the energy system

In 2013, the Commission adopted the first list of **projects of common interest** on trans-European energy infrastructure (EC, 2010b, 2018m). Infrastructure priorities focus on strategic planning, establishing a comprehensive infrastructure development scenario, smart grids, defining clear and transparent priority project criteria, ensuring fast and transparent permit-granting procedures and improving EU financial instruments. The Innovation and Networks Executive Agency was established in 2014 to implement the Trans-European Transport Network (TEN-T) programme, while also implementing the Connecting Europe Facility (EC, 2018d, 2018n).

## The Trans-European Networks for Energy (TEN-E)

(EC, 2018w) strategy is focused on improving the links between the energy infrastructure of the EU across Member States. The mandatory TEN-E guidelines adopted in 2013 (EU, 2013b) place a stronger emphasis on climate mitigation than adaptation. However, they prescribe that Member States should give due consideration in infrastructure planning to improving resilience and preventing technical failures induced by climate change (EC, 2018i, footnote 83).

The Commission proposal for a regulation on the internal market for electricity is currently proceeding through the legislative process as part of the implementation of the Clean Energy package (EC, 2016j; European Parliament, 2018). Its key objectives include the redesign of the electricity market to enable greater consumer empowerment to manage their own energy consumption, participate on an equal footing to other market participants and support demand-side solutions, storage and the integration of digitisation and innovative technologies. These last flexibilities could potentially help to achieve important benefits in terms of energy system resilience.

The **Directive on the protection of critical infrastructure** (EU, 2008a) has established the European Programme for Critical Infrastructure Protection (EPCIP). One of the pillars of the EPCIP is the Critical Infrastructure Warning Information Network, which offers accredited members of the EU's critical infrastructure protection community the opportunity to exchange and discuss information related to a range of hazards and to share good practices across multiple sectors, particularly in the case of transboundary networks. This Directive is currently being evaluated (EC, 2018o).

The **Offshore Safety Directive** (EU, 2013a) requires that risk assessments concerning offshore oil and gas activities must consider environmental hazards, including the impacts of climate change. Similarly, the **Directive on maritime spatial planning** (EU, 2014a) requires that the development of energy sectors at sea improves resilience to climate change impacts.

## Activities of European energy regulators

The **Agency for the Cooperation of Energy Regulators** (ACER), an EU agency, was created in 2011 to further complete the internal energy market for both electricity and natural gas. As an independent European structure that fosters cooperation among European energy regulators, ACER ensures that market integration and the harmonisation of regulatory frameworks are achieved within the framework of the EU's energy policy objectives (see Section 4.2.2). With the adoption of the Clean Energy package, ACER will have an enhanced role in the energy market and in the area of security of supply. ACER's roles will be adapted to the new challenges the electricity sector is facing, for example in the context of increased regional cooperation.

The **Council of European Union Energy Regulators** (CEER) is the voice of Europe's national energy regulatory authorities. It gives regulators a platform to share their experiences and develop regulatory capacities and best practices through workshops, events and training. Recent workshops organised by CEER have addressed, among other things, the security of supply, the integration of renewable energy into the energy system, digitisation and battery storage. Climate change has been an important topic in previous annual reports, albeit from a mitigation rather than adaptation perspective (CEER, 2012). CEER works very closely with, supports and complements the work of ACER.

## The European Nuclear Safety Regulators Group

(ENSREG) is an independent expert advisory group comprising senior officials from nuclear safety, radioactive waste safety and radiation protection regulatory authorities. ENSREG aims to provide improvements to nuclear safety and radioactive waste management through cooperation with Member States and improving transparency on nuclear waste and safety issues and by providing advice to the Commission. ENSREG also assists in the development of safety margins for nuclear power plants and the review of the response of nuclear facilities to extreme situations (including climate change-related events such as flooding) through the 'EU stress test' (ENSREG, 2012). Through the development of guidelines, measures to increase the robustness of plants have been implemented. These include the deployment of climate information services, the hardening of equipment, the improvement of severe accident management processes, and specific training measures.

## Activities of European network operators

ENTSO-E represents electricity transmission system operators throughout Europe. It publishes annual mid-term adequacy forecasts (ENTSO-E, 2017) as well as 10-year network development plans (TYNDPs) (ENTSO-E, 2018). Resource adequacy reports seek to analyse the balance between available energy generation and load levels through modelling tools. ENTSO-E modelling now explicitly incorporates current and projected climate data, enabling an examination of climate change impacts on the adequacy of the EU electricity transmission system. TYNDPs offer a series of possible energy futures through which to deliver European energy targets, while incorporating projected climate change impacts on energy systems within their modelling. More information on these scenarios is presented in Section 2.2.2. ENTSO-E will have additional responsibilities following the adoption of the (revised) Regulation on risk-preparedness in the electricity sector (EU, 2018c) (see Section 4.2.2).

The European Network of Transmission System Operators – Gas (ENTSO-G) follows very similar processes to ENTSO-E, with which it works closely. ENTSO-G analyses disruption scenarios to the European gas supply and the resilience of the system to such events. The results are taken into account in the TYNDP for natural gas.

## 4.2.3 Adaptation finance

The EU does not have a specific fund for climate change adaptation. Instead, the Commission provides funding opportunities for climate change adaptation through existing structures. Furthermore, the Commission supports the reorientation of private capital flows towards sustainable investment, including climate change adaptation.

# Mainstreaming adaptation in European Union funding programmes

Substantial mainstreaming of climate change adaptation in funding programmes has taken place in recent years. The EU has agreed that at least 20 % of its budget for 2014-2020 should be spent on climate-related action (EC, 2011a; European Council, 2013). The Commission proposal for the EU budget for 2021-2027 suggests that the share of climate-related spending be increased from 20 % to 25 % across all EU programmes, the equivalent of EUR 320 billion (EC, 2018b).

The key funding structures for climate action in the 2014-2020 EU budget include the LIFE programme (with about EUR 800 million spent on climate projects), Horizon 2020 (with 35 % of its EUR 70 billion budget devoted to climate-related projects), the EU Solidarity Fund (for natural disasters), as well as the five European structural and investment funds (ESIFs) (EC, 2016h):

- 1. the ERDF;
- 2. the European Social Fund (ESF);
- 3. the CF;
- 4. the European Agricultural Fund for Rural Development;
- 5. the European Maritime and Fisheries Fund.

The ERDF and the CF have provided funding for energy-related major projects, which are required to incorporate adaptation considerations (see Section 4.2.1). The ESF provides funding predominantly for investing in human capital to improve job prospects. The ESF can contribute to climate change adaptation through training high-, medium- and low-skilled workers, particularly for the refurbishment and retrofitting of buildings for climate resilience (EC, 2015b). The overall allocation of ESIFs towards supporting climate change adaptation is 15 % (and 25.6 % towards climate action in total). Nevertheless, certain measures, such as the protection of energy infrastructure from flooding, have not received sufficient attention (COWI, 2016, 2017).

Climate-related spending is tracked through 'EU climate markers', which are adapted from the 'Rio markers' developed by the Organisation for Economic Co-operation and Development (OECD). Analyses by the Commission indicate that the EU is broadly on track for the 20 % target, but further efforts are needed (EC, 2016d). A report by the European Court of Auditors acknowledged that ambitious work was under way and that the target has led to more, and better focused, climate action in the ERDF and the CF. At the same time, the report stressed the serious risk that the 20 % target will not be met, and that no significant shift towards climate action had yet occurred in some areas. The report also emphasised the methodological weaknesses of the current tracking method, including the failure of tracking mitigation and adaptation spending separately (ECA, 2016). Broadly similar conclusions, and various suggestions for improved climate mainstreaming in the next EU Multiannual Framework (2021-2027), have been reached in a recent study for the Directorate-General (DG) for Climate Action (Forster et al., 2017).

## Action plan on sustainable finance

In March 2018, the Commission adopted an action plan on sustainable finance (EC, 2018k). This action plan intends to reorient capital flows towards sustainable investment, in order to achieve sustainable and inclusive growth; manage financial risks stemming from climate change, environmental degradation and social issues; and foster transparency and long-termism in financial and economic activity. Some of the proposed actions are relevant in the context of this report.

The Directive on disclosure of non-financial information (EU, 2014b) requires most large companies (including energy companies) to include information on certain social and environmental policies in their annual reports from 2018 onwards. The Commission has published non-binding 'Guidelines on non-financial reporting' that countries may use to provide this information (EC, 2017c). In January 2019, the Technical Expert Group on Sustainable Finance published its Report on climate-related disclosures (EU TEGSF, 2019). This report recommended that climate-related impacts and risks on a company's business model be included in the non-financial reporting. Climate-related risks are distinguished into physical risks (i.e. risks related to the impacts of climate change) and transition risks (i.e. risks related to the transition to a decarbonised economy).

Key performance indicators shall be linked to national and international climate-related policies, whereby the EU adaptation strategy is mentioned explicitly. The main target of these annual reports are investors, but this kind of information could also be useful for monitoring progress in addressing adaptation by energy companies. These EU activities are aligned with the global Task Force on Climate-related Financial Disclosures (<sup>8</sup>). France introduced similar disclosure obligations for financial asset managers and institutional investors under the Energy Transition Law in 2015 (PRI/UNEP FI/UN Global Compact, 2016).

An analysis of companies' responses to the (voluntary) 2017 **Carbon Disclosure Project** questionnaire found that only 5 % of the responding companies used scenario analysis to evaluate their climate-related risks and opportunities but that this percentage is more than 20 % for oil and gas companies and energy utilities (Vailles and Metivier, 2019).

The Commission has also proposed a **Regulation on the establishment of a framework to facilitate sustainable investment** (EC, 2018t). This Regulation would establish the conditions and the framework to gradually create a 'taxonomy' of what can be considered an environmentally sustainable economic activity, with the aim of channelling investments into such activities. 'Climate change adaptation' is one of the five environmental objectives of the proposed regulation (Article 5), and a delegated act establishing its criteria should be developed by December 2019 (Article 7).

#### Activities of European financing institutions

The **European Investment Bank** (EIB) makes available dedicated adaptation funding and has developed investor guidelines with a focus on climate resilience. Furthermore, the EIB includes requirements for climate change adaptation in its climate strategy, particularly for projects in which climate change vulnerabilities present a high risk (EIB, 2015).

The European Bank for Reconstruction and Development (EBRD) is an intergovernmental organisation with a strong representation of EU Member States. The EBRD provides support to several European Neighbourhood countries on a wide breadth of topics, with 'power and energy' being the second largest sector. Targeted investment support for climate change adaptation is provided as part of the sustainable energy initiative, which was recently merged into the green economy transition approach (EBRD, 2018). Climate change risk assessment is a key element of the EBRD investment cycle. EBRD collaborates with many other organisations in order

<sup>(8)</sup> https://www.fsb-tcfd.org

to improve climate resilience building by private stakeholders, including by participating in the Working Group on Adaptation to Climate Change (EUFIWACC) (see Section 4.2.4) and collaborating with the International Hydropower Association (see Section 4.4.2).

## 4.2.4 Developing the knowledge base for adaptation

## Research and development projects

Many research and development projects and other programmes are improving the knowledge base for adaptation and resilience building in Europe. The energy sector is regularly included as one of the key sectors in projects related to climate change vulnerability assessment, adaptation planning and resilience building. Selected EU-funded research projects include (see Section 3.7 for some results):

- EU-CIRCLE (<sup>9</sup>) (a pan-European framework for strengthening critical infrastructure resilience to climate change);
- PESETA III (<sup>10</sup>) (Projections of economic impacts of climate change in sectors of the EU based on bottom-up analysis);
- RESIN (11) (Climate-resilient cities and infrastructure);
- ToPDAd (<sup>12</sup>) (Tool-supported policy development for regional adaptation);
- WATERFLEX (13) (Water-related modelling in electric power systems).

An overview of additional EU-funded projects that focus on strengthening the resilience of critical infrastructure in Europe (including, but not limited to, climate resilience) is available on the EU-CIRCLE website (<sup>14</sup>).

Various EU-funded projects focus on the development of innovative climate-resilient solutions for the renewable energy sector. Examples for concentrated solar power include MinWaterCSP (<sup>15</sup>) (Minimized water consumption in CSP plants) and WASCOP (<sup>16</sup>) (Water Saving for Concentrated Solar Power). Climate change adaptation needs in the energy sector have also been the subject of studies commissioned by the Commission. A study for DG Energy assessed the costs of adaptation to the electricity sector (Rademaekers et al., 2011) (see Section 3.7.1 for further information). A study for DG Climate Action assessed adaptation knowledge in Europe for three infrastructure sectors, including energy (Ecofys, 2017).

## Weather and climate services

Weather and climate services can help different actors in the energy system to improve operational and strategic decisions by considering relevant information about climate variability and change over various time scales.

The Copernicus Climate Change Service (C3S) (17) is an important new service and source of relevant climate information for various sectors. C3S aggregates satellite and in situ data to provide services to a wide range of actors in Europe, such as public institutions, private enterprises, scientific organisations and citizens. The data can be utilised to improve the decision-making and planning of climate adaptation measures. C3S provides climate change indicators and visualisation tools tailored for the energy sector through the Sectoral Information System, building on two demonstrator projects: CLIM4ENERGY (18) and European Climatic Energy Mixes (19). Other products offered include the Climate Data Platform, which provides historical climate data, trends, and seasonal and long-term forecasts. These products and services can, for example, support the planning of hydroelectric reservoir placement, assist solar generators to plan for adverse conditions, provide predictive models to assess conditions for grid upgrades, and help predict the potential yield of wind and solar generation sites. C3S is one of the few EU-wide information platforms with specific content and tools relating to the energy system or sector. The most recent Copernicus market report provides many examples of how C3S and other Copernicus services have been used for planning, developing and operating renewable energy technologies (EC, 2019a).

(10) https://ec.europa.eu/jrc/en/peseta-iii

- $\label{eq:linear} (\ensuremath{^{13}}) \quad https://ec.europa.eu/jrc/en/publication/eur-scientific-and-technical-research-reports/$
- water-related-modelling-electric-power-systems-waterflex-exploratory-research-project

<sup>(9)</sup> http://www.eu-circle.eu

<sup>(11)</sup> http://www.resin-cities.eu

<sup>(12)</sup> http://www.topdad.eu

<sup>(&</sup>lt;sup>14</sup>) http://www.eu-circle.eu/eu-funded-projects

<sup>(&</sup>lt;sup>15</sup>) https://www.minwatercsp.eu

<sup>(&</sup>lt;sup>16</sup>) http://wascop.eu

<sup>(&</sup>lt;sup>17</sup>) https://climate.copernicus.eu

<sup>(18)</sup> http://clim4energy.climate.copernicus.eu

<sup>(&</sup>lt;sup>19</sup>) http://ecem.climate.copernicus.eu

The Commission also supports the development of climate services through co-funding the European Research Area for Climate Services (ERA4CS (<sup>20</sup>)) under the Joint Programming Initiative Climate (<sup>21</sup>). ERA4CS is funding many projects related to the development of climate services (for an overview, see JPI Climate, 2017).

EU-funded projects that address the development of weather and climate services with relevance for (parts of) the energy system include:

- CLARA (<sup>22</sup>) (Climate forecast enabled knowledge services);
- CLIM2POWER (<sup>23</sup>) (Translating climate data into power plant operational guidance);
- IMPREX (<sup>24</sup>) (Improving predictions and management of hydrological extremes);
- MARCO (<sup>25</sup>) (Market research for a climate services observatory);
- PUCS (<sup>26</sup>) (Pan-European urban climate services);
- S2S4E (<sup>27</sup>) (Sub-seasonal to seasonal climate forecasting for energy);
- SECLI-FIRM (<sup>28</sup>) (Added value of seasonal climate forecasting for integrated risk assessment);
- WINDSURFER (<sup>29</sup>) (Wind and wave scenarios, uncertainty and climate risk assessments for forestry, energy and reinsurance).

Several of these projects include case studies that demonstrate the value of weather and climate services for various stakeholders from the energy sector.

The recently formed World Energy and Meteorology Council (<sup>30</sup>) aims to facilitate progress in understanding the link between climate, weather and energy, and using relevant information (Troccoli et al., 2014, 2018; Troccoli, 2018). Further information on the development of climate services for the energy sector at the national/subnational level and the global level is available in Sections 4.3.3 and 4.4.1, respectively.

- (23) https://clim2power.com
- (24) http://imprex.eu
- (25) http://marco-h2020.eu
- (<sup>26</sup>) http://climate-fit.city
- (<sup>27</sup>) https://www.s2s4e.eu
- (28) http://www.secli-firm.eu
- (<sup>29</sup>) https://sites.google.com/view/windsurfer
- (30) http://www.wemcouncil.org

#### Guidelines for risk assessment, adaptation and resilience

Early consideration of climate change requirements in project development cycles facilitates the cost-efficient integration of resilience measures. Various European institutions and organisations have produced guidelines to facilitate the inclusion of resilience measures in projects seeking EU funding.

In 2011, the Commission developed guidelines for investors, including for the development of energy projects, with a focus on 'making vulnerable investments climate resilient' (EC, 2011b). Later, the Commission published guidelines for actors seeking funding for major projects (i.e. projects with costs exceeding EUR 50 million) (EC, 2016a). These guidelines give an indication of activities that should be carried out in the development of major projects. Adaptation-relevant activities include implementing vulnerability and risk assessments, considering environment and climate change aspects, factoring in technical aspects during design, implementing adaptation measures and monitoring critical climate hazards. The guidance document explains how to conduct vulnerability and risk assessments by providing a sample method for project managers for making infrastructure climate resilient. The method places particular emphasis on selecting scenarios that incorporate realistic global temperature rises, selecting a timescale for the vulnerability and risk assessment that corresponds to the intended lifespan of the investment, and taking into account significant changes in weather events that could hamper the project.

Eight European financing institutions, including the Commission, the EIB and the EBRD, have created a **working group on adaptation to climate change** (EUFIWACC). EUFIWACC has developed practical guidance for project managers to incorporate appropriate climate-resilient measures within project development. The guidance gives a list of important points to consider during the scoping, planning and design, analysis of costs and benefits, and the evaluation and monitoring periods of projects. It is intended to be applicable to a diverse array of actors, including those in the energy system (EUFIWACC, 2016).

<sup>(20)</sup> http://www.jpi-climate.eu/ERA4CS

<sup>(&</sup>lt;sup>21</sup>) http://www.jpi-climate.eu

<sup>(22)</sup> http://www.clara-project.eu

The Commission published or updated several other EU guidance documents for planning infrastructure projects to include consideration of climate risk in the planning phase, including a cost-benefit analysis (EC, 2018i, footnote 86). The Commission has also supported the development of guidelines for adaptation in the energy system in developing countries through the EU Energy Initiative Partnership Dialogue Facility (Stuart, 2017).

Various EU-funded research and development projects have addressed climate change vulnerability and risk assessment for critical infrastructure, including in the energy system. The RAIN project (31) has assessed the complex interactions between extreme weather events and land-based infrastructure systems, including energy-related infrastructure. The INTACT project (32) has focused on assessing the resilience of European critical infrastructure to extreme weather conditions, on raising awareness among operators of critical infrastructure, and on identifying potential measures to increase resilience. Key project results are available through the INTACT Wiki (33). The STREST project (34) has developed and tested a harmonised approach to stress tests for critical non-nuclear infrastructure against natural hazards, including geophysical events and floods.

## Adaptation platforms

Adaptation platforms are web-based portals that provide information relevant for adaptation policymaking and planning to a range of stakeholders. The leading European adaptation platform is Climate-ADAPT (<sup>35</sup>), co-managed by the EEA and the Commission. Many countries have also developed national adaptation platforms (EEA, 2015, 2018i).

## 4.3 Activities by European countries

National governments play an important role in influencing the development of the energy system and in facilitating adaptation. Among other things, they identify climate-related vulnerabilities and risks, develop national adaptation strategies and plans, provide supporting information and guidance, and establish appropriate policies and frameworks to stimulate practical action. In some cases, they also fund or implement adaptation measures directly. For a more detailed discussion of the role of governments in building climate resilience in the energy system, see Section 4.1.3.

## 4.3.1 Overview

Table 4.4 gives an overview of the engagement of national governments with respect to climate change adaptation in the energy system, based on a number of published documents. It is important to note that this overview is necessarily incomplete and that a comparison across countries is hampered by differing national circumstances, such as the role of national versus subnational authorities. Despite those caveats, the tabular overview provides some indication of the level of engagement across Europe.

The data sources and assessment approach for each of the columns in Table 4.4 is described below. Further information on the reasoning behind the 'colour coding' for each country is provided in Annex 1. Updated information on countries' adaptation activities is available in the country profiles on Climate-ADAPT (<sup>36</sup>).

## 1. Availability of national CCIV assessment

CCIV assessments are key components of adaptation policy development. They facilitate the development of national and/or sectoral adaptation strategies and plans by identifying focal areas for action. The assessment here is based on a recent EEA report, which provides an overview of multi-sectoral CCIV assessments in EEA member countries based on a desk review and a country survey (EEA, 2018f) (updated in the case of Poland).

# 2. Coverage of the energy sector in national CCIV assessments

The assessment here is based primarily on the coverage of the energy sector in multi-sectoral CCIV assessments, in particular those reviewed by the EEA in its report *National climate change vulnerability and risk assessments in Europe*, 2018 (EEA, 2018f) (<sup>37</sup>). Sectoral CCIV assessments, national communications to the United Nations Framework Convention on Climate Change (UNFCCC) and other documents were also considered where available and relevant.

<sup>(31)</sup> http://rain-project.eu

<sup>(32)</sup> http://intact-project.eu

<sup>(&</sup>lt;sup>33</sup>) http://scm.ulster.ac.uk/~scmresearch/intactnew/index.php/INTACT\_Wiki

<sup>(&</sup>lt;sup>34</sup>) http://www.strest-eu.org

<sup>(&</sup>lt;sup>35</sup>) https://climate-adapt.eea.europa.eu

<sup>(36)</sup> https://climate-adapt.eea.europa.eu/countries-regions/countries

<sup>(37)</sup> All assessments are available here: https://projects.eionet.europa.eu/2018-eea-report-national-cciv-assessments/library/national-documents

# 3. National adaptation strategies and/or national adaptation plans adopted

National adaptation strategies (NASs) and national adaptation plans (NAPs) are government documents that help to integrate climate change adaptation considerations into the national decision-making processes. A NAS builds on an on different sectors, policy domains and/or regions. A NAP aims to implement practical measures in order to achieve the objectives established in a NAS. However, the distinction is not always clear-cut given different national circumstances. Information on the status of NASs and NAPs is based on country reporting to the EEA (EEA, 2018g; updated based on Eionet, 2019). For EU Member States, monitoring is obligatory under the Greenhouse gas Monitoring Mechanism Regulation (EU, 2013c) whereas monitoring is voluntary for other EEA member countries. In Table 4.4, green indicates that both a NAS and a NAP are in place; yellow indicates that only a NAS is in place; red indicates that neither a NAS nor a NAP is in place.

## 4. Coverage of the energy sector in NASs and/or NAPs

The assessment here is based on a brief analysis of the NAP and/or the NAS (see previous column). A NAS or NAP under development was also considered (where applicable). The colour scheme reflects both the type of policy document and the depth of coverage of the energy system. Specifically, green shading requires that significant parts of the energy system are addressed in an adopted NAP (not just a NAS).

# 5. Adaptation measures implemented in the energy sector

The assessment in these two columns is based on two different documents, which are explained below. In both cases, the colour coding takes into account several factors, including the specificity of a measure (e.g. generic activity versus activity targeted at the energy system), the proximity to implementation (e.g. preparatory activities such as studies and capacity building versus concrete actions such as legal and regulatory requirements or physical measures), the degree of initiative (e.g. participating in a transnational cooperation versus initiating action), and the scope of the activity (e.g. small pilot project versus comprehensive sectoral strategy). The final assessment unavoidably includes some subjective elements.

Adaptation preparedness scoreboard **country fiches** have been developed by the Commission, in

consultation with Member States, in order to assess the adaptation preparedness of all EU Member States based on a number of indicators. They have been published together with the evaluation of the EU adaptation strategy (see Section 4.2.1). The assessment here is based on a brief analysis of the country fiches (EC, 2018h). Literature cited in the country fiches was also considered where relevant.

The **UNFCCC** requires parties included in its Annex I to submit national communications (NCs) every 4 years. All EEA member countries are included in Annex I of the UNFCCC, and, therefore, they had to submit their seventh NC by 1 January 2018. Within these communications, parties must show the activities they have undertaken by reporting on GHG emissions, measures taken to mitigate GHG emissions, and measures to counteract adverse effects of climate change. The assessment here was based on a brief analysis of the seventh NCs to the UNFCCC (<sup>38</sup>).

Table 4.4 shows that all EU Member States and three other EEA member countries have conducted a multi-sectoral national CCIV assessment. Almost all CCIV assessments have included the energy sector to **some degree.** Some countries have conducted specific assessments for the energy sector. Most EEA member countries have adopted a NAS, and about half of them have also adopted a NAP. Most, but not all, NASs and NAPs include energy as a relevant sector for adaptation. Only a few countries mention adaptation actions in the energy sector in their country fiches or NCs. The majority of actions reported are preparatory actions, such as capacity building (see Annex 1 for details). Note that adaptation in the energy system may also have been implemented in countries that have not reported such activities.

The evaluation of the EU strategy on adaptation to climate change gives some additional insights into the adaptation needs of the energy sector. The stakeholders involved in the consultation process identified climate proofing of energy infrastructures as the priority area of intervention for adaptation in the energy sector. They have also pointed out that energy is one of the areas in which knowledge gaps persist (EC, 2018i).

A comprehensive review of all NASs and NAPs, many of which are available only in the national language(s), is beyond the scope of this report. However, the following subsections present some examples of how national governments have facilitated adaptation planning and implementation in the electricity system.

<sup>(&</sup>lt;sup>38</sup>) https://unfccc.int/process-and-meetings/transparency-and-reporting/reporting-and-review-under-the-convention/ national-communications-and-biennial-reports-annex-i-parties/seventh-national-communications-annex-i
Country	Document								
	Climate change vulnerability an assessments	impact, d risk (CCIV)	National adapta (NAS) and plans		Country fiches	UNFCCC			
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/or NAP adopted	Coverage of the energy sector in NAS and/or NAP	Adaptation mea implemented in sector	sures the energy			
Austria									
Belgium		_							
Bulgaria									
Croatia									
Cyprus									
Czechia									
Denmark									
Estonia									
Finland									
France									
Germany									
Greece									
Hungary									
Iceland									
Ireland									
Italy									
Latvia									
Liechtenstein									
Lithuania									
Luxembourg									
Malta									
Netherlands									
Norway									
Poland									
Portugal									
Romania									
Slovakia									
Slovenia									
Spain									
Sweden									
Switzerland									
Turkey									
United Kingdom									

### Table 4.4 Coverage of the energy system in key national adaptation documents

Note: Green, in place/identified; yellow, in progress/partially in place; red, not in place/limited or no action; grey, not applicable.

Source: EEA, based on data from EEA (2018d, 2018c); EEA/EC (2018); UNFCCC (2017); EC (2018h). Further information is available in the main text and in Annex 1.

### 4.3.2 Selected climate change risk assessments and adaptation plans

CCIV assessments are an important element of preparing countries for future climate change; they are also a key input to the development of NASs and NAPs. Such assessments can be initiated and conducted by various actors, including ministries, government agencies, research institutions and others (EEA, 2018f). CCIV assessments covering the energy sector have been prepared by almost all EEA member countries (see Section 4.3.1). This section presents the key features of a few CCIV assessments addressing the energy system, either separately or as part of a multi-sectoral assessment. These examples cannot be representative of all European countries, but they provide an insight into some of the approaches that can be taken and the outcomes of these processes.

#### Sectoral CCIV assessment for energy in the Netherlands

The Ministry of Infrastructure and Water Management has been appointed to act as the responsible ministry to oversee the development and implementation of the NAS in the Netherlands. The Ministry engages with multiple stakeholders from private, public and semi-public domains in order to implement CCIV studies and identify adaptation strategies. Through this stakeholder engagement, the Dutch Organisation for Applied Scientific Research (TNO) published a report on the risks and opportunities resulting from climate change for Dutch energy infrastructure (Vogel et al., 2014). This study compares the situation (and the progress made) in 2007 (when the first Dutch NAS was published) with the situation in 2014. TNO worked closely with other organisations, such as the Royal Netherlands Meteorological Institute, Knowledge for Climate and the Netherlands Environmental Assessment Agency (PBL). The report identified several climate vulnerabilities that should be addressed, including the negative impacts of extreme heat and droughts on the functioning of power stations due to effects on cooling water; sea level rise and flooding damage to various components of the energy system; limited public awareness of the negative impacts of climate change; and opportunities to design and export adaptation-based products and services. Further work by PBL constructed an energy vulnerability matrix. The matrix highlighted that the highest risks posed by a changing climate were in renewable energy, transmission lines, substations, cooling water and computer systems. The report also discussed cascading impacts of power outages, which are expected to become more significant in the future due to a higher reliance on digitised systems.

### Sectoral CCIV assessment for energy in Spain

The Spanish Office of Climate Change (OECC), belonging to the Ministry for the Ecological Transition, is responsible for the planning and implementation of the NAS. In 2015, OECC commissioned a comprehensive assessment of climate change impacts on the Spanish energy sector, which was developed by the Institute for Research in Technology of Comillas University of Madrid (Girardi et al., 2015). On the one hand, the assessment process included the analysis of major climate change trends using the results from regional climate change projections. On the other hand, the assessment quantified the expected impacts of climate change on the Spanish energy system in three main areas: energy sources, energy demand and energy supply. Water is specifically considered, as it is a key resource for energy production and because it strongly affects many parts of the energy system in Spain. The assessment included a section on adaptation options and recommendations, bringing together the results from a literature review at global level and an expert consultation at national level.

### Multi-sectoral CCIV assessment covering energy in the United Kingdom

The Department for Environment, Food and Rural Affairs publishes a United Kingdom-wide climate change risk assessment every 5 years under the Climate Change Act 2008. Other governmental organisations must also establish climate change vulnerability assessments and adaptation plans. This assessment yields a set of 'urgency scores' to prioritise adaptation actions and identify research priorities. Energy-relevant research priorities include the implications of projected changes in river flows on nearby pipelines; the risks to energy from high winds, lightning, storms and high waves; and the impacts of low or high river flows on electricity generation (Dawson et al., 2016).

### Development of a sectoral adaptation plan for energy in Ireland

The Department of Communications, Climate Action and Environment has completed a climate change action plan for the electricity and gas networks sector in Ireland (Government of Ireland, 2018a). The report implemented a step-by-step approach to assess current and potential vulnerabilities, identifying and assessing adaptation options, and monitoring and reviewing adaptation plans. The approach included stakeholder engagement through public consultations during the drafting of the adaptation plan, which were used to inform the finalisation of the plan. Electricity and gas network stakeholder groups were also established to assist in developing the scope and content of the adaptation plan. This approach allowed climate events that affected energy infrastructure and services to be directly identified by stakeholders, in addition to adaptation options to address such vulnerabilities. The key climate impacts for the energy sector in Ireland were identified as flooding, temperature increases, sea level rise and changes in wind energy. The report also identified the key items of infrastructure within the energy sector that are vulnerable to these projected climate impacts. Based on the assessment of vulnerabilities, the study drafted an adaptation implementation plan to set out measures to be taken. The recommended actions predominantly focus on building adaptive capacity through the dissemination of adaptation information, while also delivering physical adaptation actions through 'building on measures already in place', although the nature of these measures is unclear.

#### 4.3.3 Facilitating adaptation by energy system stakeholders

*Guidance material for CCIV assessments and resilience planning* 

Guidelines for CCIV assessments and/or resilience planning can support the development of adaptation policies and actions by guiding readers through the key steps and most relevant issues in a structured manner and by pointing to relevant tools and resources. Several European countries have developed such guidelines that address sectoral departments and agencies, subnational governments and/or non-state actors (EEA, 2018f). Further guidelines have been developed by the Commission (see Section 4.2.4) and by international organisations (see Section 4.4.1).

Most guidance documents provided by national authorities are generic and intended to be applicable across a wide range of sectors. As an exception, a guidance document from the United Kingdom has developed a framework for climate risk assessments of national infrastructure from a systems perspective (Dawson et al., 2018). The framework proposes that infrastructure beyond physical assets is taken account of, incorporating

resources that are conveyed by infrastructure (vehicles, data, water, etc.), processes that influence the infrastructure system (regulations, management, protocols, etc.), networks that join locations that demand resources/services (provided by individual physical assets to supply such resources/services) and services that benefit a range of users (heat, mobility, sanitation, etc.). Climate change can directly affect each of the components outlined above; therefore, adaptation measures may be implemented for any of these components. Prior to implementing measures on these components, a systems risk assessment of infrastructure must take place, which requires various steps. These are: (1) analyse climate variables; (2) characterise each infrastructure asset and understand its response to extreme events and changes in climate; (3) analyse the network-wide effects of impacts on components; (4) analyse interactions and interdependencies between infrastructure networks, in addition to cascading impacts; (5) assess the systemic risks that could occur as a result of infrastructure loss, including broader indirect impacts; and (6) implement adaptation measures.

Outside Europe, the United States Department of Energy has produced comprehensive guidelines for resilience planning in the electricity sector, which provide assistance to electricity utilities and other stakeholders in assessing vulnerabilities to climate change and extreme weather and in identifying an appropriate portfolio of resilience solutions. Among other things, these guidelines present a classification of adaptation measures in various parts of the energy system, and they outline a step-by-step approach for calculating the costs of adaptation measures along the energy value chain (US Department of Energy, 2016). The publication of these guidelines was complemented by networking and capacity-building efforts, such as the establishment of the Partnership for Energy Sector Climate Resilience (<sup>39</sup>) and the State and Regional Energy Risk Assessment Initiative (40).

#### Weather and climate services

Weather and climate services can provide crucial information for a wide range of stakeholders, including those from the energy sector, for managing short-term climate variability

<sup>(39)</sup> https://www.energy.gov/policy/initiatives/partnership-energy-sector-climate-resilience

<sup>(40)</sup> https://www.energy.gov/ceser/state-and-regional-energy-risk-assessment-initiative

and long-term climate change (EC, 2018f). Several European countries are supporting the development of climate services with a view to facilitating adaptation to climate change. Some of these services specifically address stakeholders in the energy sector, including energy developers, producers, distributors and traders. They are supported in a wide range of activities, including site selection, planning maintenance and construction works, planning energy storage and production capacity, and providing targeted warning services. A comprehensive review of weather and climate services for the energy sector is beyond the scope of this report.

For relevant activities at the EU level, in particular the development of the C3S, see Section 4.2.4. The Global Framework for Climate Services is briefly described in Section 4.4.1.

#### Reporting obligations for infrastructure providers

In the United Kingdom, the Climate Change Act 2008 introduced reporting obligations on climate change risks. The 'Adaptation Reporting Power' (ARP) invites or requires selected infrastructure providers, including those in the energy sector, to report information on the risks and opportunities of climate change (Government of the UK, 2012; Jude et al., 2017). By doing so, the ARP aims to assist these organisations in taking the required action to adapt to the future impacts of climate change, in addition to raising awareness and building capacity. These reports are then used by public authorities to inform the United Kingdom climate change risk assessment (see Section 4.3.2) and to update the NAP, and by industry stakeholders to further develop their activities.

Under the first round of ARP reporting, the 103 reporting organisations were clustered into nine sectors, four of which were part of the energy system: electricity distributors, electricity generators, electricity transmitters and gas transporters (Government of the UK, 2012). The reporting itself did not follow a specific format, although the affected organisations were obliged to follow the requirements set out in the ARP and the statutory guidance document to ensure that all relevant steps were followed and that there was consistency across responses.

The reporting process of the ARP was found to affect organisations in various ways. For example, organisations gave greater consideration to the impacts of climate change on their business, with the reporting process ensuring greater visibility of climate change risks at the organisational and board level. The climate change risks identified within the ARP reporting process also stimulated some organisations to develop and enhance their climate risk management activities. Several actors were found to have developed their own quantitative risk assessment and monitoring tools, somewhat incentivised by the ARP process (Jude et al., 2017). Further analysis of the value of the ARP to organisations concluded that the process used should reflect differing organisational capacities, provide for a learning environment and give effective and comprehensive feedback (Street et al., 2017).

The ARP reports relating to the energy system suggest that most electricity generating companies had made progress in completing actions, with most actions addressing on-site flooding and drought impacts on water availability (Energy UK, 2015). Transmission system operators (TSOs) and distribution system operators (DSOs) are also adapting their networks to climate change, with a focus on reducing the risks from floods and storms (see Box 4.5 for details).

# 4.4 Activities by international organisations

#### 4.4.1 International governmental organisations

International governmental organisations can support climate change adaptation by providing guidance, serving as information sharing platforms, facilitating communities of practice and acting as catalysts for the development of relevant services. Some of them may also co-fund the assessment, planning and implementation of adaptation measures. This section discusses selected international organisations with relevance for adaptation in the European energy system and their most relevant activities.

The **World Bank** was the first international organisation to address climate change adaptation specifically in the energy sector through its Energy Sector Management Assistance Program. In 2011, the World Bank published a comprehensive report on climate impacts and adaptation in the energy system, which was based on an expert workshop in 2009 (Ebinger and Vergara, 2011). This report addressed policymakers, energy planners and practitioners primarily in developing countries, but many of its findings are relevant also in industrialised countries. The background for this report was the recognition that developing countries were making considerable

#### Box 4.5 Adapting the United Kingdom's power network to climate change

Following the reporting requirements under the Climate Change Act 2008, United Kingdom energy network providers have released climate change adaptation reports, which include measures implemented to improve network resilience.

For example, the report by **Scottish and Southern Energy Power Distribution** presents the climate change risks posed to its network and establishes actions to mitigate these risks over timescales corresponding to the priority of the required action (SSEPD, 2015). Actions on the transmission and distribution systems include:

- 1. improving flood resilience through the building of flood walls;
- 2. sealing cables and raising substations;
- 3. initiating discussions on revising standards for overhead lines;
- 4. monitoring climate change impacts and updating projections;
- 5. updating the process for the management of trees; and
- 6. conducting research into tree-resilient overhead line designs.

National Grid, the United Kingdom transport system operator, has also undertaken climate change adaptation assessment and reporting. River floods, storm surges and sea level rise are expected to increase in substantial parts of the British Isles, presenting various threats to the electricity sector. These threats include material damage to substations, potentially resulting in power outages, fire, loss of temperature control and/or communication failures. To address such threats, National Grid is undertaking substation resilience improvements through a prioritised investment programme, using targeted risk assessments and including adaptation activities in its long-term business planning.

A site-based risk assessment using data from the United Kingdom's Environment Agency on river- and tidal-flood risk identified 47 substations as having the potential to be affected by a once-in-a-century flood, of which 13 were further prioritised based on site surveys and cost-benefit analyses. Such sites underwent various adaptation measures, ranging from rebuilding and elevating parts of these substations to drainage diversions. National Grid has invested GBP 17 million (≈EUR 19 million) at the highest risk sites. Other sites received GBP 136 million (≈EUR 153 million), with completion targeted by 2021. Permanent protection is the main target, but GBP 3 million (≈EUR 3.39 million) was set aside for mobile flood protection equipment. Distribution network companies have expressed appreciation of the benefits of early adaptation responses in order to mitigate future costs (National Grid, 2016).

One example of the success of the programme is provided by the Walham substation. Flooding of this substation in 2007 nearly caused the loss of power to 250 000 people. The site remained operational only due to the installation of a temporary barrier. This temporary barrier was replaced by a concrete flood wall and pumping station in 2012, which prevented any further damage to the substation when flooding of the area occurred again in 2013.

investments into energy infrastructure, with a focus on improving energy access and lowering GHG emissions, but that the energy sector was under-represented in the adaptation literature and activities. The report provides an overview of the knowledge on potential climate change impacts on energy systems, including its links with other sectors; it presents emerging practice and tools, and provides recommendations for integrating adaptation considerations into energy planning and operational practices. Some of the key messages and recommendations are that energy resources, infrastructure, services and demand are affected by climate change now and in the future; that adaptation is essential, in relation to both existing and new infrastructure; that integrated planning approaches for the energy sector should address climate change mitigation and adaptation jointly; and

that the energy-water nexus needs to be considered. These findings are still relevant today. The report also identified obstacles for adaptation and priority actions.

The International Energy Agency (IEA) has addressed climate resilience in the energy system in several of its reports and through the Nexus forum since 2012. These activities target governments, businesses and organisations involved in the energy system. The chapter entitled 'Managing climate risks to the energy sector' in a 2013 IEA report covered both sudden and gradual impacts of climate change on energy demand and supply; it stressed the need to address climate change mitigation and adaptation jointly, and it made recommendations for measures enhancing climate resilience (IEA, 2013). This work was expanded in 2015 and 2016 by two reports that focused on climate risks in relation to extreme weather and water stress (IEA, 2015, 2016b). These reports provided further insights into the synergies and trade-offs between different decarbonisation pathways and climate resilience. They also provided specific recommendations as regards the role of governments in building climate resilience into the energy system, through policies, information and services, the resilience of state-owned assets, and mobilising financial resources. The development of these reports has been supported by the Nexus forum, a series of workshops focusing on the climate-energy-security nexus. This series included workshops on the water and energy nexus and on policies and practices to enhance energy sector resilience (IEA, 2014, 2016a, 2016c). Issues relevant to climate change adaptation are also addressed in other IEA reports, such as The future of cooling (IEA, 2018a).

The **OECD** has a programme of work that aims to integrate climate change into infrastructure decisions. It has recently published a policy paper on climate-resilient infrastructure (CRI), including energy infrastructure, which builds on relevant experience from OECD and G20 countries (OECD, 2018). This publication provides a framework for planning and designing CRI, strengthening the enabling environment for CRI and mobilising investment in CRI. Among other things, it encourages governments to ensure that international, national and local approaches to policies, standards and regulations are aligned to facilitate adaptation within the private sector.

The OECD, the World Bank and UN Environment have jointly developed the report Financing climate futures (OECD/World Bank/UN Environment, 2018). This report identifies six transformational areas that are required to ensure that infrastructure is aligned to the aims of the Paris Agreement.

The International Renewable Energy Agency (IRENA) supports countries in their transition to a sustainable energy future. In 2015, IRENA published a report on the energy-water-food nexus (see Section 2.1.5) (IRENA, 2015). This report explores integrated solutions for jointly meeting policy goals for these strongly linked sectors under climate change, with a particular focus on the role of renewable energy technologies. It also sets out criteria for analysing the energy-water-food nexus, which help to guide policymakers away from the 'silo-approach' to integrated resource planning.

The **World Meteorological Organization** has initiated the **Global Framework for Climate Services** (GFCS). The GFCS 'Energy Exemplar' aims to develop user-tailored weather-water-climate services in close cooperation with the energy industry. The first Energy Exemplar report provides, among other things, a detailed analysis of the potential of climate services for five different focus areas: (1) Identification and resource assessment; (2) Impact assessment; (3) Site selection and financing; (4) Operation and maintenance; and (5) Energy integration. It also describes in detail seven specific examples of climate products and services for the energy sector (WMO/GFCS, 2017).

#### 4.4.2 International sector associations

Several international sector associations have developed relevant guidance material on climate change adaptation and strengthening resilience in the energy system.

The International Hydropower Association is preparing a climate resilience guide for the hydropower sector with the joint support of the World Bank and the EBRD (Climate Resilience Secretariat, 2018). This guide aims to assist hydropower companies in considering climate change risks in project design and operation, while also addressing the needs of the wider financial, policy and local communities. The preparation of a beta version of the guide is built on a stakeholder engagement phase. The release of the final version is expected in 2019.

The World Energy Council (WEC) published a report on managing the financing of energy sector infrastructure resilient to extreme weather risks as part of their 'Road to Resilience' series (World Energy Council, 2015). The report highlights the need for a strong regulatory framework that provides clear guidance on and definitions of what resilience is in the context of the energy system, what degree of resilience is sufficient, and how this may be financed. The WEC has also published reports on managing risks as part of the energy-water-food nexus (World Energy Council, 2015, 2016). The WEC encourages stakeholder cooperation to overcome current climate change information deficits among project developers and investors in order to ensure the correct pricing of investment risk under changing climates. By developing the necessary tools to compare the costs with the benefits of investing in resilience, private sector investment in the energy sector can be secured to ensure that energy assets receive adequate and stable returns throughout their lifetime.

The **World Business Council for Sustainable Development** (WBCSD) is an organisation of 200 forward-thinking companies from across the globe. It published a comprehensive report that analyses climate impacts and adaptation strategies for the power sector (WBCSD, 2014). The report distinguishes clearly between long-term climate impacts, long-term adaptation and long-term planning on the one hand and extreme weather events, climate resilience and crisis planning on the other. It presents multiple adaptation options to various climate change impacts along the energy system value chain. The report closes with specific recommendations for industry stakeholders, policymakers and public-private collaborations.

### 4.5 Adaptation case studies from energy utilities and network providers

#### 4.5.1 Overview

Public and private energy producers and other infrastructure providers are already adapting to climate change. A report for Natural Resources Canada based on an analysis of 200 projects globally found that climate change adaptation activities are becoming more mainstream in the energy sector (Braun and Fournier, 2016). The subsector that is most often represented in adaptation activities is energy generation, in particular related to hydropower. Ouranos has developed an online database on climate change adaptation in the energy sector with dozens of case studies (Ouranos and Natural Resources Canada, 2018). The geographical focus is North America, but several European case studies are also included. Other relevant case studies (termed 'business cases') for adaptation in the energy system, including several case studies from Europe, are provided in a WBCSD report (WBCSD, 2014).

This section presents a summary of five case studies on adaptation actions implemented by energy producers and network operators in Europe (see Figure 4.3). These case studies were selected from several sources based on the following criteria: relevance of long-term climate change, implementation of measures, importance in a decarbonised energy system, relevance of public policies, availability of information on success and limiting factors and on (expected) costs and benefits, and transferability of results. The geographical distribution of the case studies and their balance across energy system components was also considered. Each of the five case studies is presented in considerably more detail on Climate-ADAPT (41) (see the specific references below), which contains further case studies and is continuously expanded.

The five case studies are rather different, as they have intentionally been selected to showcase the diversity of potential adaptation actions by infrastructure operators in the energy sector. However, all of them relate to renewable sources of electricity production or to electricity transmission and distribution. This choice reflects the relevance of these technologies and infrastructure components in a decarbonised energy system as well as the availability of information on implemented adaptation actions.

The first two case studies are related to hydropower. The largest power producer in France has undertaken a systematic climate change risk assessment for its infrastructure and activities (see Box 4.6). This company has developed a cost-effective technology to reduce flood risks following heavy precipitation events. The



Figure 4.3 Overview of five adaptation case studies

 $({}^{41}) \quad https://climate-adapt.eea.europa.eu/knowledge/tools/case-studies-climate-adapt$ 

technology has not been licensed, and it has already been applied in numerous locations in different countries. The other case study is from the national power producer in Iceland, which has adapted its hydropower infrastructure and practices in order to exploit increased water availability due to (temporary) increases in glacial melt water. Sophisticated climate projections and economic analysis are central to this case study, as is international cooperation. Both hydropower case studies were implemented without specific policy triggers. The adaptation measures were regarded as cost-effective from the viewpoints of the companies.

The third case study is related to power production from biomass. A large power company in the United Kingdom has diversified its supply chains for overseas biomass supply in order to increase import capacity and to build redundancy in the case of extreme weather events. These measures required close collaboration with the owner and operator of the sea ports where biomass terminals were upgraded or newly built. This measure was facilitated by national legislation relating to both climate change mitigation (which requires power companies to increase the share of renewable energy sources) and adaptation (which requires to report on climate-related risks and adaptation measures).

The last two case studies relate to increasing the climate resilience of power lines (see also EC, 2018i, Annex XIV, Case Study 4). A large electric utility company in the United Kingdom is adapting design standards to modify its distribution network to increasing temperatures. A key cost-effectiveness measure is to increase the design height for wood poles by 0.5 m in order to stay within legal safety limits for overhead lines despite increased sagging during heat waves. The second largest DSO in Finland is substantially investing in replacing most of its overhead power lines with underground cables, which are protected against storms, snow and ice accumulation. The substantial costs of doing so are reduced by partnering with telecommunication companies, which can use the same excavations for their own communication cables. These measures were adopted in response to a recently introduced legal requirement to limit electricity outages related to extreme weather, in particular storms and heavy snow.

An additional adaptation example is presented in Box 4.6.

#### 4.5.2 Flood risk management for hydropower plants

*Electricité de France* (EDF), the largest electricity producer in France, has developed the piano key weir (PKW) system to improve flood discharge from hydropower dams during heavy precipitation events. The PKW system provides an additional spillway to manage increased water flow without changing the maximum reservoir level. This is particularly relevant in the narrow gorges present in some of the Alpine regions where PKWs have been installed. The primary benefit of this technology is to protect hydropower assets from damage while also reducing operational costs compared with alternative gate systems. PKWs can manage much higher flow levels and provide a safer solution than gated systems, with minimal risk of malfunction and easier evacuation of floating debris. PKWs also avoid human error, because they do not require human operators. However, each hydropower dam presents different risk levels, and the suitability and effectiveness of PKW systems must be evaluated on a case-by-case basis.

Past trends show an increase in extreme daily rainfall in southern France, which can lead to flash floods. This trend is expected to continue in the future. Snow and glacier melt are also expected

#### Box 4.6 Adapting the cooling of nuclear power plants in France to a changing climate

Increasing temperatures and localised water scarcity place pressures on thermal energy production by limiting cooling water sources. Following a series of exceptional heat waves in France, the French utility company *Electricité de France* (EDF) initiated the '*Grands Chauds*' plan in 2008. The aim of this plan is to ensure the safety and supply security of EDF's nuclear power plants in the face of increasing temperatures. A case study on the adaptation of EDF's hydropower plants to climate change is presented in Section 4.5.2.

The *Grand Chauds* plan includes a re-assessment of maximum water temperatures expected at each nuclear power plant up to 2030. Revised temperature estimates have informed the revision of engineering guidelines for extreme weather situations and of safety standards taking into account climate change impacts. EDF has also improved the cooling efficiency of its nuclear power plants through a range of measures, including closed-loop cooling systems. These systems require less cooling water than conventional cooling technologies, albeit at higher costs. Furthermore, EDF has established a risk management unit that supports operational teams in determining the need for derogations from regulated limits for thermal discharges during exceptional heat waves (SFEN, 2015). to have an impact on reservoir inflow and outflow in the longer term. Glacier mass loss already shows a consistent acceleration across the Alps. Releasing water safely from dams during floods or heavy precipitation presents a critical operational challenge at hydropower facilities. The goals of flood management are to prevent or minimise the impacts of dam overtopping on downstream communities, property, agriculture and ecosystems, and to protect the dams themselves against operational failure and other damage.

The Malarce dam (2.3 million m<sup>3</sup>, 16 MW) is located in the department of Ardèche in the Auvergne-Rhône-Alpes region of France. It features a PKW that increases the dam's maximum discharge capacity to a total of 4 600 m<sup>3</sup>/s. When dam water levels exceed the level in the inlet tanks, water automatically flows over the PKW into outlet tanks that run straight into the spillway channel and downstream. The PKW in the Malarce dam helps to reduce the risk of costly damage to dam infrastructure and to downstream communities.

The PKW has already been implemented in various hydropower facilities globally (e.g. in Algeria, South Africa and the United States), thanks to the collaborative approach of the original developers who did not patent this technology.

Further information on this case study is available on Climate-ADAPT (EEA, 2018c).

### 4.5.3 Hydropower expansion in response to increased glacier melt

Landsvirkjun, the national power company of Iceland, has included adaptations to climate change in managing, designing, updating and expanding its hydropower assets. Almost all Icelandic glaciers have been losing mass since the early 1990s, and this trend is expected to continue with the warming climate. It has been projected that almost no Icelandic glaciers will be left by 2200. As a result of this increased melting, increased river flows and changes in their seasonal distribution have been observed over past decades. Further increases in flows are expected in the next 50 years, after which the runoff from glacial melt is expected to decline. Hydropower stands to gain from increased water flow caused by climate change-induced glacier melt in the coming decades, but reservoir management will have to be adjusted to account for this increased flow.

*Landsvirkjun* has improved projections of water flows under climate change. Overall inflow volume

is projected to increase by an additional 15 % by 2050, compared with 2015. The existing power system can utilise only 30 % of that increase. Without increasing installed turbine capacity and reservoir storage at existing hydropower plants, the rest of the increased flow would be spilled over the spillways. The improved projections facilitate adaptation measures that minimise unnecessary water spills through the spillways. These measures include the modification of reservoir-management plans, the installation of additional infrastructure, and/or the re-design of existing infrastructure to manage increased runoff. A co-benefit is increased flood protection, as the reservoirs can function as extra buffer capacity in the case of extreme flooding. Landsvirkjun uses hydrological modelling to project future water flow, taking climate change impacts into consideration. In essence, the management and design of existing and planned assets is adjusted to take advantage of increased glacier flows, based on improved data on current and future flows.

At the Búrfell plant, improved water flow data were used to plan a capacity increase from 70 MW to 100 MW. This increase was implemented by building a new hydropower plant, which extends the original power capacity and reduces its load. Furthermore, the capacity of the Búðarháls hydropower plant, a new project commissioned in 2014, was increased from the originally planned 80 MW to 95 MW in response to climate change. Key factors in achieving these results have been the involvement of the Executive Board of *Landsvirkjun* in the adaptation process and collaboration with other power companies, universities and institutions.

The current expansion is economically sound given the project's time horizon of 50 years. Moreover, increased reservoir capacity installations can ensure increased general flood protection. The most extreme flood events in Iceland are glacial outburst floods caused by volcanic eruptions. By 2080, the volume of the glaciers is projected to have decreased so much that water flows will decline. At that point in time, existing hydropower plants may have higher capacity than needed.

Further information on this case study is available on Climate-ADAPT (EEA, 2018d).

### 4.5.4 Improved resilience of the biomass fuel supply chain

Drax Power Limited, a power company in the United Kingdom, has improved the resilience of its supply chain for biomass power plants. Three boilers at their large thermoelectric power plant in Immingham have recently been converted to be powered by biomass pellets. For a reliable stream of biomass, Drax relies on the maritime transport of wood pellets from North America. The United Kingdom has experienced several storm surges and coastal flooding in recent years; such events are expected to become more frequent and stronger under future climate change. The impacts of such events on energy infrastructure are significant. They include damage to supply lines, such as port infrastructure, and impacts to coastal or river-based power stations themselves. For example, Immingham port lost power in significant areas during a storm surge event in 2013 owing to localised flooding in substations.

Drax has implemented a multi-port strategy in order to alleviate the threat of supply chain backlog if its primary harbour is temporarily inoperable due to a weather-related event or other causes by means of additional biomass terminals at the ports of Tyne, Hull and Liverpool. Moreover, terminal designs at these harbours have been structurally adapted to avoid localised flooding events. Adaptation measures implemented at the Immingham site include raising power plant equipment above potential storm surge water levels, constructing protective sea walls, fitting subterranean tunnels with storm surge barriers, and planning above-ground construction of future tunnels. Similar measures have been implemented at the other locations. The geographic spread of ports and flood management measures in ports has allowed Drax to significantly increase its import capacity for wood pellet biomass. The adaptation benefits of this project are clear, and they seem transferable to other regions and projects.

The overall emissions balance of renewable fuel crops are often unclear, especially if existing forest or other carbon-rich habitat is cleared for energy crop production (Gasparatos et al., 2017; EASAC, 2019). Drax claims to source all wood pellets from sustainable, working forests to minimise net carbon releases to the atmosphere. An analysis of the climate benefits and potential biodiversity impacts of the fuel change from fossil fuels to biomass in this case study is beyond the scope of this report. For a discussion of the synergies and trade-offs between climate change adaptation, mitigation and other environmental and social concerns, see Section 4.1.4.

Further information on this case study is available on Climate-ADAPT (EEA, 2018e).

### 4.5.5 Adapting overhead lines to increasing temperatures

Western Power Distribution (WPD), a large electric utility in the United Kingdom, has cooperated with the Energy Network Association (ENA) in order to operationalise adaptation measures in its transmission and distribution infrastructure. Electric transmission and distribution network providers in the United Kingdom and elsewhere face increasing risks from heat waves, which pose a threat to their power infrastructure. Thermal expansion can cause power lines to sag, hence their clearance from land can become a danger to the general public. Sagging may also result in contact with trees and other structures, and hence in electrocution or fires. Furthermore, the electrical current that passes through overhead power lines must be reduced under high ambient temperatures to prevent the overheating of equipment. Finally, warmer power lines can result in decreased efficiency due to a process known as de-rating. All of these impacts ultimately result in risks of accidents, power cuts and revenue losses, as well as cascading network failures.

Adaptation measures by WPD include increasing the height of poles supporting power lines, installing conductors with hotter operating limits and implementing the use of 'low-sag' conductors. The most cost-effective measure identified was to increase the minimum design temperature of new overhead lines, which requires increasing the design height of wooden poles by 0.5 m. This adaptation accommodates the projected increases in sagging without exceeding legal limits on the height of overhead lines. WPD also participated in a project that updated the United Kingdom's distribution industry's understanding of conductor ratings for overhead lines, which is expected to lead to an update of national standards. Finally, WPD and various other British energy companies are developing a software tool that will allow energy companies to optimise their overhead line ratings.

Cooperation among relevant institutions turns out to be the major enabling factor for this initiative. Members of electricity network operators have been brought together with government and regulating agencies in the ENA-led Climate Change Adaptation Task Group. The task group develops approaches to identify impacts of climate change on electricity TSO and DSO networks, and to develop and select cost-effective and efficient adaptation measures. Further information on this case study is available on Climate-ADAPT (EEA, 2018a).

#### 4.5.6 Replacing overhead lines with underground cables

Elenia, the second largest electricity distribution system operator in Finland, is investing substantially in underground cabling. Expected climate change impacts in Finland include increased frequency and intensity of precipitation events, storms and snow loads. Furthermore, increasing temperatures are projected to lead to increased growth and age of trees and increased leaf coverage, which can result in hazards for aerial power lines. The collapse of power lines causes temporary loss of power to users and results in repair costs for power providers. In order to adapt to increasingly frequent and extreme weather events and to adhere to the outage requirements of Finland's Electricity Market Act, Elenia plans to increase the share of underground cables in its network from 41 % currently to 75 % by 2028. The planned investments for 2018 were EUR 140 million.

Underground cabling protects electricity transmission and distribution systems from the most frequent adverse weather conditions, in particular storms and excessive snow and ice loads. It can alleviate the requirement for further and more frequent investments in infrastructure maintenance and repairs. In addition to a more secure energy supply, additional benefits of underground cabling include lower visual impact and reduced accident risk (e.g. fires and impact associated with broken or fallen cables). The key success factor for underground cabling is the availability of the correct technology for underground cabling with regard to installation, monitoring and management. There are three main approaches: placing cabling in concrete-reinforced troughs, placing the cables in underground tunnels, or directly burying the cables. The technique used by Elenia involves the construction of trenches ranging from around 0.45 m to 1 m in depth, with the cabling then buried within the soil layer. Elenia cooperates with other underground cabling entities such as telecommunications companies to reduce costs through collaboration and to minimise disturbance to populations.

Excavation due to other construction or maintenance activity represents a key risk of damage to underground cables. Elenia applies digitisation and geographical information systems to inform excavators about the location of underground cables. Underground cabling could also be exposed to new climate hazards, in particular flooding, soil compaction and soil movements related to landslides, but these risks were considered to be low in Finland. When the application of underground cabling is considered in other locations, its costs and benefits would need to be compared with those of other adaptation options, such as improved vegetation management or reinforced overhead pylons and cables, on a case-by-case basis.

Further information on this case study is available on Climate-ADAPT (EEA, 2018h).

# 5 Conclusions and outlook

#### **Key messages**

- The European energy system increasingly needs to adapt and become more climate resilience in the context of continuing climate change, modern societies' increased dependence on a reliable energy supply and an increasing share of climate-sensitive renewable energy sources.
- Adaptation in the energy system should consider synergies and trade-offs with other sustainability concerns, in particular climate change mitigation, and land and water use.
- Many asset owners and managers, as well as policy-makers at the EU, national and regional levels, and other stakeholders are already addressing adaptation needs in the energy system, but there is scope for further action.
- The development of the Energy Union and the EU long-term strategy on climate action provide important opportunities for mainstreaming climate change adaptation in the planning and implementation of a decarbonised energy system in Europe through more coordinated actions, reporting and mutual learning among all involved actors.
- All countries should consider the impacts of climate change on the current and future energy system in the development of their national climate and energy plans and long-term strategies under the Energy Union, and in the development and update of their national adaptation strategies and action plans.
- All market actors in the energy sector, including business associations, should consider strengthening climate resilience as an integral part of their business.
- Governments can facilitate market actors in the energy sector in overcoming adaptation barriers through the regulation of energy markets and other policy interventions, as well as through 'soft' measures that focus on information provision and exchange. Such activities can be supported by reporting requirements on climate change risks and adaptation actions for critical infrastructure providers, in particular, where such information is not available from other sources.

### The European energy system faces increasing climate adaptation needs

Adaptation to climate change and building climate resilience are becoming increasingly important for the European energy system. First, global climate change implies that past climate conditions are no longer a reliable proxy for the future. Second, modern societies are increasingly dependent on a stable energy supply. Even short disruptions in electricity supply can have large economic and social costs, which can be up to one hundred times larger than the direct costs to the energy sector. Finally, the clean energy transition involves a rapidly growing share of renewable energy sources, most of which are sensitive to weather and climate variations at different time scales.

Climate change and extreme weather events are increasingly impacting all components of the energy system. Such events affect the availability of primary energy sources (in particular renewable energy sources), the transformation, transmission, distribution and storage of energy, and energy demand.

Climate change impacts and related adaptation needs vary significantly across energy system components and European regions. Some impacts can be economically beneficial, such as reduced energy demand for heating. However, many impacts are adverse for the energy sector and/or society as a whole, such as reduced cooling water availability for thermal power plants in many regions, increasing risks for energy infrastructure from extreme weather events and sea level rise. At a regional level, southern Europe faces the strongest adverse impacts due to increasing heat waves and decreasing water supply.

Building climate resilience comprises addressing the impacts of weather hazards on existing energy infrastructure and its operation, as well as considering the impacts of long-term climate change on newly planned infrastructure. It is generally more efficient to address adaptation needs early in the investment cycle rather than through the costly retrofitting of existing infrastructure.

The further development of the European energy system needs to address climate change adaptation and mitigation jointly in the context of wider sustainability concerns. There can be synergies as well as trade-offs between climate change mitigation and adaptation, depending on the particular technological or policy option and the regional context. On the one hand, replacing coal-fired power plants by photovoltaics and wind power radically reduces greenhouse gas emissions and water consumption, thus contributing to mitigation as well as adaptation in water-scarce regions. The same is generally true for increasing energy efficiency. On the other hand, biofuels or carbon capture and storage have a larger need for water withdrawals, water consumption and/or arable land than many conventional energy technologies. Similarly, hydropower generation can have adverse impacts on local and regional biodiversity.

The clean energy transition in Europe presents both challenges and opportunities for climate change adaptation. On the one hand, adaptation could be seen as an additional burden that needs to be managed on top of other challenges related to rapid decarbonisation and changes in the market and regulatory environment. On the other hand, the massive investments made in the energy system allow adaptation needs to be addressed early in the investment cycle. A careful assessment of the relevant risks and adaptation options, as well as coordinated action by a wide range of public and private stakeholders, is necessary to ensure the clean energy transition is also climate-resilient.

### Policy-makers and stakeholders are already addressing adaptation needs in the energy system

Key EU climate and energy policies and strategies promote the mainstreaming of climate change adaptation into energy policies. These include the EU adaptation strategy, the Regulation on the Energy Union and Climate Governance, the Commission proposal for a long-term strategy 'A Clean Planet for All' and the Regulation on risk-preparedness in the electricity sector. The Energy Union implies a stronger role for the EU to monitor and coordinate energy planning, reduction of greenhouse gas emissions and adaptation actions in its Member States. On a mandate from the European Commission, standardisation bodies are adapting standards for climate-sensitive infrastructure. The Regulation on the guidelines for trans-European energy infrastructure also contains provisions on the management of climate risks. The EU also supports building climate resilience in the energy system by funding relevant research and innovation projects, and developing climate services for the energy sector as part of the Copernicus services.

Most European countries have addressed the energy sector in national climate change impact, vulnerability and risk assessments, national adaptation strategies and/or action plans. Particular attention is given to the impacts of extreme events on energy infrastructure and services. However, some countries do not currently regard energy as a priority sector for adaptation. An assessment whether countries are considering adaptation needs in their long-term energy planning was not possible in the context of this report. However, such an assessment could be done in 2020 based on Member States' reporting under the Regulation on the Governance of the Energy Union and climate action.

### Many other stakeholders have also started to address adaptation needs in the energy system. Energy regulators have performed stress tests and increased regional cooperation. Sector associations are developing guidelines for climate vulnerability assessment, sometimes with the support of international organisations or national governments. Electric utilities are performing risk assessments and use the results as the basis for retrofitting more water-efficient cooling technologies, updating management procedures and improving contingency plans. Network providers are adapting design standards for overhead lines and investing substantial amounts for underground cabling.

### There is scope for further adaptation action by all involved actors

Considerable action has happened already, but more can and should be done to ensure that the energy system in Europe is climate resilient now and in the future. The challenges, opportunities and most suitable actions differ across countries and regions, depending on the physical and management structure of the energy system, climatic and environmental factors, and socio-economic conditions.

Some stakeholders in the energy system are only beginning to acknowledge the relevance of climate change impacts on their activities, or they are experiencing barriers to taking action. They may feel that the uncertainties about future climate change and its impacts are too large to justify significant investments at the present time with uncertain benefits in the future. They may be interested in action, but do not know where to find accessible and reliable information. They may not see the business case for adaptation investments given the asymmetric distribution of benefits and costs. They may prefer to postpone investments in adaptation technologies in the hope of reduced uncertainties, technological advances and falling prices. Following careful analysis, some energy system stakeholders may indeed come to the conclusion that they do not face urgent adaptation needs at the moment. However, all actors will benefit from assessing climate change-related risks and opportunities, given the climate sensitivity of many energy system components and the importance of energy for modern societies.

### Governments are crucial for overcoming adaptation barriers by market actors

Building climate resilience is a shared responsibility between infrastructure and service providers in the energy sector, regulators, national and regional governments, and the EU. Many adaptation actions in the energy system make a business case for market actors, because they increase revenues or reduce costs and/or reputational risks. However, other actions may require government or regulatory interventions, because their costs and benefits are unevenly distributed, or because their success depends on co-ordinated action by many actors, possibly across national borders.

Most barriers to efficient and effective climate change adaptation in the energy system can be addressed by government policies. Some policy mechanisms require cooperation at the transnational or European level due to the increasingly interconnected nature of the European energy system.

### *Opportunities for further action by European policy-makers*

The Energy Union provides important opportunities for mainstreaming climate change adaptation in the planning and implementation of a decarbonised energy system at the EU and Member State level.

The **Regulation on the Governance of the Energy Union and climate action** ('Governance Regulation') requires EU Member States to prepare and publish integrated national energy and climate plans (NECPs) (for the period up to 2030) and long-term strategies (e.g. up to 2050). These integrated planning processes call for addressing climate change mitigation and adaptation in the energy system jointly, and they facilitate learning from good practices across Europe. The Commission is currently (up to June 2019) assessing the draft NECPs, which address climate adaptation and resilience under the 'Decarbonisation' and the 'Energy Security' dimension. When issuing recommendations to Member States, the Commission may consider to what extent the draft NECPs address the impacts of climate change under the dimension 'Energy Security'.

The Commission is also developing implementing acts that specify in more detail the reporting requirements under the Governance Regulation. The implementing act under Article 17 — currently under development with assistance of the Energy Union Committee — could foresee that countries report on their consideration of climate change impacts in the implementation of the NECPs under the 'Energy Security' dimension. The implementing act under Article 19 —currently under development with assistance of the Climate Change Committee could foresee that countries report on their adaptation-related activities in selected sectors. For example, in its 'horizontal assessment' of adaptation preparedness country fiches, the evaluation of the EU Adaptation Strategy included a selection of sectors with transboundary relevance or where EU policies play a strong role, including energy.

The Commission and Member States are encouraged to consider the impacts of climate change when preparing and elaborating long-term strategies. The planned EU long-term strategy for climate action (due by early 2020) is currently being debated, based on the Commission proposal 'A Clean Planet for All' and views from EU Member States. The Governance Regulation also requires Member States to prepare national long-term strategies, which shall include 'adaptation policies and measures'. According to Article 15, 'the Commission shall support Member States in the preparation of their long-term strategies by providing information on the state of the underlying scientific knowledge and opportunities for sharing knowledge and best practices, including, where relevant, guidance for Member States during the development and implementation phase of their strategies'. The Commission is encouraged to support Member States in considering the impacts of climate change in the preparation of their national long-term strategies.

The Regulation on risk-preparedness in the electricity sector, the proposed Regulation establishing the Connecting Europe Facility and market regulation in the energy sector provide further opportunities for increasing climate resilience. Such regulation should ensure that market participants in the energy sector have a business case for building climate resilience at a level that reflects the importance of a secure and stable energy supply for modern societies.

### Opportunities for further action by national policy-makers

National (and sub-national) governments have a wide range of options for facilitating climate change adaptation in the energy system. These options include funding targeted research and development, providing climate services, building communities of practice, public-private partnerships, market regulation, reporting requirements and others.

National climate change impact, vulnerability and risk (CCIV) assessments of the energy system with a strong forward-looking component are essential inputs for making the clean energy transition climate-resilient. Such assessments should consider, as a minimum, the impacts of changing water availability on the availability of renewable energy sources and on electricity generation, the impacts of increasing temperatures on electricity generation, transmission and demand, and the impacts of extreme climate-related events (including coastal and marine hazards where relevant) on various types of energy infrastructure. The rapid changes in the energy system in the context of the clean energy transition imply that the future energy system in Europe will be very different from the current one. The long-lived nature of most energy infrastructure requires assessing the impacts of climate change — not only on the current energy system, but also on different scenarios of a future low-carbon energy system, where renewable energy sources, electricity and potentially hydrogen as energy carriers, and energy storage play a stronger role. The assessment should also address different plausible climate change scenarios, including scenarios that overshoot the temperature stabilisation target of the Paris Agreement.

Countries that have not addressed energy as a priority sector or policy area in their national CCIV assessment are encouraged to do so in the future. There are also opportunities for countries that already have addressed the energy system in multi-sectoral or sectoral CCIV assessments, in particular, to strengthen the resilience assessment of alternative decarbonisation scenarios.

All countries are encouraged to consider the impacts of climate change on the energy system in their national energy and adaptation planning. Specifically, EU Member States would benefit from addressing them in the preparation of their NECPs and long-term decarbonisation strategies under the Energy Union (see above for details).

Reporting requirements on climate change risks and adaptation actions for critical infrastructure providers can facilitate the building of climate resilience in the energy system, in particular, where such information is not available from other sources. For example, the United Kingdom has introduced systematic reporting requirements regarding climate change risks and adaptation actions for public and private providers of critical infrastructure, which include energy infrastructure. The experiences are generally positive, and independent reviews showed significant investments in improving the climate resilience of the UK energy infrastructure.

Countries can also facilitate the building of climate resilience in the energy system through 'soft' measures. Such measures include providing relevant information (e.g. through climate services), facilitating knowledge transfer (e.g. through adaptation platforms) and establishing communities of practice. Dialogue fora with energy system stakeholders can help identify barriers for adaptation action and options for overcoming them, e.g. through changes in legislation or regulation.

### Opportunities for further action by industry and other market actors

All market actors in the energy sector should consider building climate resilience as an integral part of their business. This consideration is particularly important because of the large investments in long-lived infrastructure that are being made in the context of the clean energy transition. However, also existing energy infrastructure may require climate proofing. Climate risk assessments are an important first step in addressing the impacts of climate change and variability on current and planned energy infrastructure and its operation.

Business associations in the energy sector can support their members in building climate resilience. They can do this by raising awareness of climate change impacts and adaptation needs, developing guidelines and tools, facilitating mutual learning, and encouraging and finding cooperative solutions, among others.

# Annex 1 Information sources for country overview table

This annex provides further information on the information sources for Table 4.4 Coverage of the energy system in key national adaptation documents in Section 4.3.1. Specific information sources and the colour codes are explained in the main text.

Country				Document		
		ge impact, vulnerability /) assessments		adaptation strategies I plans (NAP)	Country fiches	UNFCCC
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/or NAP adopted	Coverage of the energy sector in NAS and/ or NAP	Adaptation measures in energy sector	nplemented in the
Austria		The Austrian Assessment Report 2014 (AAR 14) contains a whole section on energy and transport (APCC, 2014).		The NAP includes an action field 'Energy' with a focus on 'Power supply' (Government of Austria, 2017).	Climate adaptation is reported as being integrated into sectoral policies for energy (among others), but no specific information is provided.	No adaptation actions reported.
Belgium		The national CCIV assessment includes a section on energy (Technum, 2013).		The NAP includes energy as one of the priority areas (National Climate Commission, 2017a).	The Benelux countries have organised workshops on adaptation in several sectors, including energy. The drafting of the New Industrial Policy to be introduced in the Flemish Region will incorporate the risks of volatile energy costs attributable to climate change (Government of Flanders, 2016).	Studies on 'Needs in adequation and flexibility of electric system (2016)' and the promotion of sustainable energy generation methods independent of the availability of water resources (National Climate Commission, 2017b).
Bulgaria		The national CCIV assessment focuses on natural disasters. It does not include a specific section on energy, but the energy sector is covered to a limited degree (Government of Bulgaria, 2013)		The draft NAS/NAP document includes a section on energy (Dale et al., 2018).	No adaptation actions reported.	No adaptation actions reported.
Croatia		The national CCIV assessment includes various sections on energy sector vulnerabilities to climate change, with high-risk areas throughout the sector identified (Government of Croatia, 2017b).		The draft NAS contains a section on energy (Government of Croatia, 2017a).	No adaptation actions reported.	No adaptation actions reported.
Cyprus		The national CCIV assessment includes a focus on energy (Government of Cyprus, 2016).		The NAP includes a section on energy (Government of Cyprus, 2017).	No adaptation actions reported.	No adaptation actions reported.

Country				Document		
		ge impact, vulnerability /) assessments		adaptation strategies I plans (NAP)	Country fiches	UNFCCC
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/or NAP adopted	Coverage of the energy sector in NAS and/ or NAP	Adaptation measures in energy sector	nplemented in the
Czechia		The national CCIV assessment includes a section on climate change risks to Czech industry and energy (Government of Czech Republic, 2015).		The NAP includes 'industry and energy' as one of its priority sectors (Government of Czech Republic, 2017).	No adaptation actions reported.	No adaptation actions reported.
Denmark		The national CCIV assessment includes a section on energy, which addresses energy consumption, energy production, wind changes, import/export changes and biomass production (Government of Denmark, 2012b).		The NAS includes energy as one of the sectors where climate change will have a significant impact (Government of Denmark, 2008). The NAP focuses on managing pluvial floods and does not specifically address the energy system (Government of Denmark, 2012a).	No adaptation actions reported.	No adaptation actions reported.
Estonia		The national CCIV assessment (a part of the NAS) includes a section on the security of the energy supply, including energy independence, security, resources, energy efficiency, heat generation and electricity generation (Government of Estonia, 2017).		The NAP includes a section on energy and security of supply (Government of Estonia, 2017). More specific objectives are included in the Estonian climate adaptation strategy for infrastructure and energy and in the national strategy for climate change in the infrastructure and the energy sector 2017-2030 (Lahtvee et al., 2015a, 2015b).	No adaptation actions reported.	No adaptation actions reported.
Finland		The national CCIV assessment includes a section on energy distribution and production, which identifies risks such as disturbances to energy production and distribution, maintenance cost increases in the energy network, flood risks to nuclear power plants, weaker solar power and increased need for cooling (Government of Finland, 2013).		The NAP acknowledges climate change impacts on energy, but no specific actions are suggested (Government of Finland, 2014).	Long-term climate changes have been included in the legislation on water resources, among which the Dam Safety Act (2009) may be relevant for hydropower. The country fiche mentions that electricity market legislation includes security of supply provisions, but it is not clear whether these cover climate-related issues.	The Electricity Market Act includes regulations aimed at improving the security of the energy supply in network fault situations. Specifically, the distribution network must be designed, built and maintained in such a way that when the network is damaged due to a storm or snow, the loss of power to customers should not exceed 6 hours in urban areas and 36 hours in other areas.
France		The national CCIV assessment includes a section on energy, which addresses climate change impacts on energy consumption, hydropower production and energy transport (Government of France, 2009).		Energy and industry are included as key sectors within the NAP (Government of France, 2011).	No adaptation actions reported.	No adaptation actions reported.

Country				Document		
		ge impact, vulnerability V) assessments	National adaptation strategies (NAS) and plans (NAP)		Country fiches	UNFCCC
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/ Coverage of the energy or NAP sector in NAS and/ adopted or NAP		Adaptation measures in energy sector	nplemented in the
Germany		The national CCIV assessment includes a section on the action field 'Energy industry', with 'strong changes' projected for heating energy demand, cooling energy demand, cooling water for thermal power stations, damage to power stations and production facilities (Government of Germany, 2015b).		The NAP includes energy (conversion, transport and supply) as one of 15 sectors covered (Government of Germany, 2011).	The progress report on the implementation of the NAP mentions adaptation activities in the climate proofing of the energy sector, e.g. in terms of water use for cooling of thermal plants and for hydropower (Government of Germany, 2015a).	The government is implementing 'a comprehensive portfolio of services addressing climate change and adaptation'. This includes the establishment of the German Climate Service in 2015. This service addresses many sectors and impact domains, including energy.
Greece		The national CCIV assessment includes a section on the built environment, which includes power supply networks. Impacts include, among other things, an increase in summer energy demand, a decrease in winter energy demand and damage from fires (Bank of Greece, 2011).		The NAS includes energy as one of the 15 sectors covered (Government of Greece, 2016).	No significant adaptation actions reported.	No adaptation actions reported.
Hungary		The national CCIV assessment includes a section on the energy sector (Government of Hungary, 2017b).		The draft second national climate change strategy contains energy as a priority sector for adaptation actions (Government of Hungary, 2017a).	No significant adaptation actions reported.	No adaptation actions reported.
Iceland		No information available.		No information available.	Country fiches applicable only to EU Member States.	No adaptation actions reported.
Ireland		The most recent national CCIV assessment includes a section on critical infrastructure and the built environment, which includes climate change impacts on energy demand, supply, transport and transmission (Desmond et al., 2017). The NAP for the electricity and gas networks sector also includes a chapter on vulnerability assessment (Government of Ireland, 2018a)		An electricity and gas networks NAP will be prepared by the Department of Communications, Climate Action and Environment for 2019 (Government of Ireland, 2018b). A previous version of this plan was released under the National Climate Change Adaptation Framework of 2012. The focus of the current plan is on electricity generation and electricity and gas transmission and distribution infrastructures and interconnectors.	No adaptation actions reported.	No adaptation actions reported.

Country	Document							
	Climate change impact, vulnerability and risk (CCIV) assessments		National adaptation strategies (NAS) and plans (NAP)		Country fiches	UNFCCC		
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/ or NAP adopted	Coverage of the energy sector in NAS and/ or NAP	Adaptation measures im energy sector	plemented in the		
Italy		CCIV assessment includes a section on climate change impacts on heating and cooling, electricity demand, production of electricity, transmission and distribution (Government of Italy, 2014).		Both the NAS (Government of Italy, 2014) and the first draft of the NAP include a specific section on energy (CMCC, 2017). The NAP includes a priority ranking of measures for energy transformation, transmission, distribution and demand.	No adaptation actions reported.	No adaptation actions reported.		
Latvia		The national CCIV assessment includes sections on the power transmission and distribution systems and on natural gas infrastructure (Government of Latvia, 2017).		No information available. A national Adaptation Strategy is under preparation.	No adaptation actions reported.	No adaptation actions reported.		
Liechtenstein		The seventh national communication to the UNFCCC mentions energy as a vulnerable sector in relation to energy consumption and hydropower generation (Government of Liechtenstein, 2017).		The NAS mentions specific adaptation actions relating to energy (Government of Liechtenstein, 2018).	Country fiches applicable only to EU Member States.	One of Liechtenstein's hydroelectric power plants (Samina) has bee transformed into a pumped-storag plant. This measure increase flexibility in power production and facilitates adaptation to expected change: in the runoff regime due to climate change.		
Lithuania		The national CCIV assessment includes a section that sets out various threats to the energy sector (Government of Lithuania, 2015).		The latest NAP includes various adaptation actions related to the energy sector (Government of Lithuania, 2018).	No adaptation actions reported.	No adaptation actions reported.		
Luxembourg		The national CCIV assessment focuses on spatial planning. Potential climate change impacts on the energy sector are mentioned briefly in a table (Government of Luxembourg, 2012).		The NAS/NAP includes a dedicated section on energy (Government of Luxembourg, 2018).	No domestic adaptation actions were reported. However, Luxembourg participates in adaptation activities of the Benelux countries, which include the energy sector.	No adaptation actions reported.		
Malta		The national CCIV assessment does not explicitly mention the energy sector (Malta Resources Authority, 2010).		Energy is not a priority sector within NAS (Government of Malta, 2012).	No adaptation actions reported.	No adaptation actions reported.		

Country		ge impact, vulnerability		Document daptation strategies	Country fiches	UNFCCC
	Availability of national CCIV assessment	<ul> <li>/) assessments</li> <li>Coverage of the energy sector in national CCIV assessments</li> </ul>	(NAS) and NAS and/ or NAP adopted	plans (NAP) Coverage of the energy sector in NAS and/ or NAP	Adaptation measures im energy sector	plemented in the
Netherlands		The national CCIV assessment includes energy as a key sector (Ligtvoet et al., 2015). In addition, a sectoral CCIV assessment for the energy sector is available (Vogel et al., 2014).		The NAS includes energy as one of nine priority sectors (Government of the Netherlands, 2016). The NAP refers to climate change impacts on the energy sector, but adaptation in this sector is not a priority for 2018-2019 (Government of the Netherlands, 2018).	There are systematic capacity-building initiatives on adaptation with regional and local governments and sector representatives from several sectors, including the energy sector. The Benelux countries have organised exploratory workshops and carried out cross-border risk analysis in several areas, including energy.	Several grid managers have conducted research to determine the potential impact of climate change events such as flooding on their section of the infrastructure. Research has also examined how a large-scale power outage would affect the chain of vital functions.
Norway		The national CCIV assessment includes a section on the vulnerabilities of power supply (Government of Norway, 2010).		The NAS includes aspects relevant to the energy sector in various places (Government of Norway, 2012).	Country fiches applicable only to EU Member States.	Several policies are in place to ensure a reliable power supply in a changing climate. For example, the NVE sets requirements to electricity utilities in terms of proper contingency planning, available spare parts, transport and communication systems, training, etc., to enable an efficient restoration of the electricity supply. Furthermore, NVE conducts research and development in the light of anticipated challenges to the energy sector as a result of climate change.
Poland		The national CCIV assessment includes a section on the energy sector (KLIMADA, 2013).		The NAS includes a separate action line on the energy sector (Government of Poland, 2013).	No significant adaptation actions reported.	No adaptation actions reported.
Portugal		The SIAM II project 'Climate Change in Portugal: Scenarios, Impacts, and Adaptation Measures' addressed climate change impacts on the energy sector, in particular on solar energy, water availability for power, and heating and cooling demand (Aguiar et al., 2006). Furthermore, a report drafted as supporting material for the NAS covers the vulnerabilities of the energy sector (Government of Portugal, 2012).		The latest NAS (ENAAC 2020) includes a separate section on energy (Government of Portugal, 2015).	The 2015-2016 interim evaluation of the implementation progress of the NAS (Government of Portugal, 2016), reports that 'Existing policy around security of energy supply and climate change mitigation are broadly in line with adaptation objectives, including policies to increase networks, adopt smart grids and promoted distributed production. An area identified for further policy development is climate proofing of energy assets.'	No adaptation actions reported.

Country				Document		
		ge impact, vulnerability /) assessments		daptation strategies plans (NAP)	Country fiches	UNFCCC
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/ or NAP adopted	Coverage of the energy sector in NAS and/ or NAP	Adaptation measures im energy sector	plemented in the
Romania		The national climate strategy includes a section on energy (Government of Romania, 2016a).		The NAP includes a separate section on energy (Government of Romania, 2016b).	No significant adaptation actions reported.	No adaptation actions reported.
Slovakia		The national CCIV assessment includes a section on the energy system (Mindas et al., 2011).		The NAS includes a separate section on energy (Government of Slovakia, 2014).	No adaptation actions reported.	No adaptation actions reported.
Slovenia		The national CCIV assessment focuses on water; no energy- specific section could be identified (Government of Slovenia, 2010).		The NAS does not specifically address the energy sector (Government of Slovenia, 2016).	The Flood Risk Mitigation Plan 2017-2021 defines a large set of flood protection projects, which include adapting the energy infrastructure. However, this plan does not currently consider climate change scenarios. At the transnational level, the Water and Climate Adaptation Plan for the Sava River Basin acts as a guidance document to governments for climate adaptation measures in various water-related issues, including hydropower.	No adaptation actions reported.
Spain		The early national CCIV assessment includes a section on energy (Government of Spain, 2005). A detailed assessment for the energy sector was conducted later (Girardi et al., 2015).		The current NAP (PNACC WP3) includes a section on energy (Government of Spain, 2014).	National adaptation activities in Spain relating to the energy sector comprise a detailed sectoral CCIV assessment, a pilot project with Endesa (the largest electric utility company in Spain), capacity-building actions and networking with regional authorities and mainstreaming activities (Government of Spain, 2018).	No adaptation actions reported.

Country				Document		
		ge impact, vulnerability ) assessments		daptation strategies plans (NAP)	Country fiches	UNFCCC
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/ or NAP adopted	Coverage of the energy sector in NAS and/ or NAP	Adaptation measures in energy sector	nplemented in the
Sweden		The national CCIV assessment includes sections on energy production, supply, heating and cooling needs, and dam safety (Andersson et al., 2015).		The Swedish Energy Agency has been requested to identify adaptation measures in the energy sector, but a comprehensive plan does not seem to exist at this time (Government of Sweden, 2019). Adaptation in the energy system may be addressed in some regional adaptation plans, but a comprehensive overview is not available.	The scope of environmental impact assessments, which includes amendments to hydropower, municipal energy planning, pipelines and nuclear activity, was broadened in order to consider climate change impacts (Government of Sweden, 2017). Furthermore, in 2018 the government tasked a number of agencies with creating a sectoral adaptation plan or report on their adaptation work, including the Swedish Energy Agency, Swedish grids and the National Electrical Safety Board.	No adaptation actions reported.
Switzerland		The national CCIV assessment addresses climate change impacts on cooling and heating needs, and the seasonal shift of hydropower generation (Köllner et al., 2017).		The NAP includes a section on energy (Government of Switzerland, 2014).	Country fiches applicable only to EU Member States.	Multiple adaptation measures have been reported at the cantonal level in relation to 'cooling of thermal power plants, hydropower production, thermal power production and energy', but it remains unclear what these measures are and how they relate to climate change projections. At the federal level, several adaptation measures are being planned and implemented, such as taking the changing climate into account in supervision and licensing processes for hydroelectric dams and reservoirs as well as for transmission and distribution networks for gas and electricity. More specific information is available in an annex to the NAP (Government of Switzerland, 2014, Annex A5).

Country				Document		
	•	ge impact, vulnerability ') assessments		daptation strategies plans (NAP)	Country fiches	UNFCCC
	Availability of national CCIV assessment	Coverage of the energy sector in national CCIV assessments	NAS and/ or NAP adopted	Coverage of the energy sector in NAS and/ or NAP	Adaptation measures in energy sector	plemented in the
Turkey		The national CCIV assessment focuses on water-related impacts, but energy-related impacts are addressed briefly in this context (Government of Turkey, 2016).		Energy-related aspects are mentioned in the section on water resources management of the NAS/NAP, with a focus on hydropower (Government of Turkey, 2011).	Country fiches applicable only to EU Member States.	No adaptation actions reported.
United Kingdom		The most recent national CCIV assessment includes a chapter on infrastructure, which covers energy infrastructure (Committee on Climate Change, 2017).		The most recent NAP includes a section on the energy sector (Government of the UK, 2018).	The 2017 progress report on the NAP acknowledged significant progress in increasing the climate resilience of the energy infrastructure, with a focus on flood risk management. A large range of measures have already been carried out, including critical upgrading of electric substations against flooding; investment in mobile flood protection; electricity generators addressing the risk of water scarcity; and risk assessment of nuclear installations and of gas networks. These measures were facilitated by government policies, including reporting requirements for critical infrastructure providers, mandatory climate risk assessments for new infrastructure projects and improved standards for network resilience (Committee on Climate Change, 2017).	The Climate Ready Support Service for England helped organisations to build their own capacity to adapt by incorporating climate risk management into their plans and decision-making and providing guidance tools and support to key sectors (including energy). A national risk assessment and sector resilience plans that consider climate change have identified key risks and vulnerabilities. The Adaptation Reporting Power enables the government to monitor the implementation of risk reduction strategies.

Notes: CCIV, climate change impact, vulnerability and risk; NAP, national adaptation plan; NAS, national adaptation strategy; NVE, Norwegian Water Resources and Energy Directorate.

# Abbreviations

ACERAgency for the Cooperation of Energy RegulatorsARSFifth Assessment Report of the Intergovernmental Panel on Climate ChangeARPAdaptation Reporting PowerC3SCopernicus Climate Change ServiceCCIVClimate change impact, vulnerability and riskCSCarbon capture and storageCDDCooling degree dayCEERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEIAEuropean Investment BankENAEuropean Investment BankENAEuropean Investment BankENAEuropean Network of Transmission System Operators - ElectricityENTSO-EEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERACEuropean Research Area for Climate ServicesERAFEuropean Social FundEUEuropean Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFIMPFiod natural gasEIFALinergovernmental Panel on Climate ChangeFIMPFiod natural gasEUFEuropean Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFIMPFiod natural gas		
ARPAdaptation Reporting PowerC3SCopernicus Climate Change ServiceCCIVClimate change impact, vulnerability and riskCCSCarbon capture and storageCDDCooling degree dayCEERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEIAEuropean Investment BankENAEuropean Investment BankENAEuropean Network of Transmission System Operators - ElectricityENTSO-EEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Climate ServicesERDFEuropean Network of Climate ServicesERDFEuropean Regional Development FundESFEuropean Regional Development FundESFEuropean Sak for Qlimate ServicesERDFEuropean Secolal FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEALiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNA	ACER	Agency for the Cooperation of Energy Regulators
ARPAdaptation Reporting PowerC3SCopernicus Climate Change ServiceCCIVClimate change impact, vulnerability and riskCCSCarbon capture and storageCDDCooling degree dayCEERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEIAEuropean Investment BankENAEuropean Investment BankENAEuropean Network of Transmission System Operators - ElectricityENTSO-EEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Climate ServicesERDFEuropean Network of Climate ServicesERDFEuropean Regional Development FundESFEuropean Regional Development FundESFEuropean Sak for Qlimate ServicesERDFEuropean Secolal FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEALiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNA	AR5	Fifth Assessment Report of the Intergovernmental Panel on Climate Change
C3SCopernicus Climate Change ServiceCCIVClimate change impact, vulnerability and riskCCSCarbon capture and storageCDDCooling degree dayCERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Environment AgencyEIBEuropean Environment AgencyEIBEuropean Nuclear Safety Regulators GroupENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Nuclear Safety Regulators GroupENTSO-5European Network of Transmission System Operators - ElectricityENSREGEuropean Network of Transmission System Operators - GasPCIPEuropean Research Area for Climate ServicesEDFElucropean Research Area for Climate ServicesEDFEuropean Regional Development FundESFEuropean Social FundEUEuropean Institutions Working Group on Adaptation to Climate ChangeFMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEALiquefied natural gasLTSLong-terms trategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation plan<	ARP	
CCIVClimate change impact, vulnerability and riskCCSCarbon capture and storageCDDCooling degree dayCERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEAAEuropean Investment AgencyEIBEuropean Investment AgencyEIBEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERACSEuropean Regional Development FundESFEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Reney AgencyIFCCInternational Reney AgencyIFCAJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategy	C3S	
CCSCarbon capture and storageCDDCooling degree dayCEERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEAAEuropean Environment AgencyEIBEuropean Investment BankENAAEnergy Network AssociationENTSO-GEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundEVEuropean Regional Development FundEVEuropean Social FundEUEuropean UnionEV-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Renewable Energy AgencyIPCCIntersoremental Panel on Climate ChangeRENAInternational Renewable Energy AgencyIPCCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTS<		
CDDCooling degree dayCEERCouncil of European Union Energy RegulatorsCFCohesion FundCRClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEAAEuropean Investment AgencyEIBEuropean Investment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENTSO-EEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Regional Development FundESFEuropean Regional Development FundESFEuropean Social FundEUEuropean Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIERAInternational Energy AgencyJRCJoint Research Centre (of the European Commission)LINGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation p		
CEERCouncil of European Union Energy RegulatorsCFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEAAEuropean Invironment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENSEGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-EEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Regional Development FundESFEuropean Regional Development FundESFEuropean Union Member StatesEUFWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyJRCJoint Research Centre (of the European Commission)LINGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational dorganisationOCCSpanish Office of		
CFCohesion FundCRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEEAEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Regional Development FundESFEuropean Regional Development FundEUEuropean Nuclear Safety Regulators Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesEUFWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIENAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNAS<		
CRIClimate-resilient infrastructureCSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEEAEuropean Environment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNACNational adaptation planNAS </td <td></td> <td></td>		
CSPConcentrated solar powerDSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEEAEuropean Invisronment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Regional Development FundESFEuropean Regional Development FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation strategyNACNational adaptation strategyNACNational adaptation planNASNational adaptation planNA		
DSODistribution system operatorEADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEEAEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERAEuropean Research Area for Climate ServicesERDFEuropean Research Area for Climate ServicesEUFWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Energy AgencyIPCCJoint Research Centre (of the European Commission)LINGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation strategyNCNational energy and climate planNGO		
EADExpected annual damageEBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEEAEuropean Environment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERA4CSEuropean Regional Development FundESFEuropean Operators Operators - ElectricityEUEuropean Regional Development FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational energy and climate planNGCNational energy and climate plan </td <td></td> <td>•</td>		•
EBRDEuropean Bank for Reconstruction and DevelopmentEDFElectricité de FranceEEAEuropean Environment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Regional Development FundESFEuropean NotionEU-28The 28 European UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInterational Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational energy and climate planNGCNor-governmental organisationOCCCSpanish Office of Climate Change		
EDFElectricité de FranceEEAEuropean Environment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Nuclear Safety Regulators GroupENTSO-GEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Programme for Critical Infrastructure ProtectionERA4CSEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Renergy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONotional adaptation planNASNational adaptation planNASNational adaptation strategyNCNational denregy AgencyNCNational denregy and climate planNASNational adaptation strategyNCNational adaptation strategyNAGNorabernmental organisationOCCSpanish Office of Climate Change		
EEAEuropean Environment AgencyEIBEuropean Investment BankENAEnergy Network AssociationENAREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Network of Transmission System Operators - GasEPCIPEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Cocial FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Renergy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renergy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational adaptation strategyNCNational energy and climate planNASNational adaptation strategyNCNational energy and climate planNASNational adaptation strategyNCNational energy and climate planN		
EIBEuropean Investment BankENAEnergy Network AssociationENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasPPCIPEuropean Network of Transmission System Operators - GasERA4CSEuropean Network of Transmission System Operators - GasERDFEuropean Regional Development FundESFEuropean OnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation strategyNACNational adaptation strategyNCNational lenergy and climate planNBCPNational lenergy and climate plan		
ENAEnergy Network AssociationENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Programme for Critical Infrastructure ProtectionERA4CSEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCCNational denergy and climate planNBCNational energy and climate planNBCNational energy and climate planNBCNational energy and climate planNBCNational energy and climate planNBCNational energ		
ENSREGEuropean Nuclear Safety Regulators GroupENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Programme for Critical Infrastructure ProtectionERA4CSEuropean Regearch Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoin Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational adaptation strategyNCNational energy and climate planNECPNational energy and climate planNECPNational energy and climate planNECPNational adaptation strategyNCNational adaptation strategyNCNational energy and climate planNECPNational energy and climate planNECPSpanish Office of Climate Change		
ENTSO-EEuropean Network of Transmission System Operators - ElectricityENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Programme for Critical Infrastructure ProtectionERA4CSEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Social FundEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAAInternational Energy AgencyIPCCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational denergy and climate planNECPNational adaptation gansLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation strategyNAONorth Atlantic OscillationNACNational adaptation strategyNCNational energy and climate planNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
ENTSO-GEuropean Network of Transmission System Operators - GasEPCIPEuropean Programme for Critical Infrastructure ProtectionERA4CSEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational adaptation strategyNCNational adaptation ganisationNDCNational energy and climate planNECPNational energy and climate planNGONon-governmental organisation	ENSREG	
EPCIPEuropean Programme for Critical Infrastructure ProtectionERA4CSEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNACNorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation strategyNCNational communicationNDCNational domunicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	ENTSO-E	
ERA4CSEuropean Research Area for Climate ServicesERDFEuropean Regional Development FundESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation planNASNational communicationNDCNational adaptation planNASNational adaptation planNASNational adaptation planNASNational adaptation strategyNCNational energy and climate planNECPNational energy and climate planNGONor-governmental organisationOECCSpanish Office of Climate Change	ENTSO-G	European Network of Transmission System Operators – Gas
ERDFEuropean Regional Development FundESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational adaptation strategyNCNational domunicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	EPCIP	European Programme for Critical Infrastructure Protection
ESFEuropean Social FundEUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAFNational adaptation planNASNational adaptation strategyNCNational communicationNDCNational dogration strategyNCNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	ERA4CS	European Research Area for Climate Services
EUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	ERDF	European Regional Development Fund
EUEuropean UnionEU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	ESF	
EU-28The 28 European Union Member StatesEUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNational energy and climate planNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	EU	
EUFIWACCEU Financial Institutions Working Group on Adaptation to Climate ChangeFRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	EU-28	
FRMPFlood risk management planGFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
GFCSGlobal Framework for Climate ServicesGHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
GHGGreenhouse gasHDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
HDDHeating degree dayIEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
IEAInternational Energy AgencyIPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
IPCCIntergovernmental Panel on Climate ChangeIRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
IRENAInternational Renewable Energy AgencyJRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
JRCJoint Research Centre (of the European Commission)LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
LNGLiquefied natural gasLTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
LTSLong-term strategyNAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change	•	
NAONorth Atlantic OscillationNAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
NAPNational adaptation planNASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
NASNational adaptation strategyNCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
NCNational communicationNDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
NDCNationally determined contributionNECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
NECPNational energy and climate planNGONon-governmental organisationOECCSpanish Office of Climate Change		
NGONon-governmental organisationOECCSpanish Office of Climate Change		•
OECC Spanish Office of Climate Change		
		• •
OECD Organisation for Economic Cooperation and Development		
	OECD	Organisation for Economic Cooperation and Development

PBL PKW	Netherlands Environmental Assessment Agency Piano key weir
PV	Photovoltaics
RBMP	River basin management plan
RCP	Representative concentration pathway
RES	Renewable energy sources
TEN-E	Trans-European Networks for Energy
TNO	Dutch Organisation for Applied Scientific Research
TSO	Transmission system operator
TYNDP	Ten year network development plan
UNEP	United Nations Environment Programme
UNFCCC	United Nations Framework Convention on Climate Change
VOLL	Value of lost load
WBCSD	World Business Council for Sustainable Development
WEC	World Economic Council
WFD	Water Framework Directive
WPD	Western Power Distribution

## References

Aguiar, R., et al., 2006, '*Energia*', in: *Iterações Climáticas em Portugal. Cenários, Impactos e Medidas de Adaptação* — *Projecto SIAM II*, Lisbon, Portugal.

Alfieri, L., et al., 2015, 'Global warming increases the frequency of river floods in Europe', *Hydrology and Earth System Sciences* 19(5), pp. 2247-2260 (DOI: 10.5194/hess-19-2247-2015).

Andersson, L., et al., 2015, *Basis for the checkpoint-2015 for adapting to a changing climate*, SMHI Klimatologi, Norrköping, Sweden (http://www.smhi.se/ polopoly\_fs/1.86329!/Menu/general/extGroup/ attachmentColHold/mainCol1/file/Klimatologi%20 Nr%2012.pdf) accessed 19 September 2018.

APCC, 2014, Austrian assessment report climate change 2014 (AAR14). Synopsis — main findings, Austrian Panel on Climate Change, Vienna (https://www.ccca. ac.at/fileadmin/00\_DokumenteHauptmenue/03\_ Aktivitaeten/APCC/summarys/Synopse\_englisch\_ finaleversion\_181214.pdf) accessed 10 April 2019.

Arent, D. J., et al., 2014, 'Key economic sectors and services', in: Field, C. B. et al. (eds), *Climate change 2014: impacts, adaptation, and vulnerability. Part A: global and sectoral aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*, Cambridge University Press, Cambridge, pp. 659-708.

Auffhammer, M. and Mansur, E. T., 2014, 'Measuring climatic impacts on energy consumption: A review of the empirical literature', *Energy Economics* 46, pp. 522-530 (DOI: 10.1016/j.eneco.2014.04.017).

Azevedo de Almeida, B. and Mostafavi, A., 2016, 'Resilience of infrastructure systems to sea-level rise in coastal areas: impacts, adaptation measures, and implementation challenges', *Sustainability* 8(11), p. 1115 (DOI: 10.3390/su8111115).

Bank of Greece, 2011, *The environmental, economic, and social impacts of climate change in Greece*, Bank of Greece, Athens, Greece (http://www.bankofgreece. gr/BoGEkdoseis/ClimateChange\_FullReport\_bm.pdf) accessed 19 September 2018. Barnett, J. and O'Neill, S., 2010, 'Maladaptation', *Global Environmental Change* 20(2), pp. 211-213 (DOI: 10.1016/j.gloenvcha.2009.11.004).

Bartók, B., et al., 2017, 'Projected changes in surface solar radiation in CMIP5 global climate models and in EURO-CORDEX regional climate models for Europe', *Climate Dynamics* 49(7-8), pp. 2665-2683 (DOI: 10.1007/s00382-016-3471-2).

Becker, S., et al., 2014, 'Transmission grid extensions during the build-up of a fully renewable pan-European electricity supply', *Energy* 64, pp. 404-418 (DOI: 10.1016/j.energy.2013.10.010).

Behrens, P., et al., 2017, 'Climate change and the vulnerability of electricity generation to water stress in the European Union', *Nature Energy* 2(8), p. 17114 (DOI: 10.1038/nenergy.2017.114).

Bell, R. J., et al., 2017, 'North Atlantic storm driving of extreme wave heights in the North Sea', *Journal of Geophysical Research: Oceans* 122(4), pp. 3253-3268 (DOI: 10.1002/2016JC012501).

Berrill, P., et al., 2016, 'Environmental impacts of high penetration renewable energy scenarios for Europe', *Environmental Research Letters* 11(1), p. 014012 (DOI: 10.1088/1748-9326/11/1/014012).

Bisselink, B., et al., 2018, *Impact of a changing climate, land use, and water usage on Europe's water resources,* JRC Technical Report No EUR 29130 EN, Publication Office of the European Union, Luxembourg (http://doi.org/10.2760/09027).

Bloemendal, M. and Jaxa-Rozen, M., 2018, 'The hidden side of cities: using aquifer thermal energy storage for energy saving', *Science Trends* (DOI: 10.31988/SciTrends.13681).

Bonelli, P., et al., 2011, 'Wet snow hazard for power lines: a forecast and alert system applied in Italy', *Natural Hazards and Earth System Science* 11(9), pp. 2419-2431 (DOI: 10.5194/nhess-11-2419-2011). Bonjean Stanton, M. C., et al., 2016, 'A systematic review of the impacts of climate variability and change on electricity systems in Europe', *Energy* 109, pp. 1148-1159 (DOI: 10.1016/j.energy.2016.05.015).

Boston, A., 2013, 'Delivering a secure electricity supply on a low carbon pathway', *Energy Policy* 52, pp. 55-59 (DOI: 10.1016/j.enpol.2012.02.004).

Böttcher, C., 2016, 'The cost of blackouts in Europe', CORDIS (https://cordis.europa.eu/news/rcn/132674/en) accessed 31 July 2018.

Braun, M. and Fournier, E., 2016, *Adaptation case studies in the energy sector overcoming barriers to adaptation*, Natural Resources Canada (https://www.ouranos.ca/ publication-scientifique/RapportBraun\_Fournier2017\_ En.pdf) accessed 10 April 2019.

Brown, S., et al., 2014, 'Implications of sea-level rise and extreme events around Europe: a review of coastal energy infrastructure', *Climatic Change* 122(1-2), pp. 81-95 (DOI: 10.1007/s10584-013-0996-9).

Buceti, G., 2015, 'Climate change and vulnerabilities of the European Energy Balance', *Journal of Sustainable Development of Energy, Water and Environment Systems* 3(1), pp. 106-117 (DOI: 10.13044/j. sdewes.2015.03.0008).

Byers, E. A., et al., 2016, 'Water and climate risks to power generation with carbon capture and storage', *Environmental Research Letters* 11(2), p. 024011 (DOI: 10.1088/1748-9326/11/2/024011).

CAL Fire, 2018a, CAL FIRE Investigators determine causes of 12 wildfires in Mendocino, Humboldt, Butte, Sonoma, Lake, and Napa Counties, CAL FIRE News release, California Department of Forestry and Fire Protection (https://calfire.ca.gov/communications/downloads/ newsreleases/2018/2017\_WildfireSiege\_Cause.pdf) accessed 10 April 2019.

CAL Fire, 2018b, *CAL FIRE Investigators determine the cause of the Cascade fire*, CAL FIRE News release, California Department of Forestry and Fire Protection (http://calfire.ca.gov/communications/downloads/ newsreleases/2018/Cascade%20Fire%20Cause%20 Release.pdf) accessed 10 April 2019.

Camia, A., et al., 2013, *Harmonized classification scheme* of fire causes in the EU adopted for the European Fire Database of EFFIS, JRC 80682, EUR 25923 EN, European Commission, Joint Research Centre (http://doi.org/10.2788/86859). Camia, A., et al., 2018, *Biomass production, supply, uses and flows in the European Union: first results from an integrated assessment*, JRC 109869, EUR 28993 EN, Joint Research Centre (http://doi.org/10.2760/539520) accessed 13 March 2019.

Capela Lourenço, T., et al., eds., 2014, *Adapting to an uncertain climate: lessons from practice*, Springer, Cham.

Capriolo, A., 2016, *Strategies for enhancing policy coherence: mainstreaming adaptation into key-sectoral and development policies*, BASE Deliverable 7.1, BASE project (https://base-adaptation.eu/sites/default/ files/D\_7.1\_final\_web.pdf) accessed 10 April 2019.

Carvalho, D., et al., 2017, 'Potential impacts of climate change on European wind energy resource under the CMIP5 future climate projections', *Renewable Energy* 101, pp. 29-40 (DOI: 10.1016/j. renene.2016.08.036).

Castelle, B., et al., 2018, 'Increased winter-mean wave height, variability, and periodicity in the Northeast Atlantic over 1949-2017', *Geophysical Research Letters* 45(8), pp. 3586-3596 (DOI: 10.1002/2017GL076884).

CEER, 2012, Annual Report of the Council of European Energy Regulators, Council of European Energy Regulators, Brussels (https://www.ceer.eu/ documents/104400/-/-/4d3a2826-0f9f-ebff-5c40-5f1bec1c21c2) accessed 9 May 2018.

CEN-CENELEC, 2016, *CEN-CENELEC Guide 32: Guide for addressing climate change adaptation in standardisation*, European Commission for Standardization, Brussels (ftp://ftp.cencenelec.eu/EN/EuropeanStandardization/ Guides/32\_CENCLCGuide32.pdf) accessed 6 January 2018.

CEN-CENELEC, 2018, 'Climate change adaptation' (https://www.cencenelec.eu/standards/sectors/ climatechange/pages/default.aspx) accessed 1 June 2018.

Chandel, M. K., et al., 2011, 'The potential impacts of climate-change policy on freshwater use in thermoelectric power generation', *Energy Policy* 39(10), pp. 6234-6242 (DOI: 10.1016/j.enpol.2011.07.022).

Chandramowli, S. N. and Felder, F. A., 2014, 'Impact of climate change on electricity systems and markets — a review of models and forecasts', *Sustainable Energy Technologies and Assessments* 5, pp. 62-74 (DOI: 10.1016/j.seta.2013.11.003).

Ciscar, J. C., et al., 2018, *Climate impacts in Europe: final report of the JRC PESETA III project*, JRC Science for Policy Report No EUR 29427 EN, Publications Office of the European Union (http://doi.org/10.2760/93257) accessed 29 November 2018.

Climate Resilience Secretariat, 2018, 'Creating climate resilience guidelines for the hydropower sector', International Hydropower Association (https://www. hydropower.org/sites/default/files/publications-docs/ climate\_resilience\_guidelines\_-\_two-pager.pdf) accessed 10 April 2019.

CMCC, 2017, *Italy Draft Adaptation Plan*, Ministero dell'Ambiente e della Tutela del Territorio e del Mare (http://www.minambiente.it/sites/default/files/archivio\_immagini/adattamenti\_climatici/documento\_pnacc\_luglio\_2017.pdf) accessed 20 September 2018.

Committee on Climate Change, 2017, *Progress in preparing for climate change. 2017 Report to Parliament,* Committee on Climate Change, London (https://www.theccc.org.uk/wp-content/uploads/2017/06/2017-Report-to-Parliament-Progress-in-preparing-for-climate-change.pdf) accessed 20 September 2018.

Cortekar, J. and Groth, M., 2015, 'Adapting energy infrastructure to climate change — is there a need for government interventions and legal obligations within the German 'Energiewende'?', *Energy Procedia* 73, pp. 12-17 (DOI: 10.1016/j.egypro.2015.07.552).

Cosentino, S. L., et al., 2012, 'Future yields assessment of bioenergy crops in relation to climate change and technological development in Europe', *Italian Journal of Agronomy* 7(2), p. 22 (DOI: 10.4081/ija.2012.e22).

Covenant of Mayors Office, 2015, 'Covenant of Mayors for Climate and Energy' (https://www. covenantofmayors.eu/en/) accessed 13 February 2019.

COWI, 2016, *Mainstreaming of climate action into ESI funds, Final report*, EC (https://ec.europa.eu/clima/sites/ clima/files/budget/docs/report\_mainstreaming\_of\_ climate\_action\_en.pdf) accessed 26 September 2018.

COWI, 2017, Mainstreaming of adaptation into the ESIF 2014-2020, Final report, EC (https://ec.europa.eu/clima/ sites/clima/files/budget/docs/report\_maindtreaming\_ adaptation\_en.pdf) accessed 26 September 2018.

Cronin, J., et al., 2018, 'Climate change impacts on the energy system: a review of trends and gaps', *Climatic Change* 151(2), pp. 79-93 (DOI: 10.1007/s10584-018-2265-4).

Cruz, A. M. and Krausmann, E., 2013, 'Vulnerability of the oil and gas sector to climate change and extreme weather events', *Climatic Change* 121(1), pp. 41-53 (DOI: 10.1007/s10584-013-0891-4).

Dale, N., et al., 2018, *Proposal for a National Climate Change Adaptation Strategy and Action Plan*, Government of Bulgaria (https://www.moew.government.bg/static/ media/ups/tiny/%D0%9F%D0%98%D0%9A/%D0%9C%D 0%9F%D0%90/Strategy%20and%20Action%20Plan%20 -%20Full%20Report%20-%20Draft%20(2018-05-03)%20 -%20EN%20V4.EM.pdf) accessed 19 September 2018.

Dallamaggiore, E., et al., 2016, *Energy system stakeholder characterisation*, ENTRUST Deliverable D2.1, ENTRUST project (http://www.entrust-h2020.eu/wp-content/ uploads/2017/01/D2.1\_Energy-System-Charact\_release-v2.pdf) accessed 6 January 2018.

Damm, A., et al., 2017, 'Impacts of +2 °C global warming on electricity demand in Europe', *Climate Services* 7, pp. 12-30 (DOI: 10.1016/j.cliser.2016.07.001).

Davy, R., et al., 2018, 'Climate change impacts on wind energy potential in the European domain with a focus on the Black Sea', *Renewable and Sustainable Energy Reviews* 81, pp. 1652-1659 (DOI: 10.1016/j. rser.2017.05.253).

Dawson, R. J., et al., 2016, *UK climate change risk assessment evidence report: Chapter 4, Infrastructure*, Adaptation Sub-Committee of the Committee on Climate Change, London (https://www.theccc.org.uk/ wp-content/uploads/2016/07/UK-CCRA-2017-Chapter-4-Infrastructure.pdf).

Dawson, R. J., et al., 2018, 'A systems framework for national assessment of climate risks to infrastructure', *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences* 376(2121), p. 20170298 (DOI: 10.1098/rsta.2017.0298).

DECC, 2010, Appraisal of sustainability. Site report for Dungeness. EN-6: Revised draft national policy statement for nuclear power generation, Department of Energy and Climate Change, London (https://assets.publishing. service.gov.uk/government/uploads/system/uploads/ attachment\_data/file/47793/1976-aos-site-reportdungeness-en6.pdf) accessed 10 April 2019.

de Rigo, D., et al., 2017, Forest fire danger extremes in Europe under climate change: variability and uncertainty, JRC Technical Report No EUR 28926 EN, Publications Office of the European Union (http://dx.doi. org/10.2760/13180). Desmond, M., et al., 2017, *A summary of the state of knowledge on climate change impacts for Ireland*, EPA Research Report No 223, Environmental Protection Agency, Dublin (https://www.epa.ie/pubs/reports/ research/climate/EPA%20RR%20223\_web.pdf) accessed 10 April 2019.

Despres, J. and Kitous, A., 2018, Assessment of the impact of climate scenarios on residential energy demand for heating and cooling, JRC Science for Policy Report No EUR 29084 EN, Publications Office of the European Union (http://doi.org/10.2760/96778) accessed 29 November 2018.

Dokulil, M., 2013, 'Impact of climate warming on European inland waters', *Inland Waters* 4(1), pp. 27-40 (DOI: 10.5268/IW-4.1.705).

Donat, M. G., et al., 2011, 'Future changes in European winter storm losses and extreme wind speeds inferred from GCM and RCM multi-model simulations', *Natural Hazards and Earth System Sciences* 11(5), pp. 1351-1370 (DOI: 10.5194/nhess-11-1351-2011).

Dowling, P., 2013, 'The impact of climate change on the European energy system', *Energy Policy* 60(C), pp. 406-417 (DOI: 10.1016/j.enpol.2013.05.093).

Dubus, L., 2010, 'Practices, needs and impediments in the use of weather/climate information in the electricity sector', in: Troccoli, A. (ed.), *Management of weather and climate risk in the energy industry*, Springer Netherlands, Dordrecht, pp. 175-188.

EASAC, 2019, Forest bioenergy, carbon capture and storage, and carbon dioxide removal: an update, EASAC Commentary, European Academies Science Advisory Council (https://easac.eu/projects/details/forestbioenergy-carbon-capture-and-storage-and-carbondioxide-removal-an-update/) accessed 10 April 2019.

EASE/EERA, 2017, European energy storage technology development roadmap. 2017 update., The European Association for Storage of Energy, European Energy Research Alliance, Brussels (https://eera-es.eu/ wp-content/uploads/2016/03/EASE-EERA-Storage-Technology-Development-Roadmap-2017-HR.pdf) accessed 9 May 2018.

Ebinger, J. and Vergara, W., 2011, *Climate impacts on energy systems: key issues for energy sector adaptation*, World Bank, Washington, DC.

EBRD, 2018, Building resilience to climate change: investing in adaptation (https://www.ebrd.com/cs/Satelli te?c=Content&cid=1395244437607&pagename=EBRD% 2FContent%2FDownloadDocument) accessed 16 August 2018.

EC, 2010a, Energy 2020: A strategy for competitive, sustainable and secure energy (COM (2010) 639 final).

EC, 2010b, Energy infrastructure. Priorities for 2020 and beyond — a blueprint for an integrated European energy network (COM(2010) 677 final).

EC, 2011a, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions — a budget for Europe 2020. Part II: policy fiches (COM(2011) 500 final).

EC, 2011b, Non-paper guidelines for project managers: Making vulnerable investments climate resilient, European Commission, Brussels (https://climate-adapt.eea. europa.eu/metadata/guidances/non-paper-guidelinesfor-project-managers-making-vulnerable-investmentsclimate-resilient/guidelines-for-project-managers.pdf) accessed 6 January 2018.

EC, 2013a, Commission Staff Working Document, 'Adapting Infrastructure to Climate Change' (SWD(2013) 137 final).

EC, 2013b, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions — an EU strategy on adaptation to climate change (COM(2013) 216 final).

EC, 2013c, 'Europe to experience pumped storage boom', *SETIS Magazine*, 2013 (https://setis.ec.europa.eu/ setis-reports/setis-magazine/power-storage/europeexperience-pumped-storage-boom) accessed 26 September 2018.

EC, 2014, Communication from the Commission to the European Parliament and the Council — European energy security strategy (COM (2014) 330 final).

EC, 2015a, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee, the Committee of the Regions and the European Investment Bank — a framework strategy for a resilient energy unon with a forward-looking climate change policy (COM(2015) 80 final).

EC, 2015b, Potential for climate action. Examples of how to mainstream climate action and the potential for doing so: European Social Fund 2014-2020, Publication Office of the European Union (https://ec.europa.eu/clima/sites/ clima/files/docs/05-climate\_mainstreaming\_fact\_sheetesf\_en.pdf) accessed 25 October 2018.

EC, 2016a, *Climate change and major projects*, European Commission, Brussels, Belgium (https://ec.europa.eu/ clima/sites/clima/files/docs/major\_projects\_en.pdf) accessed 6 January 2018.

EC, 2016b, Commission Staff Working Document — Evaluation report covering the evaluation of the EU's regulatory framework for electricity market design and consumer protection in the fields of electricity and gas (SWD(2016) 412 final).

EC, 2016c, Commission Staff Working Document — Review of available information accompanying the document Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on an EU strategy for heating and cooling (SWD(2016) 24 final).

EC, 2016d, Commission Staff Working Document Accompanying the document Communication from the Commission to the European Parliament and the Council Mid-term review/revision of the multiannual financial framework 2014-2020 An EU budget focused on results (SWD(2016) 299 final).

EC, 2016e, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: An EU Strategy on Heating and Cooling (COM(2016) 51 final).

EC, 2016f, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee, the Committee of the Regions and the European Investment Bank — Clean energy for all Europeans (COM(2016) 860 final).

EC, 2016g, EU Reference Scenario 2016 — Energy, transport and GHG emissions — Trends to 2050, Publications Office of the European Union, Luxembourg (https://ec.europa.eu/energy/sites/ener/files/ documents/ref2016\_report\_final-web.pdf) accessed 10 April 2019.

EC, 2016h, 'Financing adaptation', European Commission (https://ec.europa.eu/clima/policies/ adaptation/financing\_en) accessed 4 October 2018.

EC, 2016i, Impact assessment accompanying the document Proposal for a Regulation of the European Parliament and of the Council on risk-preparedness in the electricity sector and repealing Directive 2005/89/ EC (COM(2016) 410 final).

EC, 2016j, Proposal for a Regulation of the European Parliament and of the Council on the internal market for electricity (COM(2016) 861 final).

EC, 2017a, Commission Staff Working Document — Energy storage — the role of electricity (SWD (2017) 61 final).

EC, 2017b, Commission Staff Working Document — Second report on the state of the Energy Union. Monitoring progress towards the Energy Union objectives — key indicators (SWD (2017) 32 final).

EC, 2017c, Communication from the Commission — Guidelines on non-financial reporting (OJ C 215, 5.7.2017, p. 1-20).

EC, 2017d, Study on 'residential prosumers' in the European Energy Union, JUST/2015/CONS/FW/ C006/0127, European Commission (https://ec.europa. eu/commission/sites/beta-political/files/studyresidential-prosumers-energy-union\_en.pdf) accessed 10 April 2019.

EC, 2017e, 'Sustainable finance' (https://ec.europa.eu/ info/business-economy-euro/banking-and-finance/ sustainable-finance\_en) accessed 1 June 2018.

EC, 2017f, Third Report on the State of the Energy Union (COM(2017) 688 final).

EC, 2018a, '2030 Energy strategy' (https://ec.europa. eu/energy/en/topics/energy-strategy-and-energyunion/2030-energy-strategy) accessed 6 November 2018.

EC, 2018b, A Modern Budget for a Union that Protects, Empowers and Defends: The Multiannual Financial Framework for 2021-2027 (COM (2018) 321 final).

EC, 2018c, 'Building the Energy Union' (https:// ec.europa.eu/energy/en/topics/energy-strategy-andenergy-union/building-energy-union) accessed 1 June 2018.

EC, 2018d, 'CEF Energy', Innovation and Networks Executive Agency (https://ec.europa.eu/inea/en/ connecting-europe-facility/cef-energy) accessed 3 June 2018.

EC, 2018e, 'Climate change adaptation. Research, science and innovation', European Commission (https:// ec.europa.eu/info/sites/info/files/climate\_change\_ adaptation\_booklet\_2018.pdf) accessed 10 April 2019. EC, 2018f, 'Climate services' (https://ec.europa.eu/ research/environment/index.cfm?pg=climate\_services) accessed 14 December 2018.

EC, 2018g, Commission communication in the framework of the implementation of Commission Regulation (EU) No 206/2012 implementing Directive 2009/125/EC of the European Parliament and of the Council with regard to ecodesign requirements for air conditioners and comfort fans and of Commission Delegated Regulation (EU) No 626/2011 supplementing Directive 2010/30/EU of the European Parliament and of the Council with regard to energy labelling of air conditioners (OJ C 92, 9.3.2018, p. 98-100).

EC, 2018h, Commission Staff Working Document — Adaptation preparedness scoreboard. Country fiches accompanying the document from the Commission to the European Parliament and the Council on the implementation of the EU strategy on adaptation to climate change (SWD(2018) 460 final).

EC, 2018i, Commission Staff Working Document. Evaluation of the EU Strategy on adaptation to climate change. Accompanying the document Report from the Commission to the European Parliament and the Council on the implementation of the EU Strategy on adaptation to climate change (SWD(2018) 461 final).

EC, 2018j, Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee, the Committee of the Regions and the European Investment Bank — A clean planet for all — A European strategic long-term vision for a prosperous, modern, competitive and climate neutral economy (COM(2018) 773 final, Brussels, 28.11.2018).

EC, 2018k, Communication from the Commission to the European Parliament, the European Council, the European Central Bank, the European Economic and Social Committee and the Committee of the Regions — Action plan: financing sustainable growth (COM(2018) 97 final).

EC, 2018l, 'Electricity interconnection targets', Energy (https://ec.europa.eu/energy/en/topics/infrastructure/ projects-common-interest/electricity-interconnection-targets) accessed 14 September 2018.

EC, 2018m, 'Energy infrastructure package' (https:// acer.europa.eu/en/Gas/Infrastructure\_development/ Pages/Energy-Infrastructure-Package.aspx) accessed 6 November 2018. EC, 2018n, 'Energy Strategy and Energy Union' (https:// ec.europa.eu/energy/en/topics/energy-strategy-andenergy-union) accessed 1 June 2018.

EC, 2018o, 'Evaluation of Council Directive 2008/114 of 8 December 2008 on the identification and designation of European critical infrastructures and the assessment of the need to improve their protection' (https://ec.europa.eu/home-affairs/what-is-new/ work-in-progress/initiatives/evaluation-councildirective-2008-114\_en) accessed 14 January 2019.

EC, 2018p, *Financing a sustainable European economy*, Final report of the High-Level Expert Group on Sustainable Finance, European Commission (https:// ec.europa.eu/info/publications/180131-sustainablefinance-report\_en) accessed 2 May 2018.

EC, 2018q, *In-depth analysis in support of the Commission Communication COM(2018) 773: A clean planet for all* — *A European long-term strategic vision for a prosperous, modern, competitive and climate neutral economy,* European Commission (https://ec.europa.eu/clima/ sites/clima/files/docs/pages/com\_2018\_733\_analysis\_ in\_support\_en\_0.pdf) accessed 28 November 2018.

EC, 2018r, 'Market legislation', Energy (https:// ec.europa.eu/energy/en/topics/markets-andconsumers/market-legislation) accessed 14 September 2018.

EC, 2018s, Proposal for a Regulation of the European Parliament and of the Council establishing the Connecting Europe Facility (COM(2018) 438 final).

EC, 2018t, Proposal for a Regulation of the European Parliament and of the Council on the establishment of a framework to facilitate sustainable investment (COM(2018) 353 final).

EC, 2018u, *Quarterly report on European electricity markets*, Volume 10, Issue 4, European Commission, Brussels (https://ec.europa.eu/energy/sites/ener/files/ documents/quarterly\_report\_on\_european\_electricity\_ markets\_q4\_2017\_final.pdf) accessed 10 April 2019.

EC, 2018v, Report from the Commission to the European Parliament and the Council on the implementation of the EU Strategy on adaptation to climate change (COM(2018) 738 final).

EC, 2018w, 'Trans-European Networks for Energy' (https://ec.europa.eu/energy/en/topics/infrastructure/ trans-european-networks-energy) accessed 1 June 2018. EC, 2019a, COPERNICUS Market report — February 2019, European Commission (https://www.copernicus.eu/ sites/default/files/2019-02/PwC\_Copernicus\_Market\_ Report\_2019\_PDF\_version.pdf) accessed 10 April 2019.

EC, 2019b, Fourth Report on the State of the Energy Union (COM(2019) 175 final).

EC, 2019c, 'National energy and climate plans (NECPs)', Energy — European Commission (https://ec.europa.eu/ energy/en/topics/energy-strategy-and-energy-union/ governance-energy-union/national-energy-climateplans) accessed 26 February 2019.

EC, 2019d, 'Projects of common interest' (https:// ec.europa.eu/energy/en/topics/infrastructure/projectscommon-interest) accessed 12 March 2019.

ECA, 2016, Spending at least one euro in every five from the EU budget on climate action: ambitious work underway, but at serious risk of falling short, Special report No 31/2016, European Court of Auditors, Luxembourg (https://www.eca.europa.eu/en/Pages/ Docltem.aspx?did=39853) accessed 10 April 2019.

Ecofys, 2017, Assessing adaptation knowledge in Europe: infrastructure resilience in the transport, energy and construction sectors, European Commission (https:// ec.europa.eu/clima/sites/clima/files/adaptation/what/ docs/infrastructure\_resilience\_en.pdf) accessed 10 April 2019.

EDSO for Smart Grids, 2018, 'EDSO for Smart Grids' (https://www.edsoforsmartgrids.eu/) accessed 1 June 2018.

EEA, 2013, Adaptation in Europe — addressing risks and opportunities from climate change in the context of socio-economic developments, EEA Report No 3/2013, European Environment Agency (http://www.eea. europa.eu/publications/adaptation-in-europe) accessed 4 March 2014.

EEA, 2014, Adaptation of transport to climate change in Europe, EEA Report No 8/2014, European Environment Agency (http://www.eea.europa.eu/publications/ adaptation-of-transport-to-climate) accessed 10 April 2019.

EEA, 2015, Overview of climate change adaptation platforms in Europe, EEA Technical Report No 5/2015, European Environment Agency (http://www.eea. europa.eu/publications/overview-of-climate-changeadaptation) accessed 10 April 2019. EEA, 2016a, 'River flow (CLIM 016)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-103-en) accessed 8 June 2018.

EEA, 2016b, 'Soil moisture (LSI 007)' (https://www.eea. europa.eu/ds\_resolveuid/IND-201-en) accessed 8 June 2018.

EEA, 2016c, 'Water temperature (CLIM 019)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-202-en) accessed 8 June 2018.

EEA, 2017a, *Climate change adaptation and disaster risk reduction in Europe: enhancing coherence of the knowledge base, policies and practices*, EEA Report No 15/2017, European Environment Agency (https:// www.eea.europa.eu/publications/climate-changeadaptation-and-disaster) accessed 10 April 2019.

EEA, 2017b, *Climate change, impacts and vulnerability in Europe 2016 — an indicator-based report*, EEA Report No 1/2017, European Environment Agency (https:// www.eea.europa.eu/publications/climate-changeimpacts-and-vulnerability-2016) accessed 10 April 2019.

EEA, 2017c, 'Hail (CLIM 053)', European Environment Agency (https://www.eea.europa.eu/ds\_resolveuid/IND-424-en) accessed 11 June 2018.

EEA, 2017d, 'Mean precipitation (CLIM 002)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-91-en) accessed 9 April 2019.

EEA, 2017e, *Renewable energy in Europe 2017 — recent growth and knock-on effects*, EEA Report No 3/2017, European Environment Agency (https://www.eea. europa.eu/publications/renewable-energy-in-europe-2017) accessed 26 June 2017.

EEA, 2017f, 'River floods (CLIM 017)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-104-en) accessed 9 April 2019.

EEA, 2017g, *The Arctic environment. European perspectives on a changing Arctic*, EEA Report No 7/2017, European Environment Agency (https://www.eea. europa.eu/publications/the-arctic-environment) accessed 10 April 2019.

EEA, 2017h, 'Wind storms (CLIM 005)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-93-en) accessed 11 June 2018.

EEA, 2018a, 'Adapting overhead lines in response to increasing temperatures in UK', Climate-ADAPT case study (https://climate-adapt.eea.europa.eu/metadata/

case-studies/adapting-overhead-lines-in-response-toincreasing-temperatures-in-uk) accessed 11 April 2019.

EEA, 2018b, 'Electric vehicles as a proportion of the total fleet (TERM 034)', European Environment Agency (https://www.eea.europa.eu/data-and-maps/indicators/ proportion-of-vehicle-fleet-meeting-4/assessment-2) accessed 8 November 2018.

EEA, 2018c, 'Flood risk management for hydropower plants in France', Climate-ADAPT case study (https:// climate-adapt.eea.europa.eu/metadata/case-studies/fd) accessed 11 April 2019.

EEA, 2018d, 'Hydropower expansion and improved management in response to increased glacier melt in Iceland', Climate-ADAPT case study (https:// climate-adapt.eea.europa.eu/metadata/case-studies/ hydropower-expansion-and-improved-management-inresponse-to-increased-glacier-melt-in-iceland) accessed 11 April 2019.

EEA, 2018e, 'Improved resilience of biomass fuel supply chain in UK', Climate-ADAPT case study (https://climateadapt.eea.europa.eu/metadata/case-studies/improvedresilience-of-biomass-fuel-supply-chain-in-uk) accessed 11 April 2019.

EEA, 2018f, National climate change vulnerability and risk assessments in Europe, 2018, EEA Report No 1/2018, European Environment Agency (https://www.eea. europa.eu/publications/national-climate-changevulnerability-2018) accessed 10 April 2019.

EEA, 2018g, 'Number of countries that have adopted a climate change adaptation strategy/plan', European Environment Agency (https://www.eea.europa.eu/ airs/2018/environment-and-health/climate-changeadaptation-strategies) accessed 7 January 2019.

EEA, 2018h, 'Replacing overhead lines with underground cables in Finland', Climate-ADAPT case study (https://climate-adapt.eea.europa.eu/ metadata/case-studies/replacing-overhead-lines-withunderground-cables-in-finland) accessed 11 April 2019.

EEA, 2018i, *Sharing adaptation information across Europe*, EEA Report No 3/2018, European Environment Agency (https://www.eea.europa.eu/publications/ sharing-adaptation-information-across-europe) accessed 17 August 2018.

EEA, 2019a, 'Forest fires (CLIM 035)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-185-en) accessed 9 April 2019. EEA, 2019b, 'Global and European sea level (CSI 047)', European Environment Agency (https://www.eea. europa.eu/ds\_resolveuid/IND-193-en) accessed 9 April 2019.

EEA, 2019c, 'Global and European temperature (CSI 012)', European Environment Agency (https://www.eea. europa.eu/ds\_resolveuid/IND-4-en) accessed 9 April 2019.

EEA, 2019d, 'Heating and cooling degree days (CLIM 047)', European Environment Agency (https://www.eea. europa.eu/ds\_resolveuid/IND-348-en) accessed 9 April 2019.

EEA, 2019e, 'Heavy precipitation (CLIM 004)', European Environment Agency (https://www.eea.europa.eu/ ds\_resolveuid/IND-92-en) accessed 9 April 2019.

EEA, 2019f, 'Meteorological and hydrological droughts (CLIM 018)', European Environment Agency (https:// www.eea.europa.eu/ds\_resolveuid/IND-105-en) accessed 9 April 2019.

EEA/EC, 2019, 'Climate-ADAPT' (http://climate-adapt. eea.europa.eu/) accessed 10 April 2019.

EIB, 2015, *EIB climate strategy: mobilising finance for the transition to a low-carbon and climate-resilient economy*, European Investment Bank (http://www.eib.org/ attachments/strategies/eib\_climate\_strategy\_en.pdf) accessed 10 April 2019.

Eionet, 2019, 'Deliveries for National adaptation actions — Eionet Reporting Obligations Database' (https://rod. eionet.europa.eu/obligations/703/deliveries) accessed 8 April 2019.

Energy UK, 2015, *Climate change risks & adaptation responses for UK electricity generation: a sector overview* 2015, Energy UK, London (https://assets.publishing. service.gov.uk/government/uploads/system/uploads/ attachment\_data/file/478938/clim-adrep-energyuk-2015.pdf) accessed 6 January 2018.

ENSREG, 2012, *Stress tests performed on European nuclear power plants*, Peer review report, European Nuclear Safety Regulators Group (http://www.ensreg. eu/sites/default/files/EU%20Stress%20Test%20Peer%20 Review%20Final%20Report\_0.pdf) accessed 10 April 2019.

ENTSO-E, 2017, *Mid-term adequacy forecast 2017 edition*, European Network of Transmission System Operators, Brussels (https://docstore.entsoe.eu/Documents/ SDC%20documents/MAF/MAF\_2017\_report\_for\_ consultation.pdf) accessed 9 June 2018. ENTSO-E, 2018, *TYNDP 2018 Executive report connecting Europe: electricity, 2025 - 2030 - 2040, version for public consultation*, European Network of Transmission System Operators, Brussels (https://tyndp.entsoe. eu/Documents/TYNDP%20documents/TYNDP2018/ consultation/Main%20Report/TYNDP2018\_ Executive%20Report.pdf) accessed 9 June 2018.

ENTSOS, 2018, *TYNDP 2018 Scenario report*, ENTSO-G and ENTSO-E (https://docstore.entsoe.eu/Documents/ TYNDP%20documents/14475\_ENTSO\_ScenarioReport\_ Main.pdf) accessed 10 April 2019.

EP, 2012, *Current state and future challenges of Europe's waters*, Directorate General for Internal policies (http://www.europarl.europa.eu/RegData/etudes/etudes/ join/2012/475095/IPOL-ENVI\_ET(2012)475095\_EN.pdf) accessed 10 April 2019.

ESA, 2019, 'Nuclear observatory data' (http://ec.europa. eu/euratom/observatory\_data.html) accessed 13 March 2019.

EU, 2000, Directive 2000/60/EC of the European Parliament and of the Council of 23 October 2000 establishing a framework for Community action in the field of water policy (OJ L 327, 22.12.2000, pp. 1-73).

EU, 2007, Directive 2007/60/EC of the European Parliament and of the Council of 23 October 2007 on the assessment and management of flood risks (OJ L 288, 6.11.2007, pp. 27-34).

EU, 2008a, Council Directive 2008/114/EC of 8 December 2008 on the identification and designation of European critical infrastructures and the assessment of the need to improve their protection (OJ L 345, 23.12.2008, p. 75-82).

EU, 2008b, Regulation (EC) No 1099/2008 of the European Parliament and of the Council of 22 October 2008 on energy statistics (Text with EEA relevance) (OJ L 304, 14.11.2008, pp. 1-62).

EU, 2009, Directive 2009/125/EC of the European Parliament and of the Council of 21 October 2009 establishing a framework for the setting of ecodesign requirements for energy-related products (OJ L 285, 31.10.2009, pp. 10-35).

EU, 2012, Commission Regulation (EU) No 206/2012 of 6 March 2012 implementing Directive 2009/125/EC of the European Parliament and of the Council with regard to ecodesign requirements for air conditioners and comfort fans (OJ L 72, 10.3.2012, pp. 7–27).

EU, 2013a, Directive 2013/30/EU of the European Parliament and of the Council of 12 June 2013 on safety of offshore oil and gas operations and amending Directive 2004/35/EC (OJ L 178, 28.6.2013, pp. 66-106).

EU, 2013b, Regulation (EU) No 347/2013 of the European Parliament and of the Council of 17 April 2013 on guidelines for trans-European energy infrastructure and repealing Decision No 1364/2006/EC and amending Regulations (EC) No 713/2009, (EC) No 714/2009 and (EC) No 715/2009 (L 115/39).

EU, 2013c, Regulation (EU) No 525/2013 of the European Parliament and of the Council of 21 May 2013 on a mechanism for monitoring and reporting greenhouse gas emissions and for reporting other information at national and Union level relevant to climate change and repealing Decision No 280/2004/EC (OJ L 165, 18.6.2013, pp. 13–40).

EU, 2014a, Directive 2014/89/EU of the European Parliament and of the Council of 23 July 2014 establishing a framework for maritime spatial planning (OJ L 257, 28.08.2014, pp. 135-145).

EU, 2014b, Directive 2014/95/EU of the European Parliament and of the Council of 22 October 2014 amending Directive 2013/34/EU as regards disclosure of non-financial and diversity information by certain large undertakings and groups (OJ L 330, 15.11.2014, pp. 1-9).

EU, 2017, Regulation (EU) 2017/1369 of the European Parliament and of the Council of 4 July 2017 setting a framework for energy labelling (OJ L 198, 28.7.2017, pp. 1-23).

EU, 2018a, Directive (EU) 2018/2001 of the European Parliament and of the Council of 11 December 2018 on the promotion of the use of energy from renewable sources (OJ L 328, 21.12.2018, p. 82-209).

EU, 2018b, Directive (EU) 2018/2002 of the European Parliament and of the Council of 11 December 2018 amending Directive 2012/27/EU on energy efficiency (OJ L 328, 21.12.2018, pp. 210-230).

EU, 2018c, Proposal for a Regulation of the European Parliament and of the Council on risk-preparedness in the electricity sector and repealing Directive 2005/89/ EC (2016/0377(COD)).

EU, 2018d, Regulation 2018/1999 of the European Parliament and of the Council of 11 December 2018 on the Governance of the Energy Union and Climate Action (OJ L 328, 21.12.2018, pp. 1-77). EU TEGSF, 2019, *Report on climate-related disclosures*, European Commission, Brussels (https://ec.europa. eu/info/publications/190110-sustainable-finance-tegreport-climate-related-disclosures\_en) accessed 10 April 2019.

EUFIWACC, 2016, *EUFIWACC guidance: Integrating climate resilience into project development and implementation*, EBRD (http://www.jaspersnetwork.org/download/ attachments/21168520/4%20-NP%20Climate%20 Adaptation%20meeting%20-%20EUFIWACC%20 guidance.pdf?version=1&modificationDate=146790809 7000&api=v2) accessed 7 November 2018.

European Council, 2013, Conclusions of the European Council (Multiannual Financial Framework) (EUCO 37/13).

European Council, 2014, 2030 climate and energy policy framework (EUCO 169/14).

European Parliament, 2018, *Report on the proposal for a regulation of the European Parliament and of the Council on the internal market for electricity (recast) (COM(2016)0861 – C8-0492/2016 – 2016/0379(COD))* (http://www.europarl.europa.eu/sides/getDoc. do?pubRef=-//EP//NONSGML+REPORT+A8-2018-0042+0+DOC+PDF+V0//EN) accessed 20 September 2018.

Eurostat, 2018a, 'Annual freshwater abstraction by source and sector' (http://appsso.eurostat.ec.europa. eu/nui/show.do?dataset=env\_wat\_abs&lang=en) accessed 2 October 2018.

Eurostat, 2018b, 'Complete energy balances — annual data (nrg\_110a)' (http://appsso.eurostat.ec.europa.eu/ nui/show.do?dataset=nrg\_110a&lang=en) accessed 14 September 2018.

Eurostat, 2018c, *EU energy in figures: statistical pocketbook 2018*, Luxembourg, Publications Office of the European Union (https://publications.europa.eu/en/publication-detail/-/publication/99fc30eb-c06d-11e8-9893-01aa75ed71a1/language-en/format-PDF) accessed 6 November 2018.

Eurostat, 2018d, 'Population and population change statistics' (http://ec.europa.eu/eurostat/statisticsexplained/index.php/Population\_and\_population\_ change\_statistics#EU-28\_population\_continues\_to\_ grow) accessed 27 June 2018.

Eurostat, 2018e, 'Presentation of annual energy statistics in Eurobase: changes in datasets in node [nrg\_quanta]', Eurostat (https://ec.europa.eu/eurostat/ documents/10186/6246844/Eurobase-changes-energy. pdf) accessed 11 March 2019.

Eurostat, 2018f, 'Supply, transformation and consumption of electricity — annual data' (https://ec.europa.eu/eurostat/web/products-datasets/-/nrg\_105a) accessed 14 September 2018.

Eurostat, 2019a, 'Electricity price statistics — Statistics Explained' (https://ec.europa.eu/eurostat/statisticsexplained/index.php/Electricity\_price\_statistics) accessed 6 March 2019.

Eurostat, 2019b, 'Energy balances' (https://ec.europa. eu/eurostat/web/energy/data/database) accessed 4 October 2018.

Eurostat, 2019c, 'EU imports of energy products recent developments — Statistics Explained' (https:// ec.europa.eu/eurostat/statistics-explained/index. php?title=EU\_imports\_of\_energy\_products\_-\_recent\_ developments) accessed 13 March 2019.

Forster, D., et al., 2017, *Climate mainstreaming in the EU budget. Preparing for the next MFF: final report*, Publications Office of the European Union, Luxembourg (http://doi.org/10.2834/218038) accessed 9 April 2019.

Forzieri, G., et al., 2016, 'Multi-hazard assessment in Europe under climate change', *Climatic Change* 137(1-2), pp. 105-119 (DOI: 10.1007/s10584-016-1661-x).

Forzieri, G., et al., 2018, 'Escalating impacts of climate extremes on critical infrastructures in Europe', *Global Environmental Change* 48, pp. 97-107 (DOI: 10.1016/j. gloenvcha.2017.11.007).

Fraunhofer ISE, 2017, 'Harvesting the sun for power and produce — agrophotovoltaics increases the land use efficiency by over 60 percent' (https://www.ise. fraunhofer.de/en/press-media/press-releases/2017/ harvesting-the-sun-for-power-and-produceagrophotovoltaics-increases-the-land-use-efficiency-byover-60-percent.html) accessed 26 September 2018.

Fricko, O., et al., 2016, 'Energy sector water use implications of a 2 °C climate policy', *Environmental Research Letters* 11(3), p. 034011 (DOI: 10.1088/1748-9326/11/3/034011).

Frisvold, G. B. and Marquez, T., 2014, 'Water requirements for large-scale solar energy projects in the west', *Journal of Contemporary Water Research & Education* 151(1), pp. 106-116 (DOI: 10.1111/j.1936-704X.2013.03156.x).

Gafni, M., 2018, 'PG&E power line eyed in Camp Fire had collapsed during 2012 storm', *Mercury News*, 19 November 2018 (https://www.mercurynews. com/2018/11/19/pge-transmission-line-eyed-in-campfire-had-collapsed-during-2012-storm/) accessed 11 December 2018.

Gao, X., et al., 2018, 'Potential impacts of climate warming and increased summer heat stress on the electric grid: a case study for a large power transformer (LPT) in the Northeast United States', *Climatic Change* 147(1-2), pp. 107-118 (DOI: 10.1007/s10584-017-2114-x).

Gasparatos, A., et al., 2017, 'Renewable energy and biodiversity: Implications for transitioning to a Green Economy', *Renewable and Sustainable Energy Reviews* 70, pp. 161-184 (DOI: 10.1016/j.rser.2016.08.030).

Gaudard, L. and Romerio, F., 2014, 'The future of hydropower in Europe: interconnecting climate, markets and policies', *Environmental Science & Policy* 37, pp. 172-181 (DOI: 10.1016/j.envsci.2013.09.008).

Girardi, G., et al., 2015, Informe de Adaptación al Cambio Climático del Sector Energético Español. Análisis de la influencia del cambio climático en la oferta y la demanda de energía, Informe Tecnico, Instituto de Investigacion Tecnologica, Madrid (https://www.adaptecca.es/sites/ default/files/editor\_documentos/informe\_adaptacion\_ sector\_energ.pdf) accessed 10 April 2019.

Government of Austria, 2017, *The Austrian Strategy for Adapting to Climate Change: Part 2- Action Plan*, Federal Ministry of Sustainability and Tourism, Vienna (https:// www.bmnt.gv.at/umwelt/klimaschutz/klimapolitik\_ national/anpassungsstrategie/strategie-kontext.html) accessed 10 January 2018.

Government of Bulgaria, 2013, Analysis and evaluation of the risks and vulnerability of sectors in the Bulgarian economy from climate change, Ministry of Environment and Water, Sofia (https://www.moew.government.bg/ static/media/ups/tiny/file/Climate/obshta\_chast.pdf) accessed 19 September 2018.

Government of Croatia, 2017a, *Draft Climate Change Adaptation Strategy in the Republic of Croatia for the period to 2040 with a view to 2070*, Ministry of Environment and Energy, Zagreb, Croatia (http:// prilagodba-klimi.hr/wp-content/uploads/docs/Draft%20 CC%20Adaptation%20Strategy.pdf) accessed 20 September 2018.

Government of Croatia, 2017b, *Report on assessment of climate impacts and vulnerabilities in different sectors*, Ministry of Environmental Protection and Energy (MZOE), Zagreb, Croatia (http://prilagodba-klimi.hr/wpcontent/uploads/docs/Procjena-ranjivosti-na-klimatskepromjene.pdf) accessed 19 September 2018.

Government of Cyprus, 2016, *The Cyprus climate change risk assessment, evidence report*, Ministry of Agriculture, Rural Development and Environment (http://www. moa.gov.cy/moa/environment/environmentnew.nsf/2 76491E82F8428E1C22580C30034ABF2/\$file/Evidence-Report-v1\_final.pdf) accessed 19 September 2018.

Government of Cyprus, 2017, *Cyprus action plan for adaptation to climate change*, Ministry of Agriculture, Rural Development and Environment, Nicosia (http:// cypadapt.uest.gr/).

Government of Czech Republic, 2015, *Comprehensive study on impacts, vulnerability and risks sources connected to climate change in the Czech Republic*, Ministry of the Environment (https://www.mzp.cz/C1257458002F0DC7/ cz/studie\_dopadu\_zmena\_klimatu/\$FILE/OEOK-Komplexni\_studie\_dopady\_klima-20151201.pdf) accessed 19 September 2018.

Government of Czech Republic, 2017, National Action Plan on Adaptation to Climate Change in Czech Republic, Ministry of Environment (https://www.mzp.cz/ C1257458002F0DC7/cz/narodni\_akcni\_plan\_zmena\_ klimatu/\$FILE/OEOK-NAP\_cely\_20170127.pdf) accessed 19 September 2018.

Government of Denmark, 2008, *Danish adaptation to future climate* — *on the Danish strategy for adaptation to a changing climate*, Danish Energy Agency, Copenhagen (http://en.klimatilpasning.dk/media/5322/ klimatilpasningspjece\_uk\_web2008.pdf) accessed 6 January 2018.

Government of Denmark, 2012a, *How to manage cloudburst and rain water. Action plan for a climate-proof Denmark*, Danish Nature Agency, Copenhagen (https:// en.klimatilpasning.dk/media/590075/action\_plan.pdf) accessed 10 April 2019.

Government of Denmark, 2012b, *Mapping climate change* — *barriers and opportunities for action*, Danish Nature Agency (http://en.klimatilpasning.dk/ media/600858/130206\_mapping\_climate\_change\_final. pdf) accessed 19 September 2018.

Government of Estonia, 2017, *Climate Change Adaptation Development Plan until 2030*, Ministry of the Environment Estonia (https://www.envir.ee/sites/ default/files/national\_adaptation\_strategy.pdf) accessed 19 September 2018.
Government of Finland, 2013, *The adverse impacts of climate change and the vulnerability of sectors*, Ministry of Forestry and Agriculture (https://mmm.fi/ documents/1410837/1516663/Ilmastonmuutoksen\_ haitalliset\_vaikutukset\_ja\_toimialojen\_haavoittuvuus\_ raportti\_(final).pdf/7f1a2e21-a4cb-48e6-aff4d92dc770240a) accessed 19 September 2018.

Government of Finland, 2014, *Finland's National Climate Change Adaptation Plan 2022*, Ministry of Forestry and Agriculture (https://mmm.fi/ documents/1410837/1888935/MMM-%23193086v1-Finland\_s\_National\_climate\_Change\_Adaptation\_ Plan\_2022.pdf/c2bfec7b-ae73-4247-b666-26a3ed363f99/MMM-%23193086-v1-Finland\_s\_ National\_climate\_Change\_Adaptation\_Plan\_2022.pdf. pdf) accessed 19 September 2018.

Government of Flanders, 2016, *Progress report 2015* — *Flemish climate policy plan 2013-2020* (https://www. Ine.be/sites/default/files/atoms/files/VR%202016%20 1504%20MED.%20VORA2015%20-%203%20bijlage.pdf) accessed 20 September 2018.

Government of France, 2009, *Climate change: costs of impacts and lines of adaptation*, National Observatory for the Impacts of Global Warming (https://www.ecologique-solidaire.gouv.fr/sites/default/files/ ONERC\_rapport\_Climate%20change\_Costs%20of%20 impacts%20and%20lines%20of%20adaptation\_ENG. pdf) accessed 19 September 2018.

Government of France, 2011, *Plan National d'Adaptation au Changement Climatique*, Ministere de l'Écologie, du Développement durable et de l'Énergie (https://www.ecologique-solidaire.gouv.fr/sites/default/files/ONERC\_PNACC\_1\_complet.pdf) accessed 20 September 2018.

Government of Germany, 2008, *German strategy for adaptation to climate change* (http://www.bmu.de/fileadmin/bmu-import/files/english/pdf/application/pdf/das\_gesamt\_en\_bf.pdf) accessed 6 January 2018.

Government of Germany, 2011, Adaptation action plan of the German strategy for adaptation to climate change (http://www.bmub.bund.de/fileadmin/bmu-import/ files/pdfs/allgemein/application/pdf/aktionsplan\_ anpassung\_klimawandel\_en\_bf.pdf) accessed 19 September 2018.

Government of Germany, 2015a, *Fortschrittsbericht zur Deutschen Anpassungsstrategie an den Klimawandel*, Government of Germany, Berlin (https://www.bmu.de/ fileadmin/Daten\_BMU/Download\_PDF/Klimaschutz/ klimawandel\_das\_fortschrittsbericht\_bf.pdf) accessed 10 April 2019. Government of Germany, 2015b, *Germany's vulnerability to climate change: summary*, Federal Environment Agency, Dessau-Rosslau (https://www. umweltbundesamt.de/sites/default/files/medien/378/ publikationen/climate\_change\_24\_2015\_summary\_ vulnerabilitaet\_deutschlands\_gegenueber\_dem\_ klimawandel\_2.pdf) accessed 19 September 2018.

Government of Greece, 2016, *National strategy for adaptation to climate change*, Ministry of Environment and Energy, Athens (http://www.ypeka.gr/LinkClick.asp x?fileticket=crbjkilcLlA%3d&tabid=303&language=el-GR) accessed 10 April 2019.

Government of Hungary, 2017a, *Draft of the Second National Climate Change Strategy for 2017-2030*, Ministry of National Development (http://www. kormany.hu/download/f/6a/f0000/N%C3%895\_2\_ strat%C3%A9gia\_2017\_02\_27.pdf) accessed 20 September 2018.

Government of Hungary, 2017b, National Adaptation Strategy: Part of the Second National Climate Change Strategy (http://www.parlament.hu/ irom40/15783/15783.pdf) accessed 19 September 2018.

Government of Ireland, 2018a, *Climate change adaptation plan for the electricity and gas networks sector*, Department of Communications, Climate Action and Environment, Dublin (https://www.dccae.gov.ie/en-ie/ news-and-media/publications/Pages/Climate-Change-Adaptation-Plan-for-the-Electricity-and-Gas-Networkssector-0221-5002.aspx) accessed 10 April 2019.

Government of Ireland, 2018b, 'Sectoral adaptation planning', Department of Communications, Climate Action and Environment (https://www.dccae.gov.ie/enie/climate-action/topics/adapting-to-climate-change/ national-adaptation-framework/Pages/Sectoral.aspx) accessed 20 September 2018.

Government of Italy, 2014, *Report on the state of scientific knowledge on impacts, vulnerabilities and adaptation to climate change in Italy*, Ministry of the Environment and of the Protection of the Territory of the Sea (http://www.minambiente.it/sites/default/files/ archivio/allegati/clima/snacc\_2014\_rapporto\_stato\_ conoscenze.pdf) accessed 19 September 2018.

Government of Latvia, 2017, *Risk and vulnerability assessment and identification of adaptation measures in building and infrastructure planning*, Ministry of Environmental Protection and Regional Development (http://petijumi.mk.gov.lv/sites/default/files/title\_file/ petijums\_varam\_2016\_2017\_risk\_un\_ievain\_novert\_un\_ pielag\_pasak\_identif\_buvniec\_un\_infrastr\_joma.pdf) accessed 19 September 2018. Government of Liechtenstein, 2017, *Liechtenstein's Seventh National Communication*, Office of Environment, Vaduz, Liechtenstein (http://unfccc.int/files/national\_ reports/annex\_i\_natcom\_/application/pdf/8509731\_ liechtenstein-nc7-br3-1-nc7\_br3\_lie\_2018.pdf) accessed 20 September 2018.

Government of Liechtenstein, 2018, *Anpassungsstrategie an den Klimawandel in Liechtenstein*, Regierung des Fürstentums Liechtenstein, Vaduz, Liechtenstein (https://www.llv.li/files/au/ anpassungstrategieklimawandel-li.pdf) accessed 10 April 2019.

Government of Lithuania, 2015, *Studies, laying down the vulnerability of specific sectors to climate change, risk assessment, the most effective adaptation to climate change measures and evaluation criteria*, Ministry of the Environment Lithuania, Vilnius, Lithuania (https:// am.lrv.lt/uploads/am/documents/files/KLIMATO%20 KAITA/Studijos%2C%20metodin%C4%97%20 med%C5%BEiaga/Klimato%20kaita\_galutine%20 ataskaita\_2015\_08\_31.pdf) accessed 19 September 2018.

Government of Lithuania, 2018, Amendment to the interinstitutional action plan for the implementation of the objectives and tasks of the national climate change management strategy (2018 m. vasario 12 d. Nr. 147).

Government of Luxembourg, 2012, Adaptation to climate change — strategies for spatial planning in Luxembourg, Ministry of Long-Term Development and Infrastructure (https://amenagement-territoire. public.lu/content/dam/amenagement\_territoire/ fr/publications/documents/C-Change/CChange\_ conclusions.pdf) accessed 19 September 2018.

Government of Luxembourg, 2018, *Strategie und Aktionsplan für die Anpassung an den Klimawandel in Luxemburg 2018-2023*, Ministry of Sustainable Development and Infrastructure (https://environnement.public.lu/dam-assets/ actualites/2018/06/anpassung-an-den-klimawandel.pdf) accessed 12 April 2019.

Government of Malta, 2012, *National Climate Change Adaptation Strategy*, Ministry of Resources and Rural Affairs (https://msdec.gov.mt/en/Document%20 Repository/Malta%20Climate%20Change%20 Adaptation%20Strategy/National%20Adaptation%20 Strategy.pdf) accessed 20 September 2018.

Government of Norway, 2010, *Adapting to a changing climate: Norway's vulnerability and the need to adapt to the impacts of climate change*, Norwegian Ministry of the Environment, Oslo, Norway (https://www.regjeringen.

no/contentassets/00f70698362f4f889cbe30c75bca 4a48/pdfs/nou201020100010000en\_pdfs.pdf) accessed 29 June 2018.

Government of Norway, 2012, *Climate adaptation in Norway*, Environment Department Norway (https://www.regjeringen.no/contentassets/ e5e7872303544ae38bdbdc82aa0446d8/no/pdfs/ stm201220130033000dddpdfs.pdf) accessed 20 September 2018.

Government of Poland, 2013, *Polish National Strategy for Adaptation to Climate Change (NAS 2020)*, Ministry of Environment Poland, Warsaw (https://klimada.mos.gov. pl/wp-content/uploads/2014/12/ENG\_SPA2020\_final. pdf) accessed 20 September 2018.

Government of Portugal, 2012, *Relatório do Subgrupo Energia para uma Estratégia Nacional de Adaptação às Alterações Climáticas Medidas e Ações de Adaptação do SETOR ENERGÉTICO* (https://www.apambiente.pt/\_zdata/ Politicas/AlteracoesClimaticas/Adaptacao/ENAAC/ RelatDetalhados/Relat\_Setor\_ENAAC\_Energia.pdf) accessed 7 December 2018.

Government of Portugal, 2015, *Resolução do Conselho de Ministros n.º* 56/2015 (*Diário da República, 1.º série* — N.º 147).

Government of Portugal, 2016, *Estratégia Nacional de Adaptação às Alterações Climáticas - RELATÓRIO INTERCALAR #1 - 2015-2016* (https://www.apambiente. pt/\_zdata/Alteracoes\_Climaticas/Adaptacao/ RelatIntercalar1ENAAC2020/RelatorioIntercalarFinal. pdf) accessed 7 December 2018.

Government of Romania, 2016a, *Anexa 1 la Hotărârea Guvernului nr. 739/2016 - Strategia națională privind schimbările climatice și creșterea economică bazată pe emisii reduse* (http://www.mmediu.ro/app/webroot/ uploads/files/A1\_Strategia%20națională%20privind%20 schimbările%20climatice%20și%20creșterea%20 economică%20bazată%20pe%20emisii%20reduse.pdf) accessed 7 December 2018.

Government of Romania, 2016b, *Planul național de acțiune pentru implementarea Strategiei naționale privind schimbările climatice și creșterea economică bazată pe emisii reduse de carbon pentru perioada 2016-2020 (Anexa 2 la Hotărârea Guvernului nr. 739/2016).* 

Government of Slovakia, 2014, *National Adaptation Strategy of Slovakia*, Ministry of Environment Slovakia (http://www.minzp.sk/files/oblasti/politika-zmenyklimy/nas-sr-2014.pdf) accessed 20 September 2018. Government of Slovenia, 2010, *The environment is changing. Climate variability in Slovenia and its effects on the aquatic environment*, Environmental Agency of the Republic of Slovenia, Ljubljana (http://www.arso.gov. si/o%20agenciji/knji%C5%BEnica/publikacije/Okolje\_ se%20spreminja.pdf) accessed 19 September 2018.

Government of Slovenia, 2016, *Strategic framework for climate change adaptation* (http://www.mop.gov. si/fileadmin/mop.gov.si/pageuploads/podrocja/ podnebne\_spremembe/SOzP\_ang.pdf) accessed 20 September 2018.

Government of Spain, 2005, *A preliminary assessment* of the impacts in Spain due to the effects of climate change, Ministry of Environment, Madrid (https:// www.mapama.gob.es/es/cambio-climatico/temas/ impactos-vulnerabilidad-y-adaptacion/plan-nacionaladaptacion-cambio-climatico/evaluacion-preliminarde-los-impactos-en-espana-del-cambio-climatico/ eval\_impactos.aspx) accessed 19 September 2018.

Government of Spain, 2014, *Plan Nacional de Adaptacion al Cambio Climatico. Tercer Programa de Trabajo - 2014/2020, Ministerio de Agricultura, Alimentación y Medio Ambiente,* Madrid (https://www.mapama.gob.es/es/cambio-climatico/temas/impactos-vulnerabilidad-y-adaptacion/3er\_programa\_trabajo\_pnacc\_tcm30-70400. pdf) accessed 20 September 2018.

Government of Spain, 2018, *Plan Nacional de Adaptacion al Cambio Climatico. Cuarto informe de seguimento, Ministerio de Agricultura, Alimentación y Medio Ambiente,* Madrid (https://www.miteco.gob.es/es/cambio-climatico/temas/impactos-vulnerabilidad-y-adaptac ion/4informeseguimientopnacc\_tcm30-485659.pdf) accessed 20 September 2018.

Government of Sweden, 2017, Environmental assessments (Regeringens proposition 2016/17: 200).

Government of Sweden, 2019, '*På gång från myndigheterna 2018*', *Klimatanpassningsportalen* (http://www.klimatanpassning.se/roller-ochansvar/kommande-underlag/pa-gang-franmyndigheterna-2018-1.134104) accessed 12 April 2019.

Government of Switzerland, 2014, *Anpassung an den Klimawandel in der Schweiz. Aktionsplan 2014-2019*, Federal Office for the Environment (https://www.bafu. admin.ch/bafu/de/home/themen/klima/publikationen-studien/publikationen/anpassung-klimawandel-schweiz-2014.html) accessed 1 June 2016.

Government of the Netherlands, 2016, *Adapting with ambition: national climate adaptation strategy 2016*, Ministry of the Environment Netherlands (https://ruimtelijkeadaptatie.nl/publish/ pages/125102/2016\_12\_02\_nas\_netherlands\_4.pdf) accessed 20 September 2018.

Government of the Netherlands, 2018, *Implementation Programme 2018 – 2019: National Climate Adaptation Strategy (NAS) of the Netherlands*, Ministry of Infrastructure and Water Management (https:// ruimtelijkeadaptatie.nl/publish/pages/125102/nas\_nl\_ implementation\_progr\_2018-19\_eng.pdf) accessed 20 September 2018.

Government of the UK, 2012, Adapting to climate change: helping key sectors to adapt to climate change, Department for Environment Food & Rural Affairs (http://webarchive.nationalarchives.gov. uk/20130124040520/http://www.defra.gov.uk/ publications/2012/03/27/pb13740climate-changesummary/) accessed 10 April 2019.

Government of the UK, 2017, 'Climate change adaptation reporting: second round reports', GOV.UK (https://www.gov.uk/government/collections/climatechange-adaptation-reporting-second-round-reports) accessed 4 March 2019.

Government of the UK, 2018, *The National Adaptation Programme and the third strategy for climate adaptation reporting making the country resilient to a changing climate* (https://assets.publishing.service.gov.uk/ government/uploads/system/uploads/attachment\_ data/file/727252/national-adaptation-programme-2018. pdf) accessed 12 July 2018.

Government of Turkey, 2011, *Turkey's National Climate Change Adaptation Strategy and Action Plan*, Ministry of Environment and Urbanisation (http://www.dsi.gov.tr/ docs/iklim-degisikligi/turkeys-national-climate-changeadaptation-strategy-and-action-plan.pdf?sfvrsn=2) accessed 20 October 2018.

Government of Turkey, 2016, *Climate change impacts on water resources* — *sectoral vulnerability analysis in three river basins: Meric-Ergene, B. Menderes, Ceyhan*, Ministry of Forestry and Water Affairs (http://iklim.ormansu. gov.tr/ckfinder/userfiles/files/lklim\_NihaiRapor.pdf) accessed 19 September 2018.

Groenemeijer, P., 2015, *Past cases of extreme weather impact on critical infrastructure in Europe*, RAIN Deliverable No D 2.2 version 2.0 final, RAIN project (http://rain-project.eu/wp-content/uploads/2015/11/ D2.2-Past-Cases-final.compressed.pdf) accessed 15 August 2018.

Growitsch, C., et al., 2015, 'The costs of power interruptions in Germany: a regional and sectoral

analysis', *German Economic Review* 16(3), pp. 307-323 (DOI: 10.1111/geer.12054).

Gündüz, N., et al., 2017, 'Impacts of natural disasters on Swedish electric power policy: a case study', *Sustainability* 9(2), p. 230 (DOI: 10.3390/su9020230).

Haasnoot, M., et al., 2013, 'Dynamic adaptive policy pathways: a method for crafting robust decisions for a deeply uncertain world', *Global Environmental Change* 23(2), pp. 485-498 (DOI: 10.1016/j. gloenvcha.2012.12.006).

Heat Roadmap, 2019, *The legacy of Heat Roadmap Europe 4* (https://heatroadmap.eu/wp-content/ uploads/2019/02/HRE\_Final-Brochure\_web.pdf) accessed 10 April 2019.

Hjort, J., et al., 2018, 'Degrading permafrost puts Arctic infrastructure at risk by mid-century', *Nature Communications* 9(1) (DOI: 10.1038/s41467-018-07557-4).

Huang, J., et al., 2017, 'Recently amplified arctic warming has contributed to a continual global warming trend', *Nature Climate Change* 7(12), pp. 875-879 (DOI: 10.1038/s41558-017-0009-5).

Huneke, F., et al., 2017, *Kalte Dunkelflaute. Robustheit des Stromsystems bei Extremwetter*, Energy Brainpool (www. energybrainpool.com/fileadmin/download/Studien/ Studie\_2017-06-26\_GPE\_Studie\_Kalte-Dunkelflaute\_ Energy-Brainpool.pdf) accessed 10 April 2019.

IEA, 2013, *Redrawing the energy-climate map*, International Energy Agency, Paris (https://www.iea.org/ publications/freepublications/publication/WEO\_Special\_ Report\_2013\_Redrawing\_the\_Energy\_Climate\_Map.pdf) accessed 10 April 2019.

IEA, 2014, *4th Forum on the Climate-Energy Security Nexus: water and energy*, International Energy Agency, Paris (https://www.iea.org/media/workshops/2014/ nexus/Nexus4WorkshopReportIEAWBCSDJune2014. pdf) accessed 10 April 2019.

IEA, 2015, *Making the energy sector more resilient to climate change*, International Energy Agency, Paris (https://www.iea.org/publications/freepublications/ publication/COP21\_Resilience\_Brochure.pdf) accessed 10 April 2019.

IEA, 2016a, 5th Forum on the Climate-Energy Security Nexus: policies and practices to enhance energy sector resilience, International Energy Agency, Paris (http:// www.iea.org/media/workshops/2014/5thnexusforum/5 thNexusForumWorkshopSummary.pdf) accessed 10 April 2019.

IEA, 2016b, 'Enhancing energy sector resilience to climate change: government action and mobilising investment', in: *Energy, Climate Change & Environment — 2016 Insights*, International Energy Agency, Paris, pp. 85-98.

IEA, 2016c, *Water Energy Nexus. Excerpt from the World Energy Outlook 2016*, International Energy Agency, Paris (https://www.iea.org/ publications/freepublications/publication/ WorldEnergyOutlook2016ExcerptWaterEnergyNexus. pdf) accessed 10 April 2019.

IEA, 2017, Energy technology perspectives 2017: catalysing energy technology transformations, International Energy Agency, Paris (https://www.iea.org/publications/ freepublications/publication/EnergyTechnology Perspectives2017ExecutiveSummaryEnglishversion.pdf) accessed 20 September 2018.

IEA, 2018a, *The Future of Cooling. Opportunities for energy-efficient air conditioning*, International Energy Agency, Paris (http://www.iea.org/publications/ freepublications/publication/The\_Future\_of\_Cooling. pdf).

IEA, 2018b, *World Energy Outlook 2018*, International Energy Agency, Paris (https://www.iea.org/weo2018/) accessed 23 November 2018.

IHA, 2018, *Hydropower status report 2018*, International Hydropower Association, London (https://www. hydropower.org/publications/2018-hydropower-statusreport) accessed 10 April 2019.

INEA, 2018, 'MinWaterCSP — Minimized water consumption in CSP plants' (https://ec.europa.eu/inea/ en/horizon-2020/projects/H2020-Energy/Concentrated-Solar-Power/MinWaterCSP) accessed 1 June 2018.

International Hydropower Association, 2019, 'Sediment management hub' (https://www.hydropower.org/ sediment-management) accessed 27 February 2019.

IPCC, 2013, *Climate change 2013: the physical science basis. Working Group I contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change,* Cambridge University Press, Cambridge.

IPCC, 2014, *Climate change 2014: synthesis report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*, Cambridge University Press, Cambridge. IPCC, 2018, Global Warming of 1.5 °C: An IPCC Special Report on the impacts of global warming of 1.5 °C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty, Intergovernmental Panel on Climate Change, Geneva, Switzerland (http://report.ipcc.ch/sr15/pdf/sr15\_spm\_ final.pdf) accessed 10 November 2018.

IRENA, 2015, *Renewable energy in the water, energy & food nexus*, The International Renewable Energy Agency, Abu Dhabi, United Arab Emirates (http://www.irena.org/-/media/Files/IRENA/Agency/Publication/2015/ IRENA\_Water\_Energy\_Food\_Nexus\_2015.pdf) accessed 21 September 2018.

IRENA, 2018a, *Global landscape of renewable energy finance 2018*, IRENA (http://www.irena.org/-/media/ Files/IRENA/Agency/Publication/2018/Jan/IRENA\_ Global\_landscape\_RE\_finance\_2018.pdf) accessed 7 November 2018.

IRENA, 2018b, *Renewable energy prospects for the European Union*, International Renewable Energy Agency (https://www.irena.org/-/media/Files/IRENA/ Agency/Publication/2018/Feb/IRENA\_REmap\_EU\_2018. pdf) accessed 10 April 2019.

IRENA, 2018c, *Renewable power generation costs in 2017*, International Renewable Energy Agency, Abu Dhabi (https://www.irena.org/-/media/Files/IRENA/Agency/ Publication/2018/Jan/IRENA\_2017\_Power\_Costs\_2018. pdf) accessed 10 April 2019.

IRENA, 2019, Innovation landscape for a renewable-powered future: solutions to integrate variable renewables, International Renewable Energy Agency, Abu Dhabi (https://www.irena.org/publications/2019/ Feb/Innovation-landscape-for-a-renewable-powered-future) accessed 10 April 2019.

Jacob, D., et al., 2013, 'EURO-CORDEX: new high-resolution climate change projections for European impact research', *Regional Environmental Change* 14(2), pp. 563-578 (DOI: 10.1007/s10113-013-0499-2).

Jerez, S., et al., 2015, 'The impact of climate change on photovoltaic power generation in Europe', *Nature Communications* 6(1) (DOI: 10.1038/ncomms10014).

Johnson, C. W., 2008, 'Understanding failures in international safety-critical infrastructures: a comparison of European and North American power failures', conference paper presented at: 26th International Conference on Systems Safety, Vancouver, Canada, 25-29 August, 2008.

JPI Climate, 2017, 'Climate services at work. Projects exchange and networking lab', European Commission (https://ec.europa.eu/easme/sites/easme-site/files/ booklet-climateservices-networking\_event\_2017.pdf) accessed 10 April 2019.

Jude, S. R., et al., 2017, 'Delivering organisational adaptation through legislative mechanisms: evidence from the Adaptation Reporting Power (Climate Change Act 2008)', *Science of the Total Environment* 574, pp. 858-871 (DOI: 10.1016/j.scitotenv.2016.09.104).

Kaldellis, J. K., et al., 2014, 'Temperature and wind speed impact on the efficiency of PV installations. Experience obtained from outdoor measurements in Greece', *Renewable Energy* 66, pp. 612-624 (DOI: 10.1016/j.renene.2013.12.041).

Karagiannis, G. M., et al., 2017, *Power grid recovery after natural hazard impact*, JRC Science for Policy Report No EUR 28844 EN, Publications Office of the European Union, Luxembourg (http://doi.org/10.2760/87402).

Keramidas, K., et al., 2018, *Global energy and climate outlook 2018: sectoral mitigation options towards a low-emissions economy*, JRC Science for Policy Report No EUR 29462 EN, Publications Office of the European Union, Luxembourg (http://doi.org/10.2760/67475).

KLIMADA, 2013, 'Impacts of climate change on the energy sector' (http://klimada.mos.gov.pl/ en/2013/04/15/energy/) accessed 19 September 2018.

Koch, W., 2015, '3 reasons why Shell halted drilling in the Arctic', National Geographic News (https://news. nationalgeographic.com/energy/2015/09/150928-3reasons-shell-halted-drilling-in-the-arctic/) accessed 26 October 2018.

Köllner, P., et al., 2017, *Klimabedingte Risiken und Chancen. Eine schweizweite Synthese*, Bundesamt für Umwelt (BAFU) (https://www.bafu.admin.ch/bafu/ de/home/themen/klima/publikationen-studien/ publikationen/klimabedingte-risiken-und-chancen.html) accessed 10 April 2019.

Kovats, R. S., et al., 2014, 'Europe', in: Barros, V. R. et al. (eds), *Climate change 2014: impacts, adaptation, and vulnerability. Part B: Regional aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*, Cambridge University Press, Cambridge; New York, pp. 1267-1326. Küfeoğlu, S., 2015, *Economic impacts of electric power outages and evaluation of customer interruption costs*, Aalto University, Helsinki, Finland.

Lahtvee, V., et al., 2015a, *Estonian Climate Adaptation Strategy for Infrastructure and Energy*, Final Report, SEI Tallinn, Tallinn (http://kliima.seit.ee/files/ENFRA\_A\_ Uuringuaruanne\_01-04-2016.pdf) accessed 19 September 2018.

Lahtvee, V., et al., 2015b, *National Strategy for Climate Change in the Infrastructure and the Energy Sector 2017–2030*, SEI Tallinn, Tallinn (http://klima.seit.ee/files/ENFRA\_B\_Strateegiadokument\_01-04-2016.pdf) accessed 19 September 2018.

Leck, H., et al., 2015, 'Tracing the water-energy-food nexus: description, theory and practice', *Geography Compass* 9(8), pp. 445-460 (DOI: 10.1111/gec3.12222).

Lehane, B., 2018, 'Traders rush to move diesel up Rhine as water level ebbs', *Bloomberg.com*, 2018 (https://www. bloomberg.com/news/articles/2018-06-29/traders-rushto-move-diesel-up-rhine-as-river-s-water-level-ebbs) accessed 24 September 2018.

Lewis, M. J., et al., 2017, 'Influence of storm surge on tidal range energy', *Energy* 122, pp. 25-36 (DOI: 10.1016/j.energy.2017.01.068).

Ligtvoet, W., et al., 2015, *Adaptation to climate change in the Netherlands* — *studying related risks and opportunities*, PBL Publication No 1632, PBL Netherlands Environmental Assessment Agency, The Hague (http://www.pbl.nl/sites/default/files/ cms/publicaties/PBL-2015-Adaptation-to-climagechange-1632.pdf) accessed 3 December 2015.

Lise, W. and van der Laan, J., 2015, 'Investment needs for climate change adaptation measures of electricity power plants in the EU', *Energy for Sustainable Development* 28, pp. 10-20 (DOI: 10.1016/j. esd.2015.06.003).

Llasat, M. C., et al., 2014, 'The snow storm of 8 March 2010 in Catalonia (Spain): a paradigmatic wet-snow event with a high societal impact', *Natural Hazards and Earth System Sciences* 14(2), pp. 427-441 (DOI: 10.5194/ nhess-14-427-2014).

Macknick, J., et al., 2012, 'Operational water consumption and withdrawal factors for electricity generating technologies: a review of existing literature', *Environmental Research Letters* 7(4), p. 045802 (DOI: 10.1088/1748-9326/7/4/045802). Magnan, A. K., et al., 2016, 'Addressing the risk of maladaptation to climate change', *Wiley Interdisciplinary Reviews: Climate Change* 7(5), pp. 646-665 (DOI: 10.1002/wcc.409).

Majone, B., et al., 2016, 'Impact of climate change and water use policies on hydropower potential in the south-eastern Alpine region', *Science of the Total Environment* 543, pp. 965-980 (DOI: 10.1016/j. scitotenv.2015.05.009).

Malta Resources Authority, 2010, 'Sectoral' impacts of climate change in Malta' (http://mra.org.mt/climate-change/sectoral-impacts-of-climate-change-in-malta/) accessed 7 January 2019.

Mima, S. and Criqui, P., 2015, 'The costs of climate change for the European energy system, an assessment with the POLES model', *Environmental Modeling & Assessment* 20(4), pp. 303-319 (DOI: 10.1007/s10666-015-9449-3).

Mindas, J., et al., 2011, *Consequences of climate change and possible adaptation measures in individual sectors*, Ecological & Forestry Research Agency, Zvolen/Bratislava (http://www.shmu.sk/File/projekty/ Zaverecna%20Sprava%20projektu%20Klim.%20 zmena%20a%20Adaptacie%202012.pdf) accessed 19 September 2018.

Moemken, J., et al., 2018, 'Future changes of wind speed and wind energy potentials in EURO-CORDEX ensemble simulations', *Journal of Geophysical Research: Atmospheres* 123(12), pp. 6373-6389 (DOI: 10.1029/2018JD028473).

Mohseni, O. and Stefan, H. G., 1999, 'Stream temperature/air temperature relationship: a physical interpretation', *Journal of Hydrology* 218(3-4), pp. 128-141 (DOI: 10.1016/S0022-1694(99)00034-7).

Moser, S. C. and Ekstrom, J. A., 2010, 'A framework to diagnose barriers to climate change adaptation', *Proceedings of the National Academy of Sciences* 107(51), pp. 22026-22031 (DOI: 10.1073/pnas.1007887107).

Murrant, D., et al., 2015, 'The water-energy nexus: future water resource availability and its implications on UK thermal power generation', *Water and Environment Journal* 29(3), pp. 307-319 (DOI: 10.1111/wej.12126).

National Climate Commission, 2017a, *Belgisch Nationaal Adaptatieplan* (http://www.klimaat.be/ files/6714/9880/5758/NAP\_NL.pdf) accessed 19 September 2018. National Climate Commission, 2017b, *Belgium's Seventh National Communication and Third Biennial Report on Climate Change Under the United Nations Framework Convention on Climate Change*, National Climate Commission, Brussels, Rotterdam (http://unfccc.int/ files/national\_reports/annex\_i\_natcom\_/application/ pdf/7319685\_belgium-nc7-br3-1-nc7\_en\_lr.pdf) accessed 20 September 2018.

National Grid, 2016, *Climate Change Adaptation, Second Round Adaptation Response*, National Grid, Warwick, UK (https://assets.publishing.service.gov.uk/government/ uploads/system/uploads/attachment\_data/file/566151/ climate-adrep-national-grid.pdf) accessed 9 June 2018.

Noble, I. R., et al., 2014, 'Adaptation needs and options', in: Barros, V. R. et al. (eds), *Climate change 2014: impacts, adaptation, and vulnerability. Part B: Regional aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*, Cambridge University Press, Cambridge; New York, pp. 833-868.

Noordelijke Innovation Board, 2018, Green hydrogen economy of the Northern Netherlands (https:// www.waterstofnet.eu/\_asset/\_public/powertogas/ Conference/8-Ad-Van-Wijk\_TUDelft\_Green-hydrogeneconomy-in-the-Northern-Netherlands.pdf) accessed 26 September 2018.

Nord Stream 2, 2018, 'Nord Stream 2' (https://www. nord-stream2.com/) accessed 14 September 2018.

North Sea Energy, 2019, 'North Sea energy' (https:// www.north-sea-energy.eu/index.html) accessed 16 January 2019.

OECD, 2015, *Climate change risks and adaptation*, OECD Publishing, Paris.

OECD, 2018, *Climate-resilient Infrastructure*, OECD Environment Policy Paper No 14, OECD (http:// www.oecd.org/environment/cc/Climate-resilientinfrastructure-policy-perspectives.pdf) accessed 10 April 2019.

OECD/World Bank/UN Environment, 2018, *Financing climate futures: rethinking infrastructure*, OECD Publishing, Paris.

Ouranos and Natural Resources Canada, 2018, 'Energy adaptation map' (https://energyadaptation.ouranos.ca/placemarks/maps) accessed 31 July 2018.

Paroussos, L., et al., 2015, 'Assessment of carbon leakage through the industry channel: the EU

perspective', *Technological Forecasting and Social Change* 90, pp. 204-219 (DOI: 10.1016/j.techfore.2014.02.011).

Paskal, C., 2009, *The vulnerability of energy infrastructure to environmental change*, Chatham House Briefing Paper No EERG BP 2009/01, Chatham House, London, UK (https://www.chathamhouse.org/sites/default/ files/public/Research/Energy%2C%20Environment%20 and%20Development/bp0409energy.pdf) accessed 11 January 2018.

Pickering, M. D., et al., 2017, 'The impact of future sealevel rise on the global tides', *Continental Shelf Research* 142, pp. 50-68 (DOI: 10.1016/j.csr.2017.02.004).

Pittock, J., et al., 2015, *Climate, energy, and water: managing trade-offs, seizing opportunities*, Cambridge University Press, Cambridge (UK).

PRI/UNEP FI/UN Global Compact, 2016, French energy transition law. Global investor briefing, Principles for Responsible Investment Initiative, UNEP Finance Initiative and UN Global Compact (https://www.unepfi.org/fileadmin/documents/PRI-FrenchEnergyTransitionLaw.pdf) accessed 10 April 2019.

Quevedo, E., et al., 2013, 'Multi-use offshore platform configurations in the scope of the FP7 TROPOS Project', conference paper presented at: 2013 MTS/IEEE OCEANS, Bergen, June 2013.

Rademaekers, K., et al., 2011, Investment needs for future adaptation measures in EU nuclear power plants and other electricity generation technologies due to effects of climate change, EUR 24769, European Commission, Brussels (https://ec.europa.eu/energy/sites/ener/files/ documents/2011\_03\_eur24769-en.pdf) accessed 10 April 2019.

Rademaekers, K., et al., 2018, *Evaluation of the TEN-E Regulation and assessing the impacts of alternative policy scenarios final report*, European Commission, Brussels (http://doi.org/10.2833/736216) accessed 9 April 2019.

Ravestein, P., et al., 2018, 'Vulnerability of European intermittent renewable energy supply to climate change and climate variability', *Renewable and Sustainable Energy Reviews* 97, pp. 497-508 (DOI: 10.1016/j.rser.2018.08.057).

Reckien, D., et al., 2018, 'How are cities planning to respond to climate change? Assessment of local climate plans from 885 cities in the EU-28', *Journal of Cleaner Production* 191, pp. 207-219 (DOI: 10.1016/j. jclepro.2018.03.220). Reuters, 2018, 'France's EDF halts four nuclear reactors due to heatwave', *Reuters*, 4 August 2018 (https://www. reuters.com/article/us-france-nuclearpower-weather/ frances-edf-halts-four-nuclear-reactors-due-toheatwave-idUSKBN1KP0ES) accessed 24 September 2018.

Reyers, M., et al., 2016, 'Future changes of wind energy potentials over Europe in a large CMIP5 multi-model ensemble', *International Journal of Climatology* 36(2), pp. 783-796 (DOI: 10.1002/joc.4382).

Rojas, R., et al., 2013, 'Climate change and river floods in the European Union: socio-economic consequences and the costs and benefits of adaptation', *Global Environmental Change* 23(6), pp. 1737-1751 (DOI: 10.1016/j.gloenvcha.2013.08.006).

Rübbelke, D. and Vögele, S., 2011, 'Impacts of climate change on European critical infrastructures: the case of the power sector', *Environmental Science & Policy* 14(1), pp. 53-63 (DOI: 16/j.envsci.2010.10.007).

Russel, D., et al., 2018, 'Analyzing the policy framework for climate change adaptation', in: Sanderson, H. et al. (eds.), *Adapting to climate change in Europe*, Elsevier, pp. 273-313.

Russo, S., et al., 2014, 'Magnitude of extreme heat waves in present climate and their projection in a warming world', *Journal of Geophysical Research: Atmospheres* 119(22), pp. 12500-12512 (DOI: 10.1002/2014JD022098).

Sajn, N., 2016, *Electricity 'prosumers'*, No PE 593.518, European Parliament Research Service (http:// www.europarl.europa.eu/RegData/etudes/ BRIE/2016/593518/EPRS\_BRI(2016)593518\_EN.pdf) accessed 10 April 2019.

Sathaye, J. A., et al., 2013, 'Estimating impacts of warming temperatures on California's electricity system', *Global Environmental Change* 23(2), pp. 499-511 (DOI: 10.1016/j.gloenvcha.2012.12.005).

Schaber, K., et al., 2012, 'Parametric study of variable renewable energy integration in Europe: advantages and costs of transmission grid extensions', *Energy Policy* 42, pp. 498-508 (DOI: 10.1016/j.enpol.2011.12.016).

Schaeffer, R., et al., 2012, 'Energy sector vulnerability to climate change: a review', *Energy* 38(1), pp. 1-12 (DOI: 10.1016/j.energy.2011.11.056).

Schaps, K., 2015, 'Royal Dutch Shell pulls plug on Arctic exploration', *Reuters*, 28 September 2015 (https://www.reuters.com/article/us-shell-alaska/ royal-dutch-shell-pulls-plug-on-arctic-explorationidUSKCN0RS0EX20150928) accessed 26 October 2018.

Scheraga, J. D. and Grambsch, A. E., 1998, 'Risks, opportunities, and adaptation to climate change', *Climate Research* 10, pp. 85-95.

Schneider, T., 2014, 'Responsibility for private sector adaptation to climate change', *Ecology and Society* 19(2) (DOI: 10.5751/ES-06282-190208).

Scott Hosking, J., et al., 2018, 'Changes in European wind energy generation potential within a 1.5 °C warmer world', *Environmental Research Letters* 13(5), p. 054032 (DOI: 10.1088/1748-9326/aabf78).

SFEN, 2015, 'Adapting nuclear power plants to climate change', Société Française d' Energie Nucléaire (http:// www.sfen.org/rgn/adapter-centrales-nucleaires-changement-climatique) accessed 31 July 2018.

Shivakumar, A., et al., 2017, 'Valuing blackouts and lost leisure: estimating electricity interruption costs for households across the European Union', *Energy Research & Social Science* 34, pp. 39-48 (DOI: 10.1016/j. erss.2017.05.010).

Sjökvist, E., et al., 2013, *Framtida perioder med hög risk för skogsbrand — analyser av klimatscenarier*, Publ. nr MSB535, Myndigheten för samhällsskydd och beredskap, Karlstad, Sweden (https://www.msb.se/RibData/Filer/pdf/26595.pdf) accessed 10 April 2019.

Smid, M., et al., 2019, 'Ranking European capitals by exposure to heat waves and cold waves', *Urban Climate* 27, pp. 388-402 (DOI: 10.1016/j.uclim.2018.12.010).

Spinoni, J., et al., 2015, 'European degree-day climatologies and trends for the period 1951–2011', *International Journal of Climatology* 35(1), pp. 25-36 (DOI: 10.1002/joc.3959).

Spinoni, J., et al., 2018, 'Changes of heating and cooling degree-days in Europe from 1981 to 2100', *International Journal of Climatology* 38, pp. e191-e208 (DOI: 10.1002/joc.5362).

SSEPD, 2015, *Climate change adaptation report: second round*, ENA Engineering Report, Scottish and Southern Energy Power Distribution (https://assets.publishing. service.gov.uk/government/uploads/system/uploads/

attachment\_data/file/478927/clim-adrep-sse-powerdistribution-2015.pdf) accessed 6 January 2018.

Steinbach, J., et al., 2017, *Policy recommendations to decarbonise European heating and cooling systems*, progRESsHEAT consortium (https://ec.europa.eu/research/participants/documents/downloadPublic?documentIds=080166e5b75f5599&appId=PPGMS) accessed 25 September 2018.

Street, R., et al., 2016, *UK Climate change risk* assessment 2017: evidence report — Chapter 8: Crosscutting issues (https://www.theccc.org.uk/wp-content/ uploads/2016/07/UK-CCRA-2017-Chapter-8-Crosscutting-issues.pdf) accessed 3 March 2019.

Street, R. B., et al., 2017, *Understanding the value of the Adaptation Reporting Power process to the reporting organisations involved*, UK Climate Impacts Programme and University of Oxford (https://www.arcc-network. org.uk/wp-content/pdfs/ARP-report-for-Defra-04-2017. pdf) accessed 9 January 2018.

Stuart, C., 2017, *Energy and climate change adaptation in developing countries*, European Union Energy Initiative Partnership Dialogue Facility (EUEI PDF) (http://www.euei-pdf.org/sites/default/files/field\_publication\_file/euei\_pdf\_2017\_energy\_and\_climate\_change\_adaptation\_in\_developing\_countries.pdf) accessed 7 November 2018.

Technum, 2013, Explanation of the federal contribution to a coherent policy on climate adaptation, Federale Overheidsdienst Volksgezondheid, Veiligheid van de Voedselketen en Leefmilieu (http://www. climat.be/files/2013/8253/2115/Federale\_bijdrage\_ adaptatiebeleid\_Eindrapport\_juli\_2013.pdf) accessed 10 April 2019.

Tobin, I., et al., 2014, 'Assessing climate change impacts on European wind energy from ENSEMBLES highresolution climate projections', *Climatic Change* 128(1-2), pp. 99-112 (DOI: 10.1007/s10584-014-1291-0).

Tobin, I., et al., 2016, 'Climate change impacts on the power generation potential of a European mid-century wind farms scenario', *Environmental Research Letters* 11(3), p. 034013 (DOI: 10.1088/1748-9326/11/3/034013).

Tobin, I., et al., 2018, 'Vulnerabilities and resilience of European power generation to 1.5 °C, 2 °C and 3 °C warming', *Environmental Research Letters* 13(4), p. 044024 (DOI: 10.1088/1748-9326/aab211).

Trainor, A. M., et al., 2016, 'Energy sprawl is the largest driver of land use change in United States' Baldwin,

R. F. (ed.), *PLOS ONE* 11(9), p. e0162269 (DOI: 10.1371/ journal.pone.0162269).

Troccoli, A., et al., eds., 2014, *Weather matters for energy*, Springer, New York, NY.

Troccoli, A., et al., 2018, 'Forging a dialogue between the energy industry and the meteorological community', in: Troccoli, A. (ed.), *Weather & climate services for the energy industry*, Palgrave Macmillan, Cham, pp. 65-83.

Troccoli, A., ed., 2018, *Weather & climate services for the energy industry*, Palgrave Macmillan, Cham.

Tröltzsch, J., et al., 2018, *Knowledge synthesis and gap analysis on climate impact analysis, economic costs and scenarios*, COACHH Deliverable D1.2, COACCH project (https://www.coacch.eu/wp-content/uploads/2019/02/ D1.2\_Knowledge\_synthesis\_and\_gap\_analysis\_report. pdf) accessed 10 April 2019.

Tsang, A., 2019, 'PG&E to file for bankruptcy over deadly California fires', *New York Times*, 14 January 2019, New York, NY (https://www.nytimes.com/2019/01/14/ business/energy-environment/pge-bankruptcycalifornia.html) accessed 14 January 2019.

Tuck, G., et al., 2006, 'The potential distribution of bioenergy crops in Europe under present and future climate', *Biomass and Bioenergy* 30(3), pp. 183-197 (DOI: 10.1016/j.biombioe.2005.11.019).

UN, 1992, United Nations Framework Convention on Climate Change.

UN, 2015, Transforming our world: the 2030 agenda for sustainable development — resolution adopted by the General Assembly on 25 September 2015 (A/RES/70/1).

UNCCD/IRENA, 2017, *Energy and land use*, United Nations Convention to Combat Desertification, International Renewable Energy Agency (https:// knowledge.unccd.int/sites/default/files/2018-06/2.%20 Fritsche%2Bet%2Bal%2B%282017%29%2BEnergy%2B and%2BLand%2BUse%2B-%2BGLO%2Bpaper-corr.pdf) accessed 15 August 2018.

UNEP, 2016, Green energy choices: the benefits, risks and trade-offs of low-carbon technologies for electricity production, United Nations Environment Programme, Geneva.

UNEP, 2017, *Green technology choices: the environmental and resource implications of low-carbon technologies*, United Nations Environment Programme, Geneva.

UNEP, 2018, *Emissions gap report 2018*, United Nations Environment Programme (https://www.unenvironment. org/resources/emissions-gap-report-2018).

UNFCCC, 2015, *Paris Agreement*, Decision 1/CP.21, United Nations Framework Convention on Climate Change (http://unfccc.int/paris\_agreement/items/9485. php) accessed 1 June 2018.

UNFCCC, 2017, Submitted National Communications from Annex I Parties, United Nations Framework Convention on Climate Change (https://unfccc.int/process/ transparency-and-reporting/reporting-and-reviewunder-the-convention/national-communications-andbiennial-reports-annex-i-parties/submitted-nationalcommunications-from-annex-i-parties) accessed 10 April 2019.

US Department of Energy, 2016, *Climate change and the electricity sector: guide for climate change resilience planning*, Department of Energy, Washington DC (https://www.energy.gov/sites/prod/files/2016/10/f33/ Climate%20Change%20and%20the%20Electricity%20 Sector%20Guide%20for%20Climate%20Change%20 Resilience%20Planning%20September%202016\_0.pdf) accessed 10 April 2019.

Vailles, C. and Metivier, C., 2019, Very few companies make good use of scenarios to anticipate their climateconstrained future, I4CE Climate Brief No 61, Institute for Climate Economics, Paris (https://www.i4ce.org/ download/very-few-companies-make-good-use-ofscenarios-to-anticipate-their-climate-constrainedfuture/) accessed 10 April 2019.

Varianou Mikellidou, C., et al., 2017, 'Energy critical infrastructures at risk from climate change: a state of the art review', *Safety Science* (DOI: 10.1016/j. ssci.2017.12.022).

van Vliet, M. T. H., et al., 2012, 'Vulnerability of US and European electricity supply to climate change', *Nature Climate Change* 2(9), pp. 676-681 (DOI: 10.1038/ nclimate1546).

van Vliet, M. T. H., et al., 2013a, 'Global river discharge and water temperature under climate change', *Global Environmental Change* 23(2), pp. 450-464 (DOI: 10.1016/j. gloenvcha.2012.11.002).

van Vliet, M. T. H., et al., 2013b, 'Water constraints on European power supply under climate change: impacts on electricity prices', *Environmental Research Letters* 8(3), p. 035010 (DOI: 10.1088/1748-9326/8/3/035010).

van Vliet, M. T. H., et al., 2016a, 'Impacts of recent drought and warm years on water resources and

electricity supply worldwide', *Environmental Research Letters* 11(12), p. 124021 (DOI: 10.1088/1748-9326/11/12/124021).

van Vliet, M. T. H., et al., 2016b, 'Multi-model assessment of global hydropower and cooling water discharge potential under climate change', *Global Environmental Change* 40, pp. 156-170 (DOI: 10.1016/j. gloenvcha.2016.07.007).

van Vliet, M. T. H., et al., 2016c, 'Power-generation system vulnerability and adaptation to changes in climate and water resources', *Nature Climate Change* 6(4), pp. 375-380 (DOI: 10.1038/nclimate2903).

Vogel, R., et al., 2014, *Klimaadaptatie en energieinfrastructuur*, TNO, Den Haag, Netherlands (http:// edepot.wur.nl/344929) accessed 19 September 2018.

Voigt, S., et al., 2014, 'Energy intensity developments in 40 major economies: structural change or technology improvement?', *Energy Economics* 41, pp. 47-62 (DOI: 10.1016/j.eneco.2013.10.015).

Vousdoukas, M. I., et al., 2017, 'Extreme sea levels on the rise along Europe's coasts', *Earth's Future* 5(3), pp. 304-323 (DOI: 10.1002/2016EF000505).

Walker, A., et al., 2014, *Counting the cost: the economic and social costs of electricity shortfalls in the UK — a report for the Council for Science and Technology*, Royal Academy of Engineering, London (https://www.raeng. org.uk/publications/reports/counting-the-cost) accessed 3 March 2019.

Watkiss, P., et al., 2014, 'The use of new economic decision support tools for adaptation assessment: a review of methods and applications, towards guidance on applicability', *Climatic Change* 132(3), pp. 401-416 (DOI: 10.1007/s10584-014-1250-9).

Watkiss, P., et al., 2018, *The economic cost of climate change in Europe: synthesis report on state of knowledge and key research gaps*, COACCH Policy Brief, COACCH project (https://www.coacch.eu/wp-content/uploads/2018/05/COACCH\_Synthesis-Report-State-of-Knowledge\_1st-draft.pdf) accessed 10 April 2019.

WBCSD, 2014, *Building a resilient power sector*, World Business Council for Sustainable Development, Geneva (https://www.wbcsd.org/Programs/Energy-Circular-Economy/Climate-Energy/Resources/Building-a-Resilient-Power-Sector) accessed 10 April 2019.

Wenz, L., et al., 2017, 'North–south polarization of European electricity consumption under future

warming', Proceedings of the National Academy of Sciences, p. 201704339 (DOI: 10.1073/pnas.1704339114).

Wild, M., et al., 2015, 'Projections of long-term changes in solar radiation based on CMIP5 climate models and their influence on energy yields of photovoltaic systems', *Solar Energy* 116, pp. 12-24 (DOI: 10.1016/j. solener.2015.03.039).

Wild, M., et al., 2017, 'Impact of climate change on future concentrated solar power (CSP) production', conference paper presented at: Radiation Processes in the Atmosphere and Ocean (IRS2016): Proceedings of the International Radiation Symposium (IRC/IAMAS), Auckland, New Zealand, 2017.

WMO/GFCS, 2017, Energy exemplar to the user interface platform of the Global Framework for Climate Services, World Meteorological Organization (WMO) and Global Framework for Climate Services (GFCS), Geneva (https:// library.wmo.int/doc\_num.php?explnum\_id=3581) accessed 10 April 2019.

World Energy Council, 2015, *The road to resilience — managing and financing extreme weather risks*, World Energy Council (https://www.worldenergy.org/wp-content/uploads/2015/09/The-Road-to-Resilience-Managing-and-Financing-Extreme-Weather-Risk.pdf) accessed 10 April 2019.

World Energy Council, 2016, *The road to resilience* — *managing the risks of the energy-water-food nexus*, World Energy Council (https://www.worldenergy.org/ wp-content/uploads/2016/03/The-road-to-resiliencemanaging-the-risks-of-the-energy-water-food-nexus\_ early-findings-report.pdf) accessed 10 April 2019.

World Energy Council, 2019, 'Energy Resources | Europe | Hydropower' (https://www.worldenergy.org/ data/resources/region/europe/hydropower/) accessed 27 February 2019.

Yair, Y., 2018, 'Lightning hazards to human societies in a changing climate', *Environmental Research Letters* 13(12), p. 123002 (DOI: 10.1088/1748-9326/aaea86).

van Zalk, J. and Behrens, P., 2018, 'The spatial extent of renewable and non-renewable power generation: a review and meta-analysis of power densities and their application in the U.S.', *Energy Policy* 123, pp. 83-91 (DOI: 10.1016/j.enpol.2018.08.023).

Zanuttigh, B., et al., 2016, 'A methodology for multicriteria design of multi-use offshore platforms for marine renewable energy harvesting', *Renewable Energy* 85, pp. 1271-1289 (DOI: 10.1016/j.renene.2015.07.080).

European Environment Agency

Adaptation challenges and opportunities for the European energy system: building a climate-resilient low-carbon energy system

2019 — 117 pp. — 21 x 29.7 cm

ISBN 978-92-9480-065-7 doi:10.2800/227321

## HOW TO OBTAIN EU PUBLICATIONS

## Free publications:

- one copy: via EU Publications website (https://publications.europa.eu);
- more than one copy or posters/maps: from the European Union's representations (http://ec.europa.eu/represent\_en.htm); from the delegations in non-EU countries (http://eeas.europa.eu/delegations/index\_en.htm); by contacting the Europe Direct service (http://europa.eu/europedirect/index\_en.htm) or calling 00 800 6 7 8 9 10 11 (freephone number from anywhere in the EU) (\*).

(\*) The information given is free, as are most calls (though some operators, phone boxes or hotels may charge you).

## **Priced publications:**

• via EU Publications website (https://publications.europa.eu).

European Environment Agency Kongens Nytorv 6 1050 Copenhagen K Denmark

Tel.: +45 33 36 71 00 Web: eea.europa.eu Enquiries: eea.europa.eu/enquiries





European Environment Agency